Joepe Kaandorp Allopurinol for fetal neuro- and cardiovascular protection



Voor myte moeder

Title: *Allopurinol for fetal neuro- and cardiovascular protection*, Thesis, Utrecht University, the Netherlands Author: Joepe J Kaandorp ISBN: 978-90-393-6166-5

Editor: Jacob Kaandorp Cover design: Guus Kaandorp, "Djeva" by June Quist Layout: Guus Kaandorp and Sjoerd Kaandorp, Amsterdam / Bergen aan Zee Print: Gildeprint, Enschede

The research described in this thesis was performed at the department of Perinatology, University Medical Center Utrecht, the Netherlands.

© 2014 by J.J.Kaandorp, Amsterdam, the Netherlands.

All rights reserved, no part of this thesis may be reproduced or transmitted in any form or by any means, without permission of the copyright owner.

The author gratefully acknowledges the unconditional financial support for printing this thesis by AcePharmaceuticals.

Allopurinol for fetal neuro- and cardiovascular protection

Allopurinol voor foetale neuro- en cardiovasculaire protectie

(met een samenvatting in het Nederlands)

PROEFSCHRIFT

ter verkrijging van de graad van doctor aan de Universiteit Utrecht op gezag van de rector magnificus, prof.dr. G.J. van der Zwaan, ingevolge het besluit van het college voor promoties in het openbaar te verdedigen op donderdag 30 januari 2014 des middags te 4 uur

door

JOEPE JANNEMIEK KAANDORP geboren op 6 april 1981 te Utrecht

Promotoren

Prof.dr. G.H.A. Visser Prof.dr. F. Van Bel

Co-promotoren

Dr. J.B. Derks Dr. M.J.N.L. Benders

CONTENTS

Chapter 1	General introduction	7
Chapter 2	Fetal hypoxia is an important determinant of birth as- phyxia and subsequent adverse outcome: antenatal neuroprotection at term.	15
Chapter 3	Long-term neuroprotective effects of allopurinol after moderate perinatal asphyxia. Follow-up of two random- ized controlled trials.	31
Chapter 4	Maternal to fetal allopurinol transfer and xanthine oxi- dase suppression in the late gestation pregnant rat.	43
Chapter 5	Xanthine oxidase and the fetal cardiovascular defense to hypoxia in late gestation ovine pregnancy.	53
Chapter 6	Antenatal allopurinol reduces hippocampal brain damage after acute birth asphyxia in late gestation fetal sheep.	71
Chapter 7	Antenatal allopurinol for reduction of birth asphyxia in- duced brain damage (ALLO-trial); a randomized double blind placebo controlled multicenter study (study proto- col).	85
Chapter 8	Rapid target allopurinol concentrations in the hypoxic fetus after maternal administration during labor.	93
Chapter 9	Maternal allopurinol administration during suspected fe- tal hypoxia; a novel neuroprotective intervention? A mul- ticenter randomized controlled trial (the ALLO-trial).	105
Chapter 10	General discussion	121
Chapter 11	Summary Nederlandse samenvatting	128 132
Chapter 12	List of publications List of co-authors	138 139

Chapter 1 General introduction



Despite advances in obstetric practice, acute intra-partum fetal hypoxia-ischemia remains one of the most common forms of fetal stress and is associated with the development of cerebral palsy and cognitive disabilities in later life ^[1,3]. Hypoxia-ischemia (HI) or birth asphyxia has an incidence of 4-to-9 per 1000 neonates and is defined as an impaired maternal-fetal respiratory gas exchange resulting in hypoxemia, accompanied by the development of a combined respiratory and metabolic acidosis ^[1-5]. Asphyxia will only lead to functional and structural organ damage when the period of circulatory failure is prolonged enough to result in a significant production of toxic free radicals.

Pathophysiology of acute fetal asphyxia

Free radicals are atoms or molecules that are highly reactive with other cellular structures because they contain unpaired electrons. They are formed as necessary intermediates in a variety of normal biochemical reactions, but when generated in excess or not appropriately controlled, radicals can wreak havoc on a broad range of macromolecules. There are many types of radicals, but those of most concern in biological systems are derived from oxygen, and known collectively as reactive oxygen species (ROS). Oxygen has two unpaired electrons in separate orbits in its outer shell. This electronic structure makes oxygen especially susceptible to radical formation. Free radicals can cause damage to parts of cells such as proteins, DNA, and cell membranes by stealing their electrons through a process called oxidation. This is why free radical damage is called 'oxidative stress'. When free radicals oxidize important components of the cell, those components lose their ability to function normally, and the accumulation of such damage may cause the cell to die.

Although asphyxia is a global insult, it is the brain, the heart and the kidneys that are particularly at risk of damage from ischemia-reperfusion injury associated with asphyxia ^[5].

There is increasing evidence from experimen-

tal and clinical studies that a substantial part of birth asphyxia-related brain damage occurs upon and immediately after birth, when brain perfusion and oxygenation are recovering. The excessive formation of reactive oxygen species (ROS) and other toxic compounds peaks 30-to-60 minutes after birth and plays an important causative role in this reperfusion-reoxygenation damage to the developing brain ^[6-12]. These pathways are described in more detail in **Chapter 2** of this thesis.

A brief history of perinatal neuroprotection

Earlier and recent experimental studies in newborn animals suffering HI have suggested that postnatal pharmacologic treatment or moderate hypothermia to prevent or reduce the excessive production of free radicals can reduce post-HI damage to the newborn brain if treatment was started within 6 hours after the start of reperfusion and reoxygenation, the so-called "window of therapeutic opportunity" ^[13,14]. Up to now, in the clinical setting, only moderate hypothermia proved to be beneficial to reduce post-HI damage to the brain in mildly and moderately asphyxiated human neonates ^[15,16]. Most pharmacologic treatment modalities, even started within the window of opportunity of 6 hours after birth, did not result in an improved longterm neurodevelopmental outcome ^[14,17]. Postnatally administered neuroprotective agents may be given too late to be fully effective in terms of reduction of free radical formation, like superoxide and peroxynitrite, which have their maximal production immediately (within 30 minutes) after birth ^[9].

Earlier (pharmacologic) treatment, i.e. antenatally via the mother in case of suspected fetal hypoxia, might be a more rational approach since the concentration of neuroprotective drugs is already near or even within the therapeutical range at birth and during the reperfusion-reoxygenation period shortly after birth. It thus provides the opportunity to limit or prevent the consequences of potentially injuring mechanisms. This approach may reduce birth asphyxia-related damage to the newborn heart and brain and improve neurodevelopmental outcome of the threatened fetus and newborn in an earlier stage of the process.

To successfully neuroprotect the hypoxic fetus in utero and during reperfusion and reoxygenation around the process of birth, a 'candidate' drug should qualify to the following criteria:

- 1. it should readily cross the placenta;
- 2. it should not be toxic to the mother;
- 3. it should have a therapeutic effect on the fetus, and
- 4. it should not be toxic to the fetus.

To date there is not much reported in experimental or clinical studies about antenatal pharmacologic therapies to reduce or prevent fetal hypoxia-induced neonatal brain damage.

Allopurinol

The xanthine oxidase (XO) inhibitor allopurinol has been the cornerstone of the clinical management of gout and conditions associated with hyperuricemia since it was approved by the Food and Drug Administration for that specific indication in 1966 ^[18]. In the succeeding decades an increasing number of researchers suggested potential additional beneficial effects of allopurinol for various forms of ischemic tissue and vascular injuries, and congestive heart failure ^[18].

In the early 1980's Granger *et al.* demonstrated that ischemic bowel injury occurred especially on reperfusion and was attenuated by superoxide dismutase (SOD)^[19]. On the basis of this observation, the hypothesis was put forward that XO-derived reactive oxygen species (ROS) contribute to the ischemic injury via ATP catabolism during hypoxia and increasing electron acceptor availability on reperfusion. Since the introduction of the concept of ischemia-reperfusion injury, several lines of evidence support

the role of XO-derived ROS generation, and beneficial effects of XO inhibitors against ischemic damage of the heart, brain, intestine, liver, kidney, lung, and other tissues were shown ^[18,20].

Mechanism of action

Formation of the superoxide free radical due to conversion of hypoxanthine into xanthine by xanthine oxidase has an important role in the occurrence of hypoxia-ischemia induced injury of the brain ^[21]. Administration of the xanthine oxidase inhibitor allopurinol reduces the production of superoxide free radical formation, thereby potentially limiting the amount of hypoxia-reperfusion damage ^[22,23]. Furthermore, allopurinol also has a non-protein bound iron (pro-radical) chelating and direct free radical (hydroxyl) scavenging effect, especially when high dosages are administered ^[18,24].

Earlier studies

The first human studies investigating potential beneficial effects of allopurinol on ischemia-reperfusion damage focused on the pathogenic role of XO in myocardial ischemia and showed that inhibiting XO activation during cardiac surgery by allopurinol treatment improved cardiac performance, scored by cardiac index, and reduced the need for inotropic or mechanical support ^[25]. A more recently performed study by Clancy *et al.* showed that allopurinol provided significant neurocardiac protection in higher-risk hypoplastic heart infants who underwent cardiac surgery using deep hypothermic circulatory arrest ^[26].

Allopurinol made its way to neonatal medicine already in the early 1990's ^[22,27,28] when Palmer *et al.* showed that high-dose allopurinol administered at 15 min after recovery from cerebral hypoxia-ischemia markedly reduced both acute brain edema and long-term cerebral injury in immature rats. A prospective randomized study in human neonates, examining the effects of allopurinol in term asphyxiated neonates, showed an improvement of electrocortical brain activity, and a reduction in free radical formation after early neonatal (<4 hours of life) allopurinol administration ^[29]. A more recent paper by Gunes *et al*. ^[30] reports an improved neurological outcome after postnatal allopurinol administration compared to placebo in term moderately asphyxiated neonates. A subsequent randomized clinical trial by Benders *et al.* ^[17] however could not confirm these results and suggested that when birth asphyxia had been too severe, the inflicted brain damage could no longer be reversed. A recent follow-up study of the two clinical trials performed by van Bel and Benders et al., showed a potential beneficial effect of postnatally administered allopurinol on mortality and severe neurological disabilities at 4-8 years of age, but only in a subgroup of moderately asphyxiated infants (Chapter 3 of this thesis). These beneficial effects are in line with previous findings on neonatal head cooling after acute fetal hypoxia at term ^[15]. Most likely no advantage of treatment occurs anymore when the interval to the initiation of treatment is too long or when brain damage is already too severe. It is conceivable that earlier allopurinol treatment, e.g. maternally administered allopurinol during labor in case of suspected fetal hypoxia, may earlier limit the amount of free radical production and subsequent hypoxia-reperfusion injury.

Pharmacologic data have been published by Boda *et al.*, which showed that therapeutical plasma levels were reached within 20 minutes in the human fetus after oral allopurinol administration (600 mg) to the mother ^[31]. Some animal studies have been performed using piglets (neonatal administration), sheep and sows (maternal administration) ^[32:35]. These studies showed an adequate placental passage of allopurinol and an improved outcome after allopurinol treatment on cerebral energy status and recovery of umbilical blood flow respectively. Furthermore, we showed reduced fetal cardiac oxidative stress in chronically instrumented fetal sheep after antenatal allopurinol administration to the pregnant ewe during repeated periods of ischemia ^[35]. In addition our research group performed a prospective randomized placebo controlled double blind pilot study, in which we administered intravenously allopurinol (500 mg) to pregnant women when fetal asphyxia was suspected. Data from this pilot study showed an inverse correlation between levels of allopurinol and the amount of S100ß, a biomarker for brain tissue damage, in cord blood ^[36].

Potential side effects

Established side effects of allopurinol are rare and most commonly include symptoms of hypersensitivity like skin rashes ^[37]. Stevens-Johnson syndrome has been reported, but only after prolonged treatment (> 10 days) in case of gout ^[38]. During previously performed studies in pregnant women and neonates, including those performed by our group, no adverse reactions were seen ^[17,26,29,31,32,36].

In conclusion

It becomes increasingly clear that postnatal (pharmacologic) treatment with antioxidative drugs or with NMDA-receptor antagonists of birth asphyxia-related damage to the newborn brain has a limited neuroprotective effect, even when therapy has been started within the "window of opportunity", being within 6 hours after birth. This might be due to the maximal formation of reactive (oxygen) species and other toxic agents like peroxynitrite upon and within the first 30-to-60 minutes after reperfusion and reoxygenation. Moreover, also during fetal hypoxia, an important determinant of birth asphyxia, free radicals and other toxic agents are produced albeit in much lower concentrations as compared to the early reperfusion phase. The fetus of pregnant women on the brink of delivery, with signs of (imminent) hypoxia may therefore benefit from additional maternal treatment with drugs reducing excessive neurotransmitter formation and formation of free radicals. If necessary, early postnatal hypothermia with or

without pharmacologic neuroprotective agents can further reduce or even prevent fetal hypoxic and birth asphyxia-related brain damage of the newborn.

AIM OF THIS THESIS

The aim of this thesis is to investigate the applicability of maternally administered allopurinol for fetal neuroprotection and to define strategies for future practice.

OUTLINE OF THIS THESIS

Chapter 2 summarizes the molecular mechanisms underlying early reperfusion-reoxygenation damage and focuses on the most investigated pharmacological agents (phenobarbital, vitamin C and E, allopurinol, melatonin and xenon) to be given antenatally to the mother to neuroprotect the hypoxic fetus. **Chapter 3** investigates the long-term effects of neonatal allopurinol-treatment in infants suffering from moderate to severe birth asphyxia.

Part I: Animal studies

Chapter 4 provides a model that can be used to investigate the role of xanthine oxidase derived reactive oxygen species in the pathogenesis of fetal hypoxic brain injury and the long-term neurological effects in rats.

Chapter 5 investigated the contribution of reactive oxygen species, derived from xanthine oxidase, to the fetal peripheral vasoconstrictor response to hypoxia via interaction with Nitric Oxidase-dependent mechanisms in the ovine fetus.

In **Chapter 6** we tested the hypothesis that maternal treatment with allopurinol during repeated episodes of fetal asphyxia would limit ischemia-reperfusion damage to the fetal brain in chronically instrumented fetal sheep in late gestation.

Part II: Clinical trial

Chapter 7 describes the protocol of the multicenter randomized double blind placebo controlled clinical trial, which we performed to assess whether maternal allopurinol treatment during fetal hypoxia would reduce the release of biomarkers associated with neonatal brain damage.

Chapter 8 focuses on the placental transfer of allopurinol.

Chapter 9 shows the main outcome measures of the multicenter randomized double blind placebo controlled clinical trial.

Finally, in **Chapter 10** the results and conclusions of the conducted studies are discussed and suggestions for future practice and research with respect to the use of allopurinol in perinatal medicine are made.

REFERENCES OF CHAPTER 1

1. Hagberg B, Hagberg G, Olow I. The changing panorama of cerebral palsy in sweden. VI. prevalence and origin during the birth year period 1983-1986. Acta Paediatr. 1993;82(4):387-393.

2. Vannucci RC. Hypoxic-ischemic encephalopathy. Am J Perinatol. 2000;17(3):113-120.

3. Ferriero DM. Neonatal brain injury. N Engl J Med. 2004;351(19):1985-1995.

4. Low JA. Determining the contribution of asphyxia to brain damage in the neonate. J Obstet Gynaecol Res. 2004;30(4):276-286.

5. Bennet L, Westgate JA, Gluckman PD, Gunn AJ. Pathophysiology of asphyxia. In: Levene M, Whittle M, Chervenak FA, Punt J, Bennett MJ, eds. Fetal and neonatal neurology and neurosurgery. 3rd ed. London, England: Harcourt Health Sciences; 2001:407.

6. McCord JM. Oxygen-derived free radicals in postischemic tissue injury. N Engl J Med. 1985;312(3):159-163.

7. Rosenberg AA, Murdaugh E, White CW. The role of oxygen free radicals in postasphyxia cerebral hypoperfusion in newborn lambs. Pediatr Res. 1989;26(3):215-219.

8. Kirsch JR, Helfaer MA, Lange DG, Traystman RJ. Evidence for free radical mechanisms of brain injury resulting from ischemia/reperfusion-induced events. J Neurotrauma. 1992;9 Suppl 1:S157-63.

9. Dirnagl U, Lindauer U, Them A, et al. Global cerebral ischemia in the rat: Online monitoring of oxygen free radical production using chemiluminescence in vivo. J Cereb Blood Flow Metab. 1995;15(6):929-940.

10. Fellman V, Raivio KO. Reperfusion injury as the mechanism of brain damage after perinatal asphyxia. Pediatr Res. 1997;41(5):599-606.

11. Grow J, Barks JD. Pathogenesis of hypoxic-ischemic cerebral injury in the term infant: Current concepts. Clin Perinatol. 2002;29(4):585-602, v.

12. Ono T, Tsuruta R, Fujita M, *et al.* Xanthine oxidase is one of the major sources of superoxide anion radicals in blood after reperfusion in rats with forebrain ischemia/reperfusion. Brain Res. 2009;1305:158-167.

 van Bel F, Groenendaal F. Long-term pharmacologic neuroprotection after birth asphyxia: Where do we stand? Neonatology. 2008;94(3):203-210.

14. Jenkins DD, Chang E, Singh I. Neuroprotective interventions: Is it too late? J Child Neurol. 2009;24(9):1212-1219.

15. Gluckman PD, Wyatt JS, Azzopardi D, *et al.* Selective head cooling with mild systemic hypothermia after neonatal encephalopathy: Multicentre randomized trial. Lancet. 2005;365(9460):663-670.

16. Shankaran S, Laptook AR. Hypothermia as a treatment for birth asphyxia. Clin Obstet Gynecol. 2007;50(3):624-635.

17. Benders MJ, Bos AF, Rademaker CM, et al. Early postnatal allopurinol does not improve short term outcome after severe birth asphyxia. Arch Dis Child Fetal Neonatal Ed. 2006;91(3):F163-5.

18. Pacher P, Nivorozhkin A, Szabo C. Therapeutic effects of xanthine oxidase inhibitors: Renaissance half a century after the discovery of allopurinol. Pharmacol Rev. 2006;58(1):87-114.

19. Granger DN, Rutili G, McCord JM. Superoxide radicals in feline intestinal ischemia. Gastroenterology. 1981;81(1):22-29.

20. Granger DN, McCord JM, Parks DA, Hollwarth ME. Xanthine oxidase inhibitors attenuate ischemia-induced vascular permeability changes in the cat intestine. Gastroenterology. 1986;90(1):80-84.

21. Warner DS, Sheng H, Batinic-Haberle I. Oxidants, antioxidants and the ischemic brain. J Exp Biol. 2004;207(Pt 18):3221-3231.

22. Palmer C, Vannucci RC, Towfighi J. Reduction of perinatal hypoxic-ischemic brain damage with allopurinol. Pediatr Res. 1990;27(4 Pt 1):332-336.

23. Whitelaw A, Thoresen M. Clinical trials of treatments after perinatal asphyxia. Curr Opin Pediatr. 2002;14(6):664-668.

24. Saugstad OD. Role of xanthine oxidase and its inhibitor in hypoxia: Reoxygenation injury. Pediatrics. **1996**;98(1):103-107.

25. Johnson WD, Kayser KL, Brenowitz JB, Saedi SF. A randomized controlled trial of allopurinol in coronary bypass surgery. Am Heart J. 1991;121(1 Pt 1):20-24.

26. Clancy RR, McGaurn SA, Goin JE, *et al.* Allopurinol neurocardiac protection trial in infants undergoing heart surgery using deep hypothermic circulatory arrest. Pediatrics. 2001;108(1):61-70.

27. Palmer C, Vannucci RC. Potential new therapies for perinatal cerebral hypoxia-ischemia. Clin Perinatol. 1993;20(2):411-432.

28. Palmer C, Towfighi J, Roberts RL, Heitjan DF. Allopurinol administered after inducing hypoxia-ischemia reduces brain injury in 7-day-old rats. Pediatr Res. **1993;33(4** Pt **1)**:405-411.

29. Van Bel F, Shadid M, Moison RM, *et al.* Effect of allopurinol on postasphyxial free radical formation, cerebral hemodynamics, and electrical brain activity. Pediatrics. 1998;101(2):185-193.

30. Gunes T, Ozturk MA, Koklu E, Kose K, Gunes I. Effect of allopurinol supplementation on nitric oxide levels in asphyxiated newborns. Pediatr Neurol. 2007;36(1):17-24.

31. Boda D, Németh P, Kiss P, Orvos H. Treatment of mothers with allopurinol to produce therapeutic blood levels in newborns. Prenat Neonat Med. 1999;4:130-132, 133, 134.

32. Peeters-Scholte C, Braun K, Koster J, *et al.* Effects of allopurinol and deferoxamine on reperfusion injury of the brain in newborn piglets after neonatal hypoxia-ischemia. Pediatr Res. 2003;54(4):516-522.

33. Masaoka N, Nakajima Y, Hayakawa Y, et al. Transplacental effects of allopurinol on suppression of oxygen free radical production in chronically instrumented fetal lamb brains during intermittent umbilical cord occlusion. J Matern Fetal Neonatal Med. 2005;18(1):1-7.

34. van Dijk AJ, Parvizi N, Taverne MA, Fink-Gremmels J. Placental transfer and pharmacokinetics of allopurinol in late pregnant sows and their fetuses. J Vet Pharmacol Ther. 2008;31(6):489-495.

35. Derks JB, Oudijk MA, Torrance HL, *et al.* Allopurinol reduces oxidative stress in the ovine fetal cardiovascular system after repeated episodes of ischemia-reperfusion. Pediatr Res. 2010;68(5):374-380.

36. Torrance HL, Benders MJ, Derks JB, *et al*. Maternal allopurinol during fetal hypoxia lowers cord blood levels of the brain injury marker S-100B. Pediatrics. 2009;124(1):350-357.

37. Summary of product characteristics acepurin®. http://db.cbg-meb.nl/Bijsluiters/h09974.pdf. Updated 2012.

38. Halevy S, Ghislain PD, Mockenhaupt M, *et al.* Allopurinol is the most common cause of stevens-johnson syndrome and toxic epidermal necrolysis in europe and israel. J Am Acad Dermatol. 2008;58(1):25-32.

Chapter 2 Fetal hypoxia is an important determinant of birth asphyxia and subsequent adverse outcome: antenatal neuroprotection at term

JOEPE J. KAANDORP, MANON J.N.L. BENDERS, JAN B. DERKS, FRANK VAN BEL

Paediatrics and Child Health, Volume 20, Issue 8, August 2010, Pages 356-361



ABSTRACT

Hypoxic ischemic encephalopathy is directly associated with the development of cerebral palsy and cognitive disabilities later in life, therefore remaining an important problem in perinatal medicine. Postnatal neuroprotective strategies have been investigated elaborately, but up to now, only moderate hypothermia proved to be beneficial in reducing post hypoxic-ischemic encephalopathy in a selected group of asphyxiated neonates. Since the vast amount of toxic free radicals is produced in the reperfusion and reoxygenation period upon and immediately (30-to-60 minutes) after birth, we postulate that antenatal (i.e. maternal) pharmacological neuroprotection of the fetus, combined with postnatal hypothermia, might be a more optimal approach to prevent this free radical induced brain damage. This review summarizes the molecular mechanisms underlying early reperfusion-reoxygenation damage and focuses on the most promising pharmacological agents (phenobarbital, vitamin C and E, allopurinol, melatonin and xenon) to be given antenatally to the mother to neuroprotect the hypoxic fetus. ith an incidence of 4-to-9 per 1000 neonates, perinatal hypoxia-ischemia (HI) or birth asphyxia is one of the main causes of neonatal HI encephalopathy, which is associated with the development of cerebral palsy and cognitive disabilities later in life ^[1-4]. It therefore is one of the most fundamental problems in perinatal medicine.

Asphyxia is defined as an impaired maternal-fetal respiratory gas exchange resulting in hypoxemia, accompanied by the development of a combined respiratory and metabolic acidosis ^[5]. Asphyxia will only lead to functional and structural organ damage when the period of circulatory failure is prolonged enough to result in a significant production of toxic free radicals ^[4]. Although asphyxia is a global insult, it is the brain, the heart and the kidneys that are particularly at risk of damage from ischemia-reperfusion injury associated with asphyxia ^[5].

Possible causes of fetal hypoxia during labor include umbilical cord compression, poor placental function, placental abruption, inadequate relaxation of the uterus (leading to poor placental oxygenation), inadequate maternal oxygenation, and low maternal blood pressure ^[6]. There is increasing evidence from experimental and clinical studies that a substantial part of birth asphyxia-related brain damage occurs upon and immediately after birth when brain perfusion and oxygenation are recovering. The excessive formation of reactive oxygen species, such as free radicals, and other toxic compounds peaks 30-to-60 minutes after birth and plays an important causative role in this reperfusion-reoxygenation damage to the immature brain ^[7-10].

Earlier and recent experimental studies in newborn animals suffering HI have suggested that postnatal pharmacologic treatment or moderate hypothermia to prevent or reduce the excessive production of free radicals can reduce post-HI damage to the newborn brain if treatment was started within 6 hours after the start of reperfusion and reoxygenation, the so-called "window of therapeutic opportunity" [11,12]. Up to now, in clinical trials, only moderate hypothermia proved to be beneficial to reduce post-HI damage to the brain in mildly and moderately asphyxiated human neonates ^[13,14]. Most pharmacologic treatment modalities, even started within the therapeutic window of 6 hours after birth, did not result in an improved longterm neurodevelopmental outcome in human infants ^[11,15]. We postulate that postnatally administered neuroprotective agents are likely to be given too late to be really effective in terms of reduction of free radical formation or other toxic compounds like superoxide and peroxynitrite (see also below), which have their maximal production immediately (within 30 minutes) after birth.

We therefore hypothesize that antenatal (pharmacologic) treatment of the hypoxic fetus, via the mother, is a more rational approach since the concentration of neuroprotective drugs are already near or even within the therapeutical range at birth and during the reperfusion-reoxygenation period shortly after birth.

This review briefly summarizes the molecular mechanisms underlying early reperfusion-reoxygenation damage to the newborn brain and the current knowledge with respect to early recognition of fetal hypoxia during labor. The paper also reviews the most promising pharmacologic agents to be administered antenatally to the mother to reduce the consequences of fetal hypoxia, i.e. to reduce the production of reactive (oxygen) species like free radicals during fetal hypoxia and especially upon reperfusion and reoxygenation shortly after birth.

Molecular mechanisms of early post hypoxic-ischemic reperfusion injury to the newborn brain

Birth asphyxia-induced brain injury is caused by a chain of events. During sustained hypoxia (with or without ischemia) formation of ade-

FETAL HYPOXIA IS AN IMPORTANT DETERMINANT OF BIRTH ASPHYXIA AND SUBSEQUENT ADVERSE OUTCOME: ANTENATAL NEUROPROTECTION AT TERM

nosine triphosphate (ATP) is impaired, leading to the failure of ATP-dependent sodium-potassium pumps that normally maintain the polarity of neuronal membranes. Depolarization of the neuronal membrane induces a release of excitatory neurotransmitters causing an excessive influx of calcium into the neuronal and microglial cells through voltage regulated and receptor regulated (N-methyl-D-aspartate [NMDA]) calcium channels. This influx of calcium leads to a massive production of neurotransmitters such as glutamate, which induces an activation of many enzymatic reactions including the activation of lipases, proteases and endonucleases. These events further enhance the production of free radicals, which leads to lipid peroxidation and eventually to damage of neuronal cell membranes and nucleotides ^[16-18].



During fetal hypoxia, but especially upon reperfusion-reoxygenation opening of ion channels leads to influx of calcium into neurons with subsequent excessive neurotransmitters production and activation of enzymatic reactions. This activation gives rise to subsequent metabolisation of hypoxanthine, fatty acids originating from neuronal cell membranes, of nitric oxide production and the Fenton reaction (*) This leads to excessive free radicals formation, lipid peroxidation and ultimate neuronal cell death.

NMDA= N-methyl-D-aspartate; NO= nitric oxide; NOS= nitric oxide synthase; eNOS= endothelial NOS; nNOS= neuronal NOS; iNOS=inducible NOS, NPBI=non-protein bound iron.

Upon reoxygenation an important early source of free radical production occurs due to metabolisation of hypoxanthine (a degradation product of ATP), accumulated during the actual HI-period, into xanthine and uric acid by the activated enzyme xanthine oxidase, a process that needs oxygen. During this process large amounts of the superoxide free radical (O_{a}) are formed. The simultaneous occurring metabolisation of arachidonic acid, freed from damaged cell membranes, into prostaglandins leads to a further production of the superoxide free radical. This chain of reactions, causing an excess production of superoxide, culminates commonly within the first 30-60 minutes after the start of reperfusion and reoxygenation and plays a central role in further formation of free radicals ^[10,19]

- 1. Superoxide-derived hydrogen peroxide (H_2O_2) reacts with the pro-radical non protein-bound iron (NPBI), accumulated in neuronal and microglial cells during the actual HI insult due to a low intracellular pH, to form the very toxic hydroxyl (OH) free radical. This reaction is also known by the Fenton Reaction ^[20,21]. Several experimental and clinical studies indeed show that NPBI is related to brain damage in the immediate post-HI period ^[22,23].
- 2. Finally superoxide will also react with the free radical Nitric Oxide (NO) to form the toxic compound peroxynitrite (ONOO-). Peroxynitrite causes lipid peroxidation of neuronal cell membranes eventually leading to neuronal cell death [24]. Nitric Oxide is produced due to the activation of nitric oxide synthase (NOS), available in neurons, astrocytes (neuronal NOS) and endothelial cells (endothelial NOS) ^[25]. Further production of NO, somewhat later in the post-HI period, occurs because of activation of inducible NOS available in microglial cells and invaded white blood cells [24,26]. Although post-HI reperfusion-reoxygenation injury to the brain continues up to days and even weeks after the

completion of HI by a pro- and anti-inflammatory response and subsequent inappropriate neuronal cell apoptosis ^[27], these processes will not be discussed here in detail because the are beyond the scope of this review.

Figure 2-1 summarizes schematically the injuring mechanisms induced upon and early after reperfusion and reoxygenation.

Diagnosing fetal hypoxia

To timely start antenatal (maternal) neuroprotective treatment, it is important to appropriately identify fetal hypoxia in an early stage. Besides marked loss of fetal movements, a number of intrapartum surveillance techniques to diagnose fetal hypoxia are used nowadays.

Cardiotocography (CTG), introduced in the early 70's, allows continuous recording of the fetal heart rate, based on the R-R interval of the fetal electrocardiogram (ECG). Fetuses with progressive heart rate abnormalities caused by hypoxia such as decreased short-term variability or fetal tachycardia, followed by fetal bradycardia (due to a redistribution of organ blood perfusion), are often born with low Apgar scores and/or substantial metabolic acidosis. However, since the interpretation of the fetal heart rate trace is subject to poor inter- and intra-observer agreement on classification of tracings and a high percentage of false-positive registrations, CTG monitoring alone does not always provide us with adequate information to discriminate between normal or abnormal fetal condition during labor ^[28-30]. Therefore additional techniques to assess fetal well-being are necessary to obtain more detailed information.

Fetal blood sampling (FBS) is one of the oldest techniques to obtain complementary information about fetal oxygenation in case of a non-reassuring fetal heart rate tracing by measuring the pH in fetal scalp blood ^[31,32]. In case of a pH lower than 7.20, fetal asphyxia is imminent ^[32-34] and termination of pregnancy is advised ^{[35].} A recent Cochrane meta-analysis ^[36] reported data from 12 randomized controlled trials on the effectiveness of the CTG. They found that continuous monitoring by means of CTG without FBS leads to a significant increase the number of caesarean sections, without a marked positive effect in neonatal outcome. If, however, FBS was used alongside the CTG, a less prominent increase in caesarean section rate was seen with a reduction in neonatal seizures of 50%.

Another candidate tool for a more reliable intrapartum assessment of fetal condition is fetal pulse oximetry (FPO), in which the actual level of fetal hypoxemia is recorded. Two large trials with respect to the usefulness of FPO did not show any effectiveness in reducing neonatal morbidity or overall incidence of caesarean sections (CS) ^[37,38].

Since the 1990's intrapartum fetal monitoring by recording the fetal electrocardiogram has been possible. In case of overt fetal hypoxia, changes in the ST segment and PR interval of the fetal ECG are present. Several investigators have assessed the value of analyzing the parameters as an adjunct to conventional cardiotocography. A large randomized trial by Amer-Wahlin et al. [39] showed that analysis of the ST segment, in addition to conventional CTG monitoring, significantly reduced caesarean delivery rates as well as metabolic acidemia in umbilical arterial blood. A recent Dutch randomized clinical trial of nearly 6000 patients also showed a decrease in metabolic acidosis, although there was no improvement in short term neonatal outcome ^[40]. In addition, there was a 50 % reduction in fetal scalp sampling in patients where ST-analysis was used as intrapartum fetal monitoring compared to patients who were only monitored by CTG. For a more reliable conclusion regarding the value of this technique to appropriately diagnose fetal hypoxia in an early stage long-term follow-up data of the included children seems mandatory.

Given the above mentioned considerations fe-

tal monitoring by means of CTG or fetal ECG, combined with fetal blood sampling to obtain information about fetal acidosis seems the most optimal approach to timely diagnose fetal distress due to hypoxia at this moment.

Non-specific measures to reduce or prevent fetal hypoxia

If fetal hypoxia is suspected during labor, several general measures can be performed to optimize fetal condition, depending on the underlying problem.

Management of non-reassuring fetal heart rate patterns consists of correcting any fetal hypoxic-ischemic insult, if possible. Moving the mother to lateral position to decompress the vena cava is one of the first measures to optimize uteroplacental perfusion ^[6]. Correcting iatrogenic maternal hypotension after regional anesthesia, administrating oxygen to the mother, discontinuing oxytocin administration and a single intravenous injection of tocolysis to relax the uterus can further improve fetal oxygenation ^[6,41]. In case of umbilical cord compression a change of maternal position or amnioinfusion can give relief. A recent Cochrane review shows that amnioinfusion appears to be useful to reduce the occurrence of variable heart rate decelerations and lower the use of caesarean section. The trials reviewed, however, were too small to address the possibility of rare but serious maternal adverse effects of amnioinfusion [42]. Transvaginal amnion infusion can also serve to dilute or wash out thick meconium to possibly prevent meconium aspiration syndrome. Several studies however conclude that, although prophylactic amnioinfusion does dilute meconium, it does not improve perinatal outcome if standard peripartum surveillance is used [43-45].

Specific pharmacologic therapy to reduce or prevent fetal hypoxia-induced brain damage

As already hypothesized, antenatal pharma-

cologic treatment of the hypoxic fetus, via the mother, is a more rational approach since the concentration of neuroprotective drugs are already near or within the therapeutical range during the process of birth and during reperfusion and reoxygenation shortly after birth. This is the point of time where a maximal formation of reactive (oxygen) species such as free radicals and other toxic compounds can be expected, eventually leading to substantial damage to the newborn brain.

To successfully neuroprotect the hypoxic fetus in utero and during reperfusion and reoxygenation around the process of birth, a 'candidate' drug should qualify to the following criteria: 1) it should readily cross the placenta; 2) it should not be toxic to the mother; and 3) it should not be toxic to the fetus. To date there is not much reported in experimental or clinical studies about antenatal pharmacologic therapies to reduce or prevent fetal hypoxia-induced neonatal brain damage. Below we will summarize the scarce reports concerning this topic and discuss the most relevant drugs that may classify for this purpose.

Phenobarbital is a potential neuroprotective drug that easily passes the placenta and reaches concentrations that are equal to maternal concentrations^[46]. It is reported to be an NMDA-receptor antagonist and also has gamma-aminobutyric-acid (GABA)-receptor activating properties ^[47-49]. It might furthermore reduce fluctuations in blood pressure [50] and cerebral perfusion, prevent intracranial hemorrhage [51], and ischaemic brain injury due to a possible free radical scavenging effect ^[52]. Especially its effectiveness in preterm infants by reducing brain damage such as periventricular/intraventricular hemorrhages (PIVHs) is elaborately investigated. Postnatal treatment with phenobarbital has been the subject of a Cochrane review by Whitelaw and Odd [53], who stated that postnatal administration of phenobarbital cannot be recommended as prophylaxis to prevent PIVH

in preterm infants since there is no significant improvement in neurodevelopmental outcome and appeared to be associated with an increased need for mechanical ventilation due to respiratory depression. Early studies on maternal phenobarbital showed promising results in preventing PIVH in preterm infants [54-59]. However, adverse effects of antenatal phenobarbital were seen in a trial by Reinisch *et al* ^[60], who showed a possible detrimental influence on intelligence in adult males after in utero exposure to phenobarbital. Ajavi et al. [61] stated that early phenobarbital administration in term newborns with perinatal asphyxia increased the incidence of seizures with a trend towards increased mortality. However, up to now, no antenatal trials are known directed to reduce neonatal brain damage during and after fetal hypoxia. From above mentioned data we must conclude that it seems unrealistic to expect a reducing effect on the newborn brain of maternal phenobarbital treatment in case of fetal hypoxia and subsequent development of neonatal hypoxic-ischemic encephalopathy.

Vitamin C and E are important antioxidative agents that cross the placenta easily because these water-soluble vitamins are actively transported across the placenta from mother to fetus ^[62]. Therapeutic or prophylactic suppletion of vitamin C and E to pregnant women was suggested to have a protective effect in case of pre-eclampsia, since it should reduce oxidative-induced placental endothelial dysfunction [63]. However, a large study by Poston et al. showed no reduction in the incidence of preeclampsia, but did increase the rate of babies born with a low birth weight ^[64]. Also a large WHO study did not find any reduction in preeclampsia after administration of vitamin C and E to women with poor nutritional status ^[65].

Allopurinol. Formation of the superoxide free radical due to conversion of hypoxanthine into xanthine by xanthine oxidase plays a very important role in the occurrence of hypoxia- ischemia induced injury of the brain ^[66]. Administration of the xanthine oxidase inhibitor allopurinol (ALLO) reduces the production of superoxide free radical formation, thereby limiting the amount of hypoxia-reperfusion damage ^[67,68]. Furthermore, ALLO has also a non-protein bound iron (pro-radical) chelating and direct free radical (hydroxyl) scavenging effect, especially when high dosages are administered ^[19,69].

A prospective randomized study in human neonates, examining the effects of ALLO in term asphyxiated neonates, showed an improvement of electrocortical brain activity and a reduction in free radical formation after early neonatal (<4 hours of life) ALLO administration ^[70]. A recent paper by Gunes *et al.* ^[71] reports an improved neurological outcome after postnatal ALLO administration compared to a placebo in term moderately asphyxiated neonates. Moreover a randomized clinical trial by Benders *et al*. ^[15] had comparable results and showed a beneficial effect of ALLO on neurological development when the most severely asphyxiated neonates were excluded. Apparently, when asphyxia has been too severe, the inflicted brain damage can no longer be reversed. It is conceivable, that earlier ALLO treatment, i.e. the use of ALLO during labor in the case of suspected hypoxia, gives the opportunity to start even earlier with the treatment, thereby limiting the amount of hypoxia-reperfusion injury and further improving neurological outcome.

Pharmacologic data have been published by Boda *et al.*, which showed that therapeutical plasma levels were reached within 20 minutes in the human fetus after ALLO administration (600 mg) to the mother ^[72]. Some animal studies have been performed using piglets (neonatal administration), sheep and sows (maternal administration) ^[73-76]. These studies showed an adequate placental passage of ALLO and an improved outcome after ALLO treatment on cerebral energy status and recovery of umbilical blood flow respectively. Our research group recently performed a prospective randomized placebo controlled double blind pilot study, in which we administered intravenously ALLO (500 mg) to pregnant women when fetal asphyxia was imminent. Data from this pilot study showed an inverse correlation between levels of ALLO and the amount of S100ß, a biomarker for brain tissue damage, in cord blood ^[77]. In addition, we performed a study in chronically instrumented fetal sheep, in which we showed evidence of cardio- and neuroprotection after antenatal ALLO administration to the pregnant ewe during repeated periods of ischemia ^[74].

Allopurinol has been used in internal medicine for many years for the treatment of gout. Side effects of ALLO are rare and most commonly include symptoms of hypersensitivity like skin rashes^[78]. Stevens-Johnson syndrome has been reported, but only after prolonged treatment (> 10 days) in case of gout ^[79]. During previously performed studies in pregnant women and neonates, including those performed by our group, no adverse reactions were seen ^[72,77,80-82]. Together with the fact that the costs of ALLO-treatment are low, antenatal allopurinol seems to be promising as a neuroprotective agent.

Currently a national double blind placebo controlled multicenter trial has started in The Netherlands in which pregnant women on the brink of delivery are treated with 500 mg allopurinol or a placebo in case of suspected fetal distress.

Melatonin (N-acetyl-5-methoxytryptamine) is a small lipophilic indoleamine generated primarily in the pineal gland and secreted in a circadian manner ^[83]. Melatonin plays an important role in a variety of physiologic functions, including regulation of circadian rhythms as well as visual, reproductive, cerebrovascular, neuro-endocrine and neuro-immunological actions. Furthermore melatonin acts as a powerful scavenger of oxygen and nitrogen free radicals ^[84].

In adult animals melatonin is neuroprotective in models of focal cerebral ischemia ^[85] and the pre-treatment of melatonin (10 mg/kg intraperitoneal injection) reduces microglia activation in the hippocampus after kainic acid-induced inflammation in rats ^[86].

Previous experimental and clinical reports show that melatonin is rapidly transferred from maternal to fetal circulation in pregnant near term women up to 80% of the maternal concentration ^[87,88]. Prophylactic maternally administered melatonin (10 mg/kg intraperitoneally) prevents oxidative lipid peroxidation, DNA and mitochondrial damage in the brain of mature and premature fetal rats [89]. Besides proven benefits of prophylactic treatment, Hamada et al. [90] showed that maternal administration of melatonin within one hour after an HI-episode can prevent reperfusion-reoxygenation-induced oxidative cerebral damage in neonatal rats. Miller et al. ^[91] showed that intravenously administered melatonin to the mother (1 mg bolus, then 1 mg/h for 2 h) reduces hydroxyl free radical generation and lipid peroxidation in fetal sheep in response to asphyxia following umbilical cord occlusion. Another experimental study [92] demonstrated that post-asphyxial antenatal melatonin treatment attenuates the increase in activated microglia and 8-isoprostane (a marker of lipid peroxidation) production in mid-gestation fetal sheep. These protective actions of melatonin are likely due to its direct scavenging activity ^[93] and its indirect effects on antioxidant enzyme activities. Okatani et al showed that melatonin given to the pregnant rat increased antioxidant enzyme activity in the fetal brain which, thereby, provided indirect protection against free radical induced injury ^[94].

Many experimental studies have shown that maternal administration of melatonin before hypoxia significantly decreases oxidative damage in the fetal brain ^[88-94]. The choice of maternal melatonin treatment used as a single drug or in combination with other antioxidants such as ALLO and conventional treatments may lead to the development of a new approach for neuronal injury induced by fetal hypoxia and subsequent birth asphyxia.

Xenon. The noble anesthetic gas xenon (Xe) is an upcoming neuroprotectant in both in vitro and in vivo experimental studies ^[95-98].

Several incompletely understood characteristics of xenon might explain its neuroprotective effects ^[99]. Xenon is directly neuroprotective by being an NMDA-receptor antagonist ^[100], reducing overall neurotransmitter release ^[101], thereby limiting the excitotoxic apoptotic activity in the cascade of neuronal damage. Some authors postulate that xenon can also have a preconditioning effect, if administered before a hypoxic-ischemic insult, by inducing a pro-survival response by upregulation of a 'defensive' gene ^[102].

For these reasons xenon is an attractive candidate for neuroprotection because of the lack of adverse reactions and the absence of fetotoxicity ^[103]. Most studies are experimental and studied xenon in combination with hypothermia in neonatal rats ^[95-98]. Thoresen *et al*. ^[99] showed an optimal neuroprotective long-term effect on functional and pathological outcome in neonatal rats after 3 hours of hypothermia combined with 3 hours of xenon administration, with the neuroprotective effects of the two agents being additive. They even found a significant beneficial effect after 3 hours of hypothermia combined with one hour of xenon administration, thereby allowing transport to a specialized center and lowering costs. The latter being very important, since xenon administration is very expensive. When xenon concentrations are 70% or more xenon has a sedative effect. To our knowledge no human studies with xenon as a neuroprotective agent have been performed until now. At this moment one trial using postnatal xenon in combination with hypothermia in human neonates is running in the UK (trial register; Azzopardi, UK).

Besides the promising neuroprotective effects of postnatal treatment there might be a possible role for xenon as an antenatal neuroprotective agent. Xenon has received a marketing license as an anesthetic drug in several countries in Europe ^[103] and can for example be used as an anesthetic during caesarean section in case of fetal hypoxia, thereby allowing neuroprotection in an even earlier phase.

We are fully aware of the fact that our choice of drugs and agents discussed in this section may appear arbitrary and that it may be worthwhile to consider other drugs like N-acetylcysteine and other compounds with antioxidative properties for maternal treatment of the consequences of fetal hypoxia. On the other hand, the above-mentioned agents are drugs with which most experience has been obtained in maternal administration. Moreover in most instances knowledge of placental passage, a pre-requisite to treat the fetus via the mother, was best investigated in the drugs and agents discussed above.

Conclusions and recommendations

It becomes increasingly clear that postnatal (pharmacologic) treatment with antioxidative or with NMDA-receptor antagonists of birth asphyxia-related damage to the newborn brain

Figure 2-2. Proposed treatment scheme in case of fetal hypoxia with or without a subsequent occurrence of hypoxic-ischemic encephalopathy due to severe reperfusion-reoxygenation damage to the newborn brain



has a limited neuroprotective effect, even when therapy has been started within the "window of opportunity", being within 6 hours after birth. This is probably because the maximal formation of reactive (oxygen) species and other toxic agents like peroxynitrite occurs upon and within the first 30-to-60 minutes after reperfusion and reoxygenation. Moreover, also during fetal hypoxia, an important determinant of birth asphyxia, free radicals and other toxic agents are produced albeit in much lower concentrations as compared to the early reperfusion phase. The fetus of pregnant women on the brink of delivery, with signs of (imminent) hypoxia may therefore benefit from additional maternal treatment with drugs reducing excessive neurotransmitter formation and formation of free radicals. If necessary, early postnatal hypothermia with or without pharmacologic neuroprotective agents can further reduce or even prevent fetal hypoxic and birth asphyxia-related brain damage of the newborn. Figure 2-2 summarizes schematically our proposed treatment protocol.

REFERENCES OF CHAPTER 2

1. Ferriero DM. Neonatal brain injury. N Engl J Med 2004; Nov 4;351(19):1985-95.

2. Hagberg B, Hagberg G, Olow I. The changing panorama of cerebral palsy in Sweden. VI. Prevalence and origin during the birth year period 1983-1986. Acta Paediatr 1993; Apr;82(4):387-93.

 Low JA. Determining the contribution of asphyxia to brain damage in the neonate. J Obstet Gynaecol Res 2004; Aug;30(4):276-86.

4. Vannucci RC, Perlman JM. Interventions for perinatal hypoxic-ischemic encephalopathy. Pediatrics 1997; Dec;100(6):1004-14.

5. Bennet L, Westgate JA, Gluckman PD, Gunn AJ. Pathophysiology of asphyxia. In: Levene M, Whittle M, Chervenak FA, Punt J, Bennett MJ, editors. Fetal and neonatal neurology and neurosurgery. 3rd ed. London, England: Harcourt Health Sciences; 2001. p. 407.

6. Cunningham F.G., Leveno K.J.', Bloom S.L., Hauth J.C., Gilstrap III L.C., Wenstrom K.D. Williams Obstetrics 2005;1441.

7. Fellman V, Raivio KO. Reperfusion injury as the mechanism of brain damage after perinatal asphyxia. Pediatr Res 1997; May;41(5):599-606.

8. Grow J, Barks JD. Pathogenesis of hypoxic-ischemic cerebral injury in the term infant: current concepts. Clin Perinatol 2002; Dec;29(4):585,602, v.

9. Rosenberg AA, Murdaugh E, White CW. The role of oxygen free radicals in postasphyxia cerebral hypoperfusion in newborn lambs. Pediatr Res 1989; Sep;26(3):215-9.

10. McCord JM. Oxygen-derived free radicals in postischemic tissue injury. N Engl J Med 1985; Jan 17;312(3):159-63.

11. Jenkins DD, Chang E, Singh I. Neuroprotective interventions: is it too late?. J Child Neurol 2009; Sep;24(9):1212-9.

12. van Bel F, Groenendaal F. Long-term pharmacologic neuroprotection after birth asphyxia: where do we stand?. Neonatology 2008;94(3):203-10.

13. Gluckman PD, Wyatt JS, Azzopardi D, Ballard R, Edwards AD, Ferriero DM, et al. Selective head cooling with mild systemic hypothermia after neonatal encephalopathy: multicentre randomized trial. Lancet 2005; Feb 19-25;365(9460):663-70.

14. Shankaran S. Neonatal encephalopathy: treatment with hypothermia. J Neurotrauma 2009; Mar;26(3):437-43.

15. Benders MJ, Bos AF, Rademaker CM, Rijken M, Torrance HL, Groenendaal F, *et al.* Early postnatal allopurinol does not improve short term outcome after severe birth asphyxia. Arch Dis Child Fetal Neonatal Ed 2006; May;91(3):F163-5.

16. Girard S, Kadhim H, Larouche A, Roy M, Gobeil F, Sebire G. Pro-inflammatory disequilibrium of the IL-1 beta/IL-1ra ratio in an experimental model of perinatal brain damages induced by lipopolysaccharide and hypoxia-ischemia. Cytokine 2008; Jul;43(1):54-62.

17. Hilton GD, Nunez JL, Bambrick L, Thompson SM, McCarthy MM. Glutamate-mediated excitotoxicity in neonatal hippocampal neurons is mediated by mGluR-induced release of Ca++ from intracellular stores and is prevented by estradiol. Eur J Neurosci 2006; Dec;24(11):3008-16.

18. Kumar A, Mittal R, Khanna HD, Basu S. Free radical injury and

blood-brain barrier permeability in hypoxic-ischemic encephalopathy. Pediatrics 2008; Sep;122(3):e722-7.

19. Saugstad OD. Role of xanthine oxidase and its inhibitor in hypoxia: reoxygenation injury. Pediatrics 1996; Jul;98(1):103-7.

20. Ferriero DM. Oxidant mechanisms in neonatal hypoxia-ischemia. Dev Neurosci 2001;23(3):198-202.

21. Kirsch JR, Helfaer MA, Lange DG, Traystman RJ. Evidence for free radical mechanisms of brain injury resulting from ischemia/ reperfusion-induced events. J Neurotrauma 1992; Mar;9 Suppl 1:S157-63.

22. Papazisis G, Pourzitaki C, Sardeli C, Lallas A, Amaniti E, Kouvelas D. Deferoxamine decreases the excitatory amino acid levels and improves the histological outcome in the hippocampus of neonatal rats after hypoxia-ischemia. Pharmacol Res 2008; Jan;57(1):73-8.

23. Ogihara T, Hirano K, Ogihara H, Misaki K, Hiroi M, Morinobu T, et al. Non-protein-bound transition metals and hydroxyl radical generation in cerebrospinal fluid of newborn infants with hypoxic ischemic encephalopathy. Pediatr Res 2003; Apr;53(4):594-9.

24. Suzuki M, Tabuchi M, Ikeda M, Tomita T. Concurrent formation of peroxynitrite with the expression of inducible nitric oxide synthase in the brain during middle cerebral artery occlusion and reperfusion in rats. Brain Res 2002; Sep 27;951(1):113-20.

25. van den Tweel ER, Nijboer C, Kavelaars A, Heijnen CJ, Groenendaal F, van Bel F. Expression of nitric oxide synthase isoforms and nitrotyrosine formation after hypoxia-ischemia in the neonatal rat brain. J Neuroimmunol 2005; Oct;167(1-2):64-71.

26. Yang L, Sameshima H, Yamaguchi M, Ikenoue T. Expression of inducible nitric oxide synthase and cyclooxygenase-2 mRNA in brain damage induced by lipopolysaccharide and intermittent hypoxia-ischemia in neonatal rats. J Obstet Gynaecol Res 2005; Apr;31(2):285-91.

27. Northington FJ, Ferriero DM, Graham EM, Traystman RJ, Martin LJ. Early Neurodegeneration after Hypoxia-Ischemia in Neonatal Rat Is Necrosis while Delayed Neuronal Death Is Apoptosis. Neurobiol Dis 2001; Apr;8(2):207-19.

28. Prentice A, Lind T. Fetal heart rate monitoring during labor-too frequent intervention, too little benefit?. Lancet 1987; Dec 12;2(8572):1375-7.

29. MacDonald D, Grant A, Sheridan-Pereira M, Boylan P, Chalmers I. The Dublin randomized controlled trial of intrapartum fetal heart rate monitoring. Am J Obstet Gynecol 1985; Jul 1;152(5):524-39.

30. Nielsen PV, Stigsby B, Nickelsen C, Nim J. Intra- and inter-observer variability in the assessment of intrapartum cardiotocograms. Acta Obstet Gynecol Scand 1987;66(5):421-4.

31. Saling E. Fetal scalp blood analysis. J Perinat Med 1981;9(4):165-77.

32. van den Berg P, Schmidt S, Gesche J, Saling E. Fetal distress and the condition of the newborn using cardiotocography and fetal blood analysis during labor. Br J Obstet Gynaecol 1987; Jan;94(1):72-5.

33. Beard RW, Morris ED, Clayton SG. pH of fetal capillary blood as an indicator of the condition of the foetus. J Obstet Gynaecol Br Commonw 1967; Dec;74(6):812-22.

34. SALING E. Microblood Studies on the Fetus. Clinical Application and 1st Results. Z Geburtshilfe Gynakol 1964; Apr;162:56-75.

35. Zalar RW, Jr, Quilligan EJ. The influence of scalp sampling on the cesarean section rate for fetal distress. Am J Obstet Gynecol 1979; Sep 15;135(2):239-46.

36. Alfirevic Z, Devane D, Gyte GM. Continuous cardiotocography (CTG) as a form of electronic fetal monitoring (EFM) for fetal assessment during labor. Cochrane Database Syst Rev 2006; Jul 19;3:CD006066.

37. Bloom SL, Spong CY, Thom E, Varner MW, Rouse DJ, Weininger S, *et al.* Fetal pulse oximetry and cesarean delivery. N Engl J Med 2006; Nov 23;355(21):2195-202.

38. Garite TJ, Dildy GA, McNamara H, Nageotte MP, Boehm FH, Dellinger EH, et al. A multicenter controlled trial of fetal pulse oximetry in the intrapartum management of nonreassuring fetal heart rate patterns. Am J Obstet Gynecol 2000; Nov;183(5):1049-58.

39. Amer-Wahlin I, Hellsten C, Noren H, Hagberg H, Herbst A, Kjellmer I, et al. Cardiotocography only versus cardiotocography plus ST analysis of fetal electrocardiogram for intrapartum fetal monitoring: a Swedish randomized controlled trial. Lancet 2001; Aug 18;358(9281):534-8.

40. Westerhuis MEHM. Intrapartum fetal monitoring by ST-analysis of the fetal ECG [dissertation]. Enschede, the Netherlands: University Utrecht; 2010.

41. Simpson KR. Intrauterine resuscitation during labor: review of current methods and supportive evidence. J Midwifery Womens Health 2007; May-Jun;52(3):229-37.

42. Hofmeyr GJ, Xu H. Amnioinfusion for meconium-stained liquor in labor. Cochrane Database Syst Rev 2010; Jan 20;(1) (1):CD000014.

43. Fraser WD, Hofmeyr J, Lede R, Faron G, Alexander S, Goffinet F, *et al.* Amnioinfusion for the prevention of the meconium aspiration syndrome. N Engl J Med 2005; Sep 1;353(9):909-17.

44. Xu H, Hofmeyr J, Roy C, Fraser WD. Intrapartum amnioinfusion for meconium-stained amniotic fluid: a systematic review of randomized controlled trials. BJOG 2007; Apr;114(4):383-90.

45. Hofmeyr GJ. What (not) to do before delivery? Prevention of fetal meconium release and its consequences. Early Hum Dev 2009; Oct;85(10):611-5.

46. De Carolis S, De Carolis MP, Caruso A, Oliva GC, Romagnoli C, Ferrazzani S, *et al.* Antenatal phenobarbital in preventing intraventricular hemorrhage in premature newborns. Fetal Ther 1988;3(4):224-9.

47. Asimiadou S, Bittigau P, Felderhoff-Mueser U, Manthey D, Sifringer M, Pesditschek S, *et al.* Protection with estradiol in developmental models of apoptotic neurodegeneration. Ann Neurol 2005; Aug;58(2):266-76.

48. Ikonomidou C, Bosch F, Miksa M, Bittigau P, Vockler J, Dikranian K, *et al.* Blockade of NMDA receptors and apoptotic neurodegeneration in the developing brain. Science 1999; Jan 1;283(5398):70-4.

49. Olney JW. New insights and new issues in developmental neurotoxicology. Neurotoxicology 2002; Dec;23(6):659-68.

50. Wimberley P.D., Lou H.C., Pedersen , Hejl M., Lassen N.A., Friis-Hansen B. Hypertensive peaks in the pathogenesis of intraventricular hemorrhage in the newborn. Abolition by phenobarbitone sedation. Acta Paediatrica Scandinavica 1982;71:537--42. **51.** Goddard-Finegold J, Armstrong DL. Reduction in incidence of periventricular, intraventricular hemorrhages in hypertensive newborn beagles pretreated with phenobarbital. Pediatrics 1987; Jun;79(6):901-6.

52. Steen PA, Michenfelder JD. Mechanisms of barbiturate protection. Anesthesiology 1980; Sep;53(3):183-5.

53. Whitelaw A, Odd D. Postnatal phenobarbital for the prevention of intraventricular hemorrhage in preterm infants. Cochrane Database Syst Rev 2007; Oct 17;(4)(4):CD001691.

54. Donn SM, Roloff DW, Goldstein GW. Prevention of intraventricular haemorrhage in preterm infants by phenobarbitone. A controlled trial. Lancet 1981; Aug 1;2(8240):215-7.

55. Shankaran S, Cepeda EE, Ilagan N, Mariona F, Hassan M, Bhatia R, *et al.* Antenatal phenobarbital for the prevention of neonatal intracerebral hemorrhage. Am J Obstet Gynecol 1986; Jan;154(1):53-7.

56. Shankaran S, Cepeda E, Muran G, Mariona F, Johnson S, Kazzi SN, *et al.* Antenatal phenobarbital therapy and neonatal outcome. I: Effect on intracranial hemorrhage. Pediatrics 1996; May;97(5):644-8.

57. Shankaran S, Woldt E, Nelson J, Bedard M, Delaney-Black V. Antenatal phenobarbital therapy and neonatal outcome. II: Neurodevelopmental outcome at 36 months. Pediatrics 1996; May;97(5):649-52.

58. Kaempf JW, Porreco R, Molina R, Hale K, Pantoja AF, Rosenberg AA. Antenatal phenobarbital for the prevention of periventricular and intraventricular hemorrhage: a double-blind, randomized, placebo-controlled, multihospital trial. J Pediatr 1990; Dec;117(6):933-8.

59. Morales WJ. Antenatal therapy to minimize neonatal intraventricular hemorrhage. Clin Obstet Gynecol 1991; Jun;34(2):328-35.

60. Reinisch JM, Sanders SA, Mortensen EL, Rubin DB. In utero exposure to phenobarbital and intelligence deficits in adult men. JAMA 1995; Nov 15;274(19):1518-25.

61. Ajayi OA, Oyaniyi OT, Chike-Obi UD. Adverse effects of early phenobarbital administration in term newborns with perinatal asphyxia. Trop Med Int Health 1998; Jul;3(7):592-5.

62. Malone JI. Vitamin passage across the placenta. Clin Perinatol 1975; Sep;2(2):295-307.

63. Fiore G, Capasso A. Effects of vitamin E and C on placental oxidative stress: an in vitro evidence for the potential therapeutic or prophylactic treatment of preeclampsia. Med Chem 2008; Nov;4(6):526-30.

64. Poston L, Briley AL, Seed PT, Kelly FJ, Shennan AH, Vitamins in Pre-eclampsia (VIP) Trial Consortium. Vitamin C and vitamin E in pregnant women at risk for pre-eclampsia (VIP trial): randomized placebo-controlled trial. Lancet 2006; Apr 8;367(9517):1145-54.

65. Villar J, Purwar M, Merialdi M, Zavaleta N, Thi Nhu Ngoc N, Anthony J, *et al.* World Health Organisation multicentre randomized trial of supplementation with vitamins C and E among pregnant women at high risk for pre-eclampsia in populations of low nutritional status from developing countries. BJOG 2009; May;116(6):780-8.

66. Warner DS, Sheng H, Batinic-Haberle I. Oxidants, antioxidants and the ischemic brain. J Exp Biol 2004; Aug;207(Pt 18):3221-31.

67. Palmer C, Vannucci RC, Towfighi J. Reduction of perinatal hypoxic-ischemic brain damage with allopurinol. Pediatr Res 1990;

Apr;27(4 Pt 1):332-6.

68. Whitelaw A. Systematic review of therapy after hypoxic-ischaemic brain injury in the perinatal period. Semin Neonatol 2000; Feb;5(1):33-40.

69. Pacher P, Nivorozhkin A, Szabo C. Therapeutic effects of xanthine oxidase inhibitors: renaissance half a century after the discovery of allopurinol. Pharmacol Rev 2006: Mar:58(1):87-114.

70. Van Bel F, Shadid M, Moison RM, Dorrepaal CA, Fontijn J, Monteiro L, et al. Effect of allopurinol on postasphyxial free radical formation, cerebral hemodynamics, and electrical brain activity. Pediatrics 1998; Feb;101(2):185-93.

71. Gunes T, Ozturk MA, Koklu E, Kose K, Gunes I. Effect of allopurinol supplementation on nitric oxide levels in asphyxiated newborns. Pediatr Neurol 2007; Jan;36(1):17-24.

72. Boda D, Németh P, Kiss P, Orvos H. Treatment of mothers with allopurinol to produce therapeutic blood levels in newborns. Prenat Neonat Med 1999;4:130,132, 133, 134.

73. Peeters-Scholte C, Braun K, Koster J, Kops N, Blomgren K, Buonocore G, et al. Effects of allopurinol and deferoxamine on reperfusion injury of the brain in newborn piglets after neonatal hypoxia-ischemia. Pediatr Res 2003; Oct;54(4):516-22.

74. Derks JB, Oudijk MA, Rosén KG, Thakor AS, Visser GHA, van Bel F. et al. Allopurinol maintains umbilical blood flow following repeated episodes of ischaemia-reperfusion in fetal sheep during late gestation. J Soc Gynecol Invest 2006;13(#35).

75. Masaoka N, Nakajima Y, Hayakawa Y, Ohgame S, Hamano S, Nagaishi M, et al. Transplacental effects of allopurinol on suppression of oxygen free radical production in chronically instrumented fetal lamb brains during intermittent umbilical cord occlusion. | Matern Fetal Neonatal Med 2005; Jul;18(1):1-7.

76. van Dijk AJ, Parvizi N, Taverne MA, Fink-Gremmels J. Placental transfer and pharmacokinetics of allopurinol in late pregnant sows and their fetuses. J Vet Pharmacol Ther 2008; Dec;31(6):489-95.

77. Torrance HL, Benders MJ, Derks JB, Rademaker CM, Bos AF, Van Den Berg P, et al. Maternal allopurinol during fetal hypoxia lowers cord blood levels of the brain injury marker S-100B. Pediatrics 2009; Jul;124(1):350-7.

78. Summary of Product Characteristics Acepurin®.

79. Halevy S, Ghislain PD, Mockenhaupt M, Fagot JP, Bouwes Bavinck JN, Sidoroff A, et al. Allopurinol is the most common cause of Stevens-Johnson syndrome and toxic epidermal necrolysis in Europe and Israel. J Am Acad Dermatol 2008; Jan;58(1):25-32.

80. Chaudhari T, McGuire W. Allopurinol for preventing mortality and morbidity in newborn infants with suspected hypoxic-ischaemic encephalopathy. Cochrane Database Syst Rev 2008; Apr 16;(2)(2):CD006817.

81. Maternal treatment with allopurinol diminishes fetal cardiac oxidative stress following repeated episodes of ischaemia-reperfusion in sheep. Proceedings Fetal and Neonatal Physiological Society; august 2007Sendai, Japan.

82. van Kesteren C, Benders MJ, Groenendaal F, van Bel F, Ververs FF, Rademaker CM. Population pharmacokinetics of allopurinol in full-term neonates with perinatal asphyxia. Ther Drug Monit 2006; Jun; 28(3): 339-44.

83. Tamura H, Nakamura Y, Terron MP, Flores LJ, Manchester LC, Tan DX, et al. Melatonin and pregnancy in the human. Reprod Toxicol 2008; Apr;25(3):291-303.

84. Hardeland R, Pandi-Perumal SR, Cardinali DP. Melatonin. Int | Biochem Cell Biol 2006; Mar; 38(3): 313-6.

85. Macleod MR, O'Collins T, Horky LL, Howells DW, Donnan GA. Systematic review and meta-analysis of the efficacy of melatonin in experimental stroke. | Pineal Res 2005; Jan; 38(1): 35-41.

86. Chung SY, Han SH. Melatonin attenuates kainic acid-induced hippocampal neurodegeneration and oxidative stress through microglial inhibition. J Pineal Res 2003; Mar;34(2):95-102.

87. Wakatsuki A, Okatani Y, Izumiya C, Ikenoue N. Melatonin protects against ischemia and reperfusion-induced oxidative lipid and DNA damage in fetal rat brain. J Pineal Res 1999; Apr;26(3):147-52.

88. Okatani Y, Wakatsuki A, Shinohara K, Taniguchi K, Fukaya T. Melatonin protects against oxidative mitochondrial damage induced in rat placenta by ischemia and reperfusion. J Pineal Res 2001; Sep;31(2):173-8.

89. Wakatsuki A, Okatani Y, Shinohara K, Ikenoue N, Fukaya T. Melatonin protects against ischemia/reperfusion-induced oxidative damage to mitochondria in fetal rat brain. J Pineal Res 2001; Sep;31(2):167-72.

90. Hamada F, Watanabe K, Wakatsuki A, Nagai R, Shinohara K, Hayashi Y, et al. Therapeutic Effects of Maternal Melatonin Administration on Ischemia/Reperfusion-Induced Oxidative Cerebral Damage in Neonatal Rats. Neonatology 2009; Dec 3;98(1):33-40.

91. Miller SL, Yan EB, Castillo-Melendez M, Jenkin G, Walker DW. Melatonin provides neuroprotection in the late-gestation fetal sheep brain in response to umbilical cord occlusion. Dev Neurosci 2005; Mar-Aug;27(2-4):200-10.

92. Welin AK, Svedin P, Lapatto R, Sultan B, Hagberg H, Gressens P, et al. Melatonin reduces inflammation and cell death in white matter in the mid-gestation fetal sheep following umbilical cord occlusion. Pediatr Res 2007; Feb;61(2):153-8.

93. Reiter RJ, Paredes SD, Manchester LC, Tan DX. Reducing oxidative/nitrosative stress: a newly-discovered genre for melatonin. Crit Rev Biochem Mol Biol 2009; Jul-Aug;44(4):175-200.

94. Okatani Y, Wakatsuki A, Kaneda C. Melatonin increases activities of glutathione peroxidase and superoxide dismutase in fetal rat brain. J Pineal Res 2000; Mar;28(2):89-96.

95. Hobbs C, Thoresen M, Tucker A, Aquilina K, Chakkarapani E, Dingley J. Xenon and hypothermia combine additively, offering long-term functional and histopathologic neuroprotection after neonatal hypoxia/ischemia. Stroke 2008; Apr;39(4):1307-13.

96. Dingley J, Tooley J, Porter H, Thoresen M. Xenon provides short-term neuroprotection in neonatal rats when administered after hypoxia-ischemia. Stroke 2006; Feb;37(2):501-6.

97. Ma D, Hossain M, Chow A, Arshad M, Battson RM, Sanders RD, et al. Xenon and hypothermia combine to provide neuroprotection from neonatal asphyxia. Ann Neurol 2005; Aug;58(2):182-93.

98. Ma D, Wilhelm S, Maze M, Franks NP. Neuroprotective and neurotoxic properties of the 'inert' gas, xenon. Br J Anaesth 2002; Nov;89(5):739-46.

99. Thoresen M, Hobbs CE, Wood T, Chakkarapani E, Dingley J. Cooling combined with immediate or delayed xenon inhalation provides equivalent long-term neuroprotection after neonatal hypoxia-ischemia. J Cereb Blood Flow Metab 2009; Apr;29(4):707-14.

100. Franks JJ, Horn JL, Janicki PK, Singh G. Halothane, isoflurane, xenon, and nitrous oxide inhibit calcium ATPase pump activity in rat brain synaptic plasma membranes. Anesthesiology 1995; Jan;82(1):108-17.

101. Petzelt C, Blom P, Schmehl W, Muller J, Kox WJ. Prevention of neurotoxicity in hypoxic cortical neurons by the noble gas xenon. Life Sci 2003; Mar 14;72(17):1909-18.

102. Ma D, Hossain M, Pettet GK, Luo Y, Lim T, Akimov S, *et al.* Xenon preconditioning reduces brain damage from neonatal asphyxia in rats. J Cereb Blood Flow Metab 2006; Feb;26(2):199-208.

103. Burov NE, Arzamastsev EV, Kornienko LI, Kudimova LA. Investigation of the teratogenic and embryotoxic action of xenon. Anesteziol Reanimatol 2002; Jul-Aug;(4)(4):69-70.

Chapter 3 Long-term neuroprotective effects of allopurinol after moderate perinatal asphyxia. Follow-up of two randomized controlled trials

JOEPE J. KAANDORP, FRANK VAN BEL, SYLVIA VEEN, JAN B. DERKS, FLORIS GROENENDAAL, MONIQUE RIJKEN, ELISE ROZE, MONICA M.A. UNIKEN VENEMA, CARIN M.A. RADEMAKER, AREND F. BOS, MANON J.N.L. BENDERS

Archives of Diseases in Childhood, Fetal and Neonatal Edition 2012 May; 97(3): F162-6



ABSTRACT

Objective

Free-radical-induced reperfusion injury has been recognized as an important cause of brain tissue damage after birth asphyxia. Allopurinol reduces the formation of free radicals, thereby potentially limiting the amount of hypoxia–reperfusion damage. In this study the long-term outcome of neonatal allopurinol treatment after birth asphyxia was examined.

Design

Follow-up of 4 to 8 years of two earlier performed randomized controlled trials.

Setting

Leiden University Medical Center, University Medical Center Groningen and University Medical Center Utrecht, The Netherlands.

Patients

Fifty-four term infants were included when suffering from moderate-to-severe birth asphyxia in two previously performed trials.

Intervention

Infants either received 40 mg/kg allopurinol (with an interval of 12h) starting within 4h after birth or served as controls.

Main outcome measures

Children, who survived, were assessed with the Wechsler Preschool and Primary Scales of Intelligence test or Wechsler Intelligence Scale for Children and underwent a neurological examination. The effect of allopurinol on severe adverse outcome (defined as mortality or severe disability at the age of 4–8 years) was examined in the total group of asphyxiated infants and in a predefined subgroup of moderately asphyxiated infants (based on the amplitude integrated electroencephalogram).

Results

The mean age during follow-up (n=23) was 5 years and 5 months (SD 1 year and 2 months). There were no differences in long-term outcome between the allopurinol-treated infants and controls. However, subgroup analysis of the moderately asphyxiated group

showed significantly less severe adverse outcome in the allopurinol-treated infants compared with controls (25% vs 65%; RR 0.40, 95%Cl 0.17 to 0.94).

Conclusions

The reported data may suggest a (neuro)protective effect of neonatal allopurinol treatment in moderately asphyxiated infants.

ypoxia-ischemia (HI) during birth asphyxia damages the susceptible developing brain. Reperfusion of previously ischemic brain tissue is now recognized as an important mechanism for substantial additional brain injury due to the formation of toxic free radicals ^[1,2]. This reperfusion injury may be ameliorated by neuroprotective strategies such as hypothermia and/or early post-asphyxial pharmacological intervention like allopurinol ^[3]. Experimental studies investigating specific pharmacological therapies showed promising results ^[4,5], but appeared to be less successful in studies in the human neonate ^[6]. Up to now only therapeutic hypothermia of the brain or the whole body proved to reduce post-HI damage to the brain in moderately asphyxiated human neonates [7-10]. In severely asphyxiated neonates (defined by a severely abnormal amplitude integrated electroencephalogram (aEEG) plus seizures) hypothermia, however, did not seem to significantly improve neurodevelopmental outcome [7].

We previously performed two randomized controlled trials on the effect of high dose allopurinol (40mg/kg/day) for reducing post-HI brain damage in respectively 22 and 32 asphyxiated human neonates ^[11,12]. The xanthine oxidase inhibitor allopurinol potentially protects against reperfusion-induced brain injury by reducing free radical formation and in high dosages also acts as a free radical scavenger and free iron chelator ^[4,13,14]. A recent Cochrane meta-analysis, involving our two studies and a comparable Turkish trial ^[15], concluded that the currently available data are insufficient to determine whether allopurinol is beneficial as a neuroprotective treatment for birth asphyxia-induced HI encephalopathy [6]. However, the neuroprotective effect of allopurinol in the subgroup of moderately asphyxiated children was not yet analyzed.

In this study we investigated neurodevelopmental and cognitive outcome between 4 and 8 years of age in the surviving patients of our two previously performed randomized controlled trials. Because recent studies showed that in particular infants suffering from moderate birth asphyxia benefit from neuroprotection with hypothermia in contrast to those suffering from severe birth asphyxia, we analyzed differences in mortality and long-term developmental outcome between allopurinol-treated and non-treated children not only in the entire group of asphyxiated infants, but also in a subgroup of moderately asphyxiated infants ^[7].

Methods

Participants

The infants included in this follow-up study were the participants in our two randomized controlled trials concerning the effect of early neonatal allopurinol-treatment in reducing post-HI brain damage ^[11,12]. The inclusion criteria of these studies were similar according to the following criteria; (near) term infants, without known chromosomal anomalies, who suffered from birth asphyxia defined as fetal distress (late decelerations on fetal monitoring or meconium staining; the need for resuscitation for > 2 minutes; cord or lowest pH < 7.00 and multi-organ failure).

After admission, electrical brain activity was monitored by aEEG (CFM, Lectromed; Oxford Instruments, Oxford, UK). Amplitude intergraded EEG has been proven to be of value for evaluating background and seizure activity. The type of background pattern predicts longterm outcome ^[16,17]. The following patterns, of increasing abnormality, can be distinguished; continuous normal voltage, discontinuous normal voltage, burst suppression, continuous extremely low voltage and flat trace. The last three patterns, which are abnormal, are induced by hypoxic-ischemic encephalopathy. Van Bel et al. included all children with the above mentioned inclusion criteria, irrespective of the aEEG. In the study by Benders et al. only those children, with the above-mentioned inclusion criteria, which had a burst suppression pattern or worse on the amplitude-integrated electroencephalogram (aEEG) were included.

Intervention

If all inclusion criteria were met, the infants received either two dosages of 20 mg/kg allopurinol (Apurin; Multipharma, Copenhagen, Denmark) intravenously starting within 4 hours after birth (with an interval of 12 hours) or served as controls. Because high dosages were used for the first time, the study by van Bel *et al.* was performed randomized, but not blinded to monitor possible side effect. Benders *et al.* performed the study randomized, double blind and placebo controlled.

Outcome assessment

Children, who survived and could be tested, were assessed with the Wechsler Preschool and Primary Scale of Intelligence (WPPSI-III, an individually administered instrument for assessing the intelligence of children aged 2:6-7:3) ^[18] or the Wechsler Intelligence Scale for Children (WISC-III-NL for children aged 6-17) by psychologists ^[19]. With these tests Verbal, Performance and Full Scale Intelligence Quotients (IQs) were measured. Total IQs were classified according to the guidelines of the Dutch Professional Association of Psychologists; IQ was defined normal if \geq 90, moderately delayed if 70 \leq IQ < 90 and severely delayed if < 70 ^[20]. logical examination by a neonatologist trained in neurodevelopmental assessments. Cerebral Palsy (CP) was classified using the Gross Motor Function Classification System (GMFCS). The GMFCS describes the major functional characteristics of children with CP. It is a five-level pattern-recognition system. Children that are classified as GMFCS levels I and II have the potential to walk independently both indoors and outdoors, and in the community as well. In contrast, children classified in GMFCS levels III to V are limited in their self mobility. They walk with a mobility device and are potential wheelchair users ^[21,22]. The psychologists and neonatologists who assessed neurodevelopmental outcome were independent and not informed about patient allocation.

Neurodevelopmental outcome was classified as normal outcome, mild disability or severe disability. A Full Scale IQ of \geq 90 without any physical abnormalities was defined as normal. Gerebral Palsy classified as GFMCS levels I or II, epilepsy with good response to treatment, hearing impairment and/or 70 \leq IQ < 90 were classified as mild disabilities. Gerebral Palsy classified as GFMCS levels III to V, epilepsy not responding to treatment, blindness, deafness and/or a Full Scale IQ < 70 were regarded as severe disabilities.

The combined frequency of mortality or severe disability in survivors at 4-8 years of age was considered as severe adverse outcome.

Table 3-1. Su	rvival rates of two l Van Bel et al.	RCTs concerning nec	natal allopurinol after birth asphyxia Benders et al.	
	ALLO (n=11), n	CONT (n=11), n	ALLO (n=17), n	CONT (n=15), n
Died	2	6	13	10
Survivors	9	5	4	5

Every child underwent a standardized neuro-

Table 3-2. Baseline Characteristics

	ALLO (n=28)	CONT (n=25)*	Р
Gestational age (wks.days)	40.0 (37.0-41.4)	39.6 (36.0-42.0)	0.253
Birth weight (g)	3500 (2425-5270)	3240 (2300-4720)	0.747
Cord pH	6.90 (6.49-7.05)	6.94 (6.60-7.15)	0.112
Apgar score at 5 minutes	4 (2-7)	4 (1-8)	0.471
aEEG			
CNV, n (%)	8 (29%)	5 (20%)	0.405
BS, n (%)	8 (29%)	12 (48%)	0.371
CLV, n (%)	2 (7%)	4 (16%)	0.414
FT, n (%)	10 (36%)	4 (16%)	0.109

Data are reported as median (range), unless otherwise stated. *One child was excluded because of a suspected syndrome.

ALLO, allopurinol; BS, burst suppression; CLV, continuous low voltage; CNV, continuous normal voltage; CONT, controls; FT, flat trace.

Survival rates and long-term neurodevelopmental outcome were compared in both groups. All infants were then divided in two predefined groups for subgroup analysis based on the aEEG-signal; moderately asphyxiated neonates were defined by having a burst suppression pattern or better on the aEEG and severely asphyxiated neonates were defined by having continuous low voltage or flat trace patterns on the aEEG. The analysis of the tracings was done by two neonatologists (FvB, MB) with expertise in the interpretation of the aEEG. The aEEGs were classified using well-established criteria as published by Toet *et al* ^[16]. This is a clinically used classification system to distinguish moderately asphyxiated infants from severely asphyxiated infants ^[17]. Disagreements were solved by discussion. This subgroup analysis was performed to test the hypothesis that no advantage of neonatal allopurinol treatment is seen when brain damage is too severe.

Statistical analysis

Statistical analyses were performed with the SPSS 15.0 statistical package. Categorical variables were compared by Fisher's exact test. Differences regarding continuous data were assessed by Student's t test or Mann-Whitney U test depending on normality. The analyses concerning severe adverse outcome were stratified by trial. We report separate risk ratios (RR) per trial and a pooled RR with 95% confidence interval (95%CI), generated by using a Mantel Haenszel approach (Review Manager (RevMan) [Computer program]. Version 5.1. Copenhagen: The Nordic Cochrane Centre, The Cochrane Collaboration, 2011). Statistical significance was based on two-sided tests with a cut-off level of 0.05.

Ethical Approval

Both studies were approved by the local ethical committees of the three participating hospitals (the Leiden University Medical Center, the University Medical Center Groningen and the University Medical Center Utrecht [the Netherlands]). Informed consent was given by the parents of all participating children.

Results

Fifty-four full term infants were included in the two previously performed studies; 22 children in the study by van Bel *et al*. and 32 infants in the study by Benders *et al*. Twenty-eight asphyxiated infants received allopurinol (ALLO) and 26 asphyxiated infants served as controls (CONT) (table 3-1).

There were no significant differences between the groups regarding gestational age, birth weight, cord pH, Apgar scores, distribution of aEEG patterns (table 3-2) or the age at follow-up.

In total, 31 infants died in the neonatal period, leaving 23 survivors (13 in the ALLO-group and 10 in the CONT-group) available for long-term follow-up. During the years it became clear

Table 3-3. Mortality and developmental outcome in survivors of two RCTs concerning neonatal allopurinol after birth asphyxia

	Total group of a	Total group of asphyxiated infants			
	ALLO (n=28)	CONT (n=26)			
Mortality, n (%)					
Died	15 (54%)	16 (62%)	0 376		
Survivors	13 (46%)	10 (39%)	0.570		
Excluded	0	1 ⁶			
IQ °	(n=11)	(n=8)			
Verbal	92.4 (13.4)	100.1 (21.3)	0.361		
Performance	94.3 (13.6)	94.0 (16.4)	0.957		
Full Scale	92.8 (13.8)	96.6 (15.7)	0.590		
Long-term neurodevelopmental outcome, n (%)	(n=13)	(n=9)			
Severe disabled ^d	1 (8%)	1 (11%)	1.000		
Mild disabled ^e	5 (39%)	4 (44%)	1.000		
Normal ^f	7 (54%)	4 (44%)	0.680		
Overall long-term outcome, n (%)					
Severe adverse outcome ^g	16 (57%)	17 (68%)	0.571		

^a Subgroup of moderately asphyxiated children; children with a burst suppression pattern or better on the aEEG.

^bExcluded because of suspected syndrome.

 $^{\rm c}$ Values are reported as mean (SD).

^d The two children with a severe disability both had Cerebral Palsy (GMFCS \ge 3).
LONG-TERM NEUROPROTECTIVE EFFECTS OF ALLOPURINOL AFTER MODERATE PERINATAL ASPHYXIA. FOLLOW-UP OF TWO RANDOMIZED CONTROLLED TRIALS

that one child in the CONT-group, with consanguine parents, was suspected to have a syndrome. The infant had mental retardation and epilepsy (not responding to anti-epileptic treatment), which could not be solely explained by perinatal asphyxia since the neonatal MRI was normal. He was therefore excluded from further analyses.

The age of the children at follow-up was 5 years 5 months (SD 1y 2mo) in the ALLO-treated and 5 years 6 months (SD 1y 1mo) in the non-treated children. There were no infants with Full Scale IQ-scores lower than 70 as tested with the WISC or WPPSI. Full Scale IQs of \geq 70 and < 90 were found in seven infants; three in the CONT-group and four in the ALLO-group. Hearing impairment was apparent in one child with normal IQ-scores in the ALLO-group. In the CONT-group one child, with normal IQscores, suffered from epilepsy well responding to treatment. In the ALLO-group one child had developed a spastic quadriplegic Gerebral Palsy classified as GMFCS level V with cortical blind-

Subgroup of moderately asphyxiated infants ^a							
ALLO (n=16)	CONT (n=17)		RR (95%CI)				
4 (25%)	10 (59%)	0.053					
12 (75%)	7 (41%)	0.000					
(11)							
(n=11)	(n=6)	a (a (
92.4 (13.4)	98.8 (17.1)	0.414					
94.3 (13.6)	99.8 (13.9)	0.470					
92.8 (13.8)	98.7 (14.3)	0.429					
(n=12)	(n=7)						
0 (0%)	1 (14%)	0.368					
5 (42%)	2 (29%)	0.656					
7 (58%)	4 (57%)	1.000					
4 (25%)	11 (65%)	0.047	0.40 (0.17-0.94) ^h				

 $^{\rm e}$ The group of mild disabled children included children with epilepsy well responding to treatment, hearing impairment or 70 \leq IQ < 90.

^f Normal outcome: $IQ \ge 90$, no physical impairment.^g Severe adverse outcome was defined as mortality or severe disability.^h Pooled RR, generated by using a Mantel Haenszel approach to stratify by trial.

ALLO, allopurinol; CONT, controls.

LONG-TERM NEUROPROTECTIVE EFFECTS OF ALLOPURINOL AFTER MODERATE PERINATAL ASPHYXIA. FOLLOW-UP OF TWO RANDOMIZED CONTROLLED TRIALS

ness and severe mental retardation. One child in the CONT-group developed a spastic quadriplegic Cerebral Palsy classified as GMFCS level IV. Both children were not tested because severe neurodevelopmental delay was already established and therefore the parents refused to cooperate.

One child in the ALLO-group did not speak Dutch as his first language so the verbal IQ could not be tested reliably. In this specific case the performal IQ, instead of the total IQ, was therefore used to classify outcome.

One child, in the ALLO-group in the study by van Bel *et al.* did not undergo the psychological test. The parents refused to cooperate, because the child did not have neurological problems at 5 years of age and received normal education. At the age of 12 all children in the Netherlands undergo a CITO-test ^[23], which is a standardized Dutch test to assess the intelligence of children in different subareas (e.g. mathematics, language). The outcome is a transformed scale score. The raw score (i.e. the number of questions answered correctly) is converted to a percentile. Individual scores are compared with scores of children of the same age in the Netherlands. This child scored p70, which means 70% of all other children of that age who performed the same test scored lower. Her outcome was therefore classified as "normal".

No significant differences in long-term outcome were seen in children in the ALLO-group compared to the CONT-group (table 3). However, after excluding the most severely asphyxiated children based on the aEEG, significantly less severe adverse outcome was seen in the allopurinol treated infants compared to controls (25% vs 65%; p = 0.047). Separate risk ratios (RR) for severe adverse outcome per trial are shown

Figure 3-1. Severe adverse outcome in two randomized controlled trials concerning neonatal allopurinol after birth asphyxia

	Experimental		Control		M	
Study of Subgroup		Total		Total	vveignt	
Moderate asphyxia						
van Bel 1998	1	9	4	9	21.9%	
Benders 2006	3	7	7	8	35.8%	
Subtotal (95% CI)		16		17	57.7%	
Total events	4		11			
Heterogeneity: Chi ² =0.42, df=1 (P=0.52); l ² =0%						
Test for overall effect: Z=2.1	.0 (P=0.04)					
Total asphyxia		28		26	100%	
Total events	16		17			
Heterogeneity: Chi ² =7.80, df=3 (P=0.05); l ² =62%						
Test for overall effect: 7=0.7	72 (P=0 47)					

Test for subgroup differences: Chi²=6.70, df=1 (P=0.010), l²=85.1%

Occurrence of severe adverse outcome (indicated as 'events') in a subgroup of moderately asphyxiated infants, stratified by trial. A pooled RR for severe adverse outcome in the experimental (allopuin figure 3-1. The pooled RR for severe adverse outcome in the ALLO-group was 0.40 (0.17-0.94) (table 3-3, figure 3-1).

Discussion

Although no significant effect of neonatal allopurinol treatment is seen in reducing mortality or developmental disability in the overall group of asphyxiated infants, the present data suggest that neonatal treatment with allopurinol does lower the risk of severe adverse outcome (i.e. mortality or severe disability) in moderately asphyxiated infants. These findings are in line with the hypothesis that no advantage of neonatal treatment is seen when birth asphyxia is too severe, as stated earlier by Gluckman *et al.* when treating them with moderate hypothermia ^[7]. However, the number of children involved in the analyses is small, so the conclusions drawn have to be taken with caution. There were also some small differences in inclusion criteria between both studies. The study by Benders *et al.* only included children with a burst suppression pattern or worse on the amplitude-integrated electroencephalogram (aEEG). This might explain the differences in overall mortality rate between the two studies, since it probably resulted in the inclusion of more severely asphyxiated infants in the study by Benders *et al.* Furthermore, there was a substantial variation between subjects in follow-up age, but since developmental scores calculated by WPPSI and WISC are adjusted for age, this should not be regarded as a big drawback.

In a true intention-to-treat analysis, the infant who was excluded because of a suspected syndrome should have been included. We nevertheless decided to exclude this child from further analyses. The child was diagnosed with epilepsy not responding to treatment, so would therefore be classified in the "severe adverse out-



rinol) group was generated using a Mantel Haenszel approach.

come^{*}-group. With a small sample size, like in the present study, including this child could have had enormous influence on the final results and conclusions. Since the child made part of the control group, this would have made the results more significantly in favor of allopurinol, which could not be justified.

It must be stated that there are limitations of aEEG interpretation depending on experience, especially in interpreting baseline drift, seizure recognition (with or without seizure detection) and the fact that it is only possible to give a general and not localized assessment of background pattern of brain activity. However, in our opinion, this method to assess the severity of brain damage is reliable in both clinical care and research settings if used by well-trained neonatologists ^[17,24].

To our knowledge, this is the first report concerning follow-up data after neonatal allopurinol treatment and, despite the above-mentioned limitations, we are of the opinion that this follow-up study provides us with some interesting data.

As already stated we could not find a neuroprotective effect of neonatal treatment with allopurinol in the overall group of asphyxiated infants such as is the case with hypothermia. Our study, however, had insufficient power to exclude a modest but important effect size.

Another possible explanation might be the relatively late administration of allopurinol with a median of 3 hours postnatally. Since the vast amount of toxic free radicals, an important reason for post-HI damage to the developing brain is produced during the hypoxic ischemic event itself and in particular upon reperfusion and reoxygenation in the first 30 to 60 minutes after birth, we suggest that earlier treatment might be more effective. By intravenous administration of allopurinol to the mother when fetal distress, an important determinant of perinatal asphyxia, is suspected, therapeutic levels of allopurinol and its active metabolite oxypurinol can be reached even before birth ^[25,26]. In this way, therapeutic levels of allopurinol are already available upon reperfusion, thereby reducing the formation of toxic free radicals in an earlier and crucial stage, namely during maximal free radical formation.

A recent pilot study in this respect, performed by our research group, showed a significant inversed correlation between levels of fetal allopurinol and S100ß (a clinically used marker for brain damage) in cord blood ^[26]. A large multicenter randomized clinical trial on maternal treatment with allopurinol in case of signs of fetal hypoxia is now running in the Netherlands (NCT00189007) to investigate whether or not maternal allopurinol-treatment can further improve long-term outcome of neonates exposed to (severe) fetal hypoxia ^[27].

In conclusion, the effects of neonatal treatment with allopurinol seem promising in moderately asphyxiated infants. Although the window of opportunities is probably quite small it is not clear whether the more severely asphyxiated children can also benefit from allopurinol treatment if treatment is started upon or even before birth in case of (severe) fetal hypoxia. Given the fact that superoxide production occurs upon and during the early reperfusion/reoxygenation phase, it seems to be appropriate to already treat the mother in case of suspected fetal distress during labor to reach therapeutic levels in the newborn before reoxygenation, thereby reducing free radical formation in the earliest possible state. Therapeutic hypothermia did not play any role in this follow-up study, since both initial trials were performed before the introduction of therapeutic mild hypothermia in clinical practice. For future research, however, it is very important to take the impact of mild therapeutic hypothermia on allopurinol metabolism and pharmacokinetics into account.

REFERENCES OF CHAPTER 3

1. Ferriero DM. Neonatal brain injury. N Engl J Med 2004;351:1985-95.

2. McCord JM. Oxygen-derived free radicals in postischemic tissue injury. N Engl J Med 1985;312:159-63.

3. Van Bel F, Groenendaal F. Long-term pharmacologic neuroprotection after birth asphyxia: where do we stand? Neonatology 2008;94:203-10.

4. Palmer C, Towfighi J, Roberts RL, *et al.* Allopurinol administered after inducing hypoxia-ischemia reduces brain injury in 7-day-old rats. Pediatr Res 1993;33:405-11.

5. Fan X, van Bel F. Pharmacological neuroprotection after perinatal asphyxia. J Matern Fetal Neonatal Med 2010;23:17-9.

 Chaudhari T, McGuire W. Allopurinol for preventing mortality and morbidity in newborn infants with suspected hypoxic-ischaemic encephalopathy. Cochrane Database Syst Rev 2008;(2):CD006817.

7. Gluckman PD, Wyatt JS, Azzopardi D, et al. Selective head cooling with mild systemic hypothermia after neonatal encephalopathy: multicentre randomized trial. Lancet 2005;365:663-70.

8. Shankaran S, Laptook AR. Hypothermia as a treatment for birth asphyxia. Clin Obstet Gynecol 2007;50:624-35.

9. Edwards AD, Brocklehurst P, Gunn AJ, et al. Neurological outcomes at 18 months of age after moderate hypothermia for perinatal hypoxic ischaemic encephalopathy: synthesis and meta-analysis of trial data. BMJ 2010;340:c363.

10. Jacobs S, Hunt R, Tarnow-Mordi W, *et al.* Cooling for newborns with hypoxic ischaemic encephalopathy. Cochrane Database Syst Rev 2007;(4):CD003311.

11. Van Bel F, Shadid M, Moison RM, *et al.* Effect of allopurinol on postasphyxial free radical formation, cerebral hemodynamics, and electrical brain activity. Pediatrics 1998;101:185-93.

12 .Benders MJ, Bos AF, Rademaker CM, et al. Early postnatal allopurinol does not improve short term outcome after severe birth asphyxia. Arch Dis Child Fetal Neonatal Ed 2006;91:F163-5.

13. Peeters-Scholte C, van den Tweel E, Groenendaal F, et al. Redox state of near infrared spectroscopy-measured cytochrome aa(3) correlates with delayed cerebral energy failure following perinatal hypoxia-ischaemia in the newborn pig. Exp Brain Res 2004;156:20-6.

14. Russell GA, Cooke RW. Randomized controlled trial of allopurinol prophylaxis in very preterm infants. Arch Dis Child Fetal Neonatal Ed 1995;73:F27-31. **15.** Gunes T, Ozturk MA, Koklu E, *et al.* Effect of allopurinol supplementation on nitric oxide levels in asphyxiated newborns. Pediatr Neurol 2007;36:17-24.

16. Toet MC, Hellström-Westas L, Groenendaal F, *et al.* Amplitude integrated EEG 3 and 6 hours after birth in full term neonates with hypoxic-ischaemic encephalopathy. Arch Dis Child Fetal Neonatal Ed 1999;81:F19-23.

17. Toet MC, Lemmers PM, Schelven LJ, *et al*. Cerebral oxygenation and electrical activity after birth asphyxia: their relation to outcome. Pediatrics. 2006;117:333-9.

 Wechsler D. Wechsler Preschool and Primary Scale of Intelligence - Third Edition (WPPSI-III). San Antonio, TX: PsychCorp, 2002.

19. Van Iterson L, Kaufman AS. Intra-individual subtest variability on the Dutch Wechsler Intelligence Scales for Children-Revised (WISC-R(NL)) for children with learning disabilities, psychiatric disorders, and epilepsy. Psychol Rep 2009;105:995-1008.

20. Resing W, Blok J. De classificatie van intelligentiescores: voorstel voor een eenduidig systeem. De Psycholoog 2002;37:244-8.

21. Van Haastert IC, de Vries LS, Eijsermans MJ, *et al.* Gross motor functional abilities in preterm-born children with cerebral palsy due to periventricular leukomalacia. Dev Med Child Neurol 2008;50:684-9.

22. Palisano R, Rosenbaum P, Walter S, *et al.* Development and reliability of a system to classify gross motor function in children with cerebral palsy. Dev Med Child Neurol 1997;39:214-23.

23. Cito. Mission and Core values. http://www.cito.com/about_ cito/mission_and_core_values.aspx (Accessed September 2011).

24. De Vries LS, Hellström-Westas L. Role of cerebral function monitoring in the newborn. Arch Dis Child Fetal Neonatal Ed. 2005;90:F201-7.

25. Boda D, Németh P, Kiss P, *et al.* Treatment of mothers with allopurinol to produce therapeutic blood levels in newborns. Prenat Neonat Med 1999;4:130-4.

26. Torrance HL, Benders MJ, Derks JB, *et al.* Maternal allopurinol during fetal hypoxia lowers cord blood levels of the brain injury marker S-100B. Pediatrics 2009;124:350-7.

27. Kaandorp JJ, Benders MJ, Rademaker CM, *et al.* Antenatal allopurinol for reduction of birth asphyxia induced brain damage (ALLO-Trial); a randomized double blind placebo controlled multicenter study. BMC Pregnancy Childbirth 2010;10:8.

Chapter 4 Maternal to fetal allopurinol transfer and xanthine oxidase suppression in the late gestation pregnant rat

A.D. KANE, E. CAMM, H. RICHTER, C. LUSBY, D. TIJSSELING, J.J. KAANDORP, J.B. DERKS, S. OZANNE AND D.A. GIUSSANI

Accepted by Physiological Reports



ABSTRACT

Fetal brain hypoxic injury remains a concern in high risk delivery. There is significant clinical interest in agents that may diminish neuronal damage during birth asphyxia, such as in allopurinol, an inhibitor of the pro-oxidant enzyme xanthine oxidase. Here, we established in a rodent model the capacity of allopurinol to be taken up by the mother, cross the placenta, rise to therapeutic levels and suppress xanthine oxidase activity in the fetus. On day 20 of pregnancy, Wistar dams were given 30 or 100 mg.kg¹ allopurinol orally. Maternal and fetal plasma allopurinol and oxypurinol concentrations were measured, and xanthine oxidase activity in the placenta and maternal and fetal tissues determined. There were significant strong positive correlations between maternal and fetal plasma allopurinol (r=0.97, P<0.05) and oxypurinol (r=0.88, P<0.05) levels. Under baseline conditions, maternal heart (2.18±0.62 mU mg⁻¹), maternal liver (0.29 \pm 0.08 mU mg⁻¹), placenta (1.36 \pm 0.42 mU mg⁻¹), fetal heart (1.64 \pm 0.59 mU mg⁻¹) and fetal liver (0.14±0.08 mU mg⁻¹) samples all showed significant xanthine oxidase activity. This activity was suppressed in all tissues 2h after allopurinol administration and remained suppressed 24h later (P<0.05), despite allopurinol and oxypurinol levels returning towards baseline. The data establish a mammalian model of xanthine oxidase inhibition in the mother, placenta and fetus, allowing investigation of the role of xanthine oxidase-derived reactive oxygen species in the maternal, placental and fetal physiology during healthy and complicated pregnancy.

espite advances in obstetric practice, acute intra-partum fetal hypoxia remains one of the most common forms of fetal stress, with substantial morbidity and mortality ^[1]. One possible strategy to combat the detrimental effects of fetal hypoxia would be to decrease the associated excessive generation of reactive oxygen species (ROS) from stimulated pro-oxidant mechanisms within the cell, such as the xanthine oxidase pathway ^[2]. Indeed, maternal treatment with the xanthine oxidase inhibitor allopurinol is being considered in human pregnancy complicated by intra-partum hypoxia in order to protect the infant from excessive generation of ROS ^[3-9]. This clinical interest in antenatal maternal administration of allopurinol follows previous studies which reported that allopurinol treatment in the asphyxiated neonate improved neonatal outcome ^[10], but if the time-interval between hypoxia and treatment had been too long, or when fetal hypoxia had been too severe, no reduction in serious morbidity or mortality was observed ^[4]. Therefore, there has been growing clinical and scientific interest in establishing whether perinatal outcome may be improved in complicated labor if the window of treatment with allopurinol could be initiated before birth, for instance via maternal treatment to cover the actual period of fetal hypoxia and reperfusion ^[7,11].

Whilst there have been studies in large animals including the pregnant ewe ^[11,12] and sow ^[13], no small animal model of allopurinol administration to the mother during pregnancy has been established. Such a model is indispensable in species with a comparatively shorter lifespan to allow follow up of the effects of xanthine oxidase inhibition during pregnancy on the physiology of the offspring in later life. This is essential when many of the conditions associated with a suboptimal fetal environment such as cardiovascular disease and type 2 diabetes only emerge in later life.

Therefore, the aims of this study were to in-

vestigate in rodent pregnancy: 1) whether xanthine oxidase is active in the placenta, maternal and fetal tissues in late gestation; 2) if maternal oral administration of allopurinol could increase maternal plasma levels of allopurinol and its active metabolite oxypurinol; 3) if maternal allopurinol and oxypurinol could cross the placenta, yielding therapeutic levels in the fetal circulation, and 4) whether elevations in maternal and fetal plasma levels of allopurinol and oxypurinol resulted in suppression of xanthine oxidase activity in the placenta, maternal and fetal tissues. These aims were determined using a clinically-relevant dose of allopurinol (30 mg kg⁻¹), comparable to the dose given in previous human studies ^[4,10], and a larger dose (100 mg kg⁻¹) to investigate any pharmacological benefit in terms of either greater fetal levels of allopurinol and/or longer inhibition of xanthine oxidase activity.

Materials and methods

All procedures involving animals were carried out under the Animals (Scientific Procedures) Act 1986 and approved by the Local Ethics Review Committee of the University of Cambridge, UK. Thirty-five time-mated pregnant Wistar rats (Charles River Limited, UK) were delivered to the University of Cambridge between 10-14 days of gestation and were individually housed under standard conditions (21±1°C, 55% humidity, 12h/12h light/dark cycle) with free access to food and water.

On day 20 of pregnancy all animals were randomized to either control or allopurinol treatment. Allopurinol (30mg.kg⁻¹ or 100mg.kg⁻¹, Sigma, UK, suspended in 3ml Hartley's Strawberry Jelly) was placed in a clear glass bowl inside the cage at 8am. Animals had been given an untreated jelly dose the day before to condition them to eat it. Following allopurinol treatment, animals underwent euthanasia (CO₂ followed by cervical dislocation) at either 2h, 6h or 24h following administration (n=5 all time points and doses). A further five animals underwent euthanasia at 8am on day 20 of pregnancy to act as controls. Following euthanasia, a maternal blood sample was taken by cardiac puncture for the measurement of maternal circulating allopurinol and oxypurinol levels. Fetal rats were then exposed by laparotomy. The fetuses were subjected to cervical transection and blood was collected and pooled from all fetuses. Maternal and fetal blood samples were anti-coagulated with ethylenediaminetetraacetic acid (EDTA), centrifuged (3000g for 5 minutes), and the plasma was frozen in liquid nitrogen and subsequently stored at -80°C until analysis for allopurinol and oxypurinol. Samples of placenta, maternal heart, maternal liver, and fetal liver of *ca.* 100mg were taken, weighed and homogenized in buffer (1ml of 100mM TRIS-HCl, pH 7.5 with 10ml protease inhibitor cocktail, SIGMA, P8340). Fetal heart samples were *ca*. 20-25mg. All further analysis was adjusted for the weight of tissue taken. Fetal tissue samples and associated placentas were taken only from male pups to reduce any differences attributable to sex. The sex of each fetal pup was determined by their ano-genital distance as previously described ^[14]. Following centrifugation of the tissue homogenate (30 min, 10 000g at 4°C), the supernatant was decanted and frozen in aliquots at -80°C until further analysis.

Plasma allopurinol and oxypurinol measurements

Allopurinol and oxypurinol plasma concentrations were determined by using reversed-phase, high-performance liquid chromatography with UV-detection at 254 nm for the quantification of allopurinol and oxypurinol in plasma ^[15]. The method was linear between 0.5 and 25 mg.ml⁻¹ with a lower limit of detection of 0.2 mg.ml⁻¹ for both compounds.

Xanthine oxidase assay

Fetal plasma xanthine oxidase activity levels were measured using a commercially available assay kit (A22182, Invitrogen, UK). The assay works on the principle that xanthine oxidase produces superoxide anions. In vitro, superoxide spontaneously degrades to hydrogen peroxide (H_0O_1) . H_0O_2 , in the presence of horseradish peroxidase (HRP), reacts stoichiometrically with Amplex Red reagent to generate the red-fluorescent oxidation product, resorufin. In brief, 50 ml of tissue homogenate or xanthine oxidase standard solution was added to 50 ml of 100 µM Amplex Red reagent solution which also contained 0.4 U.mL⁻¹ HRP and 200 µM hypoxanthine. The resulting solutions were mixed and then incubated for 30 min at 37°C. The presence of resorufin was detected by fluorescence using excitation in the range of 530-560 nm and emission detection at 590 nm. Activity was then expressed per mg of wet tissue.

Data and statistical analysis

All values are expressed as mean ± S.E.M. The correlation between fetal and maternal allopurinol and oxypurinol levels was assessed using the Pearson Product Moment correlation (Sigma-Stat 3.5; Chicago, IL, USA). Allopurinol, oxypurinol and xanthine oxidase activity levels were compared using one-way ANOVA comparing with reference to time oh. Where significant differences were found, the Student-Newman-Keuls *post-hoc* test was applied (Sigma-Stat 3.5; Chicago, IL, USA). Statistical significance was accepted when P<0.05.

Results

Allopurinol and oxypurinol levels in maternal and fetal plasma

Administration of 30 mg.kg⁻¹ and 100 mg.kg⁻¹ allopurinol to the pregnant rat led to increases in both maternal and fetal allopurinol and oxypurinol by 2h (Figure 4-1). Following the administration of allopurinol at both doses of 30 mg.kg⁻¹ and 100 mg.kg⁻¹, allopurinol levels returned to baseline at 6h following dosing, however oxypurinol levels remained significantly elevated in both mother and fetus. Although maternal and fetal plasma oxypurinol levels were detectable 24h following dosing, they were not significantly different from baseline.

When all paired maternal and fetal plasma concentrations of allopurinol and oxypurinol were related to determine maternal to fetal transfer, a significant positive correlation was obtained for both compounds (P<0.05; Figure 4-2). The correlation coefficient (r) for the relationship between maternal and fetal allopurinol was 0.97, and between maternal and fetal oxypurinol was 0.88, indicating a high degree of correlation (Figure 4-2).

Plasma xanthine oxidase levels

Xanthine oxidase activity was detectable in maternal heart, maternal liver, placenta, fetal heart and fetal liver (Figure 3). Maternal and fetal cardiac tissue displayed the greatest activity (2.17±0.6 mU.mg⁻¹ and 1.65±0.59 mU.mg⁻¹, respectively), followed by placenta (1.36±0.41 mU.mg⁻¹) and then maternal liver and fetal liver (0.29±0.09 mU.mg⁻¹ and 0.14±0.08 mU.mg⁻¹). Treatment with allopurinol at 30 mg.kg⁻¹ and 100 mg.kg⁻¹ significantly depressed activity in all tissues by 2 h and this suppression was sustained 6 and 24 h following treatment.

Discussion

Despite intense basic science research and clinical trials investigating the potential neuroprotective effects of the xanthine oxidase inhibitor allopurinol on the fetal brain during and following acute fetal hypoxia [4,7,8,10,13,16,17], no animal model exists that offers the possibility of maternal administration of allopurinol and long term follow up of the treated offspring. Here, we show that allopurinol administered orally to the pregnant rat during late pregnancy leads to therapeutic rises in circulating allopurinol and oxypurinol in the mother and fetus. The increased levels achieved were functionally relevant as demonstrated by significantly reduced, xanthine oxidase activity levels in the placenta, and maternal and fetal tissues following 24h of maternal allopurinol administration. The level



Values are mean \pm SEM for allopurinol (\odot) and oxypurinol (\bullet) concentrations in maternal and fetal plasma (n=5 per time point and group) at oh, 2h, 6h and 24h post administration of (A) 30 mg kg⁻¹, and (B) 100 mg kg⁻¹ of allopurinol to the mother at time oh. One way ANOVA with post-hoc Student-Newman-Keuls where appropriate. Significant differences (P<0.05): *, vs. 0 h control.



points and both 30 mg kg⁻¹ and 100 mg kg⁻¹ allopurinol dosing protocols. N=35 paired samples per analysis. Both relationships show significant positive correlation (P<0.05, Pearson correlation).

of suppression of xanthine oxidase activity was similar with the low and high doses of maternal allopurinol administration.

Currently, human clinical studies based in the Netherlands are assessing whether antenatal allopurinol administration during intra-partum fetal hypoxia can reduce overall morbidity and mortality in the newborn ^[7, 9, 18]. Recent follow-up data from those trials suggests that there may be a reduction in the relative risk of severe disability between 4 and 8 years of age in children who were moderately asphyxiated intra-partum but received postnatal allopurinol treatment starting within 4 h after birth ^[9]. Importantly, postnatal allopurinol was not associated with any negative side effects in the children who were followed up ^[9]. The rationale behind moving clinical therapy from postnatal to antenatal allopurinol administration is that by shortening the period between the beginning of ROS production and the antioxidant cover, the overall tissue injury, and thereby clinical disability, may be reduced. In support of this hypothesis, we have recently reported that maternal antenatal allopurinol administration can reduce hippocampal brain damage in an ovine model of repeated birth asphyxia^[18].

Previous reports have confirmed that maternal treatment with allopurinol crosses the ovine ^[11,12], porcine ^[13] and human ^[19,20] placenta to increase fetal plasma concentrations of allopurinol and oxypurinol. In sheep, a dose of 20 mg kg-1 i.v. to the maternal ewe led to peak maternal allopurinol and oxypurinol concentrations of 47 and 17 mg ml⁻¹ respectively and allopurinol concentrations of approximately 5 mg ml⁻¹ in the fetus ^[11]. In a pregnant sow and piglet model, 15 mg kg⁻¹ allopurinol given intravenously to the mother led to fetal peak concentrations of 5 mg mg⁻¹ ^[13]. In the present study, the plasma levels of allopurinol and oxypurinol reached 3.3 mg mg⁻¹ and 8.5 mg mg⁻¹ respectively in maternal plasma and 2.1 mg mg-1 and 4.0 mg mg⁻¹ respectively in fetal plasma for the 30 mg kg⁻¹ oral dose, values that are slightly lower than those measured in the sheep and pig for doses of similar magnitude. This may reflect species differences in the pharmacokinetics of allopurinol ^[21]. For example, there may be reduced bioavailability of allopurinol and oxypurinol following oral administration relative to other routes of administration and/or differences in drug transfer across the placenta. It is also possible that basal xanthine oxidase activity in the rodent dam and its fetus may be different, altering the rate of conversion of allopurinol to oxypurinol. Finally, differences in the clearance of allopurinol and oxypurinol from the circulation may also exist. Following treatment with 100 mg kg⁻¹ of allopurinol to the mother, fetal values of allopurinol and oxypurinol were 11.6 mg mg⁻¹ and 5.2 mg mg⁻¹ at 2h, respectively. It might have been expected that these values would be approximately three times greater than those recorded following 30 mg kg⁻¹. However, given that xanthine oxidase catalyses the conversion of allopurinol to oxypurinol, and that this reaction is prevented by oxypurinol as the active metabolite, the production of oxypurinol may prevent conversion of allopurinol to oxypurinol, lead-



Values are mean \pm SEM for tissue xanthine oxidase activity in samples of maternal heart and liver, placenta, fetal heart and liver maternal and fetal organ activities of xanthine oxidase assayed *in vitro* at 0, 2, 6 and 24 h following oral administration of allopurinol to the pregnant dam on day 20 of pregnancy (\blacksquare , 0 h control; \blacksquare , 30 mg kg⁻¹; \blacksquare , 100 mg kg⁻¹; n=5 per time point). One way ANOVA with post-hoc Student-Newman-Keuls where appropriate. Significant differences (P<0.05): *, vs. 0 h control.

ing to a relatively high allopurinol concentration and therefore decreasing the rate of further generation of oxypurinol ^[2,22]. There was also a high correlation between paired maternal and fetal concentrations of allopurinol and oxypurinol, independent of time point of dose, indicating that placental transfer is reliable and consistent. However, fetal levels of allopurinol and oxypurinol were lower than maternal, reflected by the coefficient of the straight line relationship being < 1.

Effective inhibition of xanthine oxidase has been previously reported in the adult rat over the range of allopurinol doses from 2-50 mg kg⁻¹ ^[23]. In the sheep fetus, concentrations of allopurinol of 2.25 mg ml⁻¹ achieved a lowering of ROS production ^[12], and 5 mg ml⁻¹ allopurinol increased umbilical blood flow following umbilical cord occlusion ^[11]. In ovine pregnancy, larger doses of allopurinol led to suppression of a1 mediated vasoconstriction in the fetal femoral vascular bed via decreasing ROS mediated NO depletion in the circulation ^[24]. In this study, allopurinol concentrations as low as 2.1 mg ml⁻¹ were sufficient to inhibit xanthine oxidase activity in the fetal rat, with higher doses not offering greater inhibition. Despite the decrease in allopurinol and oxypurinol levels over the 24h period examined in this experiment, xanthine oxidase activity was still suppressed in all tissues investigated. Therefore, doses of allopurinol as low as 30 mg kg⁻¹ administered to the mother are effective to promote the long-term suppression of xanthine oxidase activity in placenta and the fetal organs.

In summary, the current study in rodent pregnancy has established a human clinically relevant model of maternal allopurinol administration that leads to therapeutic concentrations of allopurinol and oxypurinol in the fetal circulation. A single dose of 30 mg kg⁻¹ allopurinol is sufficient to suppress xanthine oxidase activity in the mother, placenta and fetus for at least 24h. The model is indispensable to investigate the long-term consequences of maternal treatment with allopurinol on the physiology of the offspring, and to establish the lasting safety of xanthine oxidase inhibition in human high-risk pregnancy in current clinical obstetric practice.

REFERENCES OF CHAPTER 4

1. Low JA. (2004). Reflections on the occurrence and significance of antepartum fetal asphyxia. Best Practice & Research Clinical Obstetrics & Gynaecology 18, 375-382.

2. Berry C & Hare M. (2004). Xanthine oxidoreductase and cardiovascular disease: molecular mechanisms and pathophysiological implications. J Physiol 555, 589-606.

3. Peeters-Scholte C, Braun K, Koster J, Kops N, Blomgren K, Buonocore G, van Buul-Offers S, Hagberg H, Nicolay K, van Bel F & Groenendaal F. (2003). Effects of allopurinol and deferoxamine on reperfusion injury of the brain in newborn piglets after neonatal hypoxia-ischemia. Pediatr Res 54, 516 - 522.

4. Benders M, Bos A, Rademaker C, Rijken M, Torrance H, Groenendaal F & van Bel F. (2006). Early postnatal allopurinol does not improve short term outcome after severe birth asphyxia. Arch Dis Child Fetal Neonatal Ed 91, 163 - 165.

5. Chaudhari T & McGuire W. (2008). Allopurinol for preventing mortality and morbidity in newborn infants with suspected hypoxic-ischaemic encephalopathy. Cochrane Database Syst Rev 16, CD006817.

6. Torrance HL, Benders MJ, Derks JB, Rademaker CMA, Bos AF, Van Den Berg P, Longini M, Buonocore G, Venegas M, Baquero H, Visser GHA & Van Bel F. (2009b). Maternal Allopurinol During Fetal Hypoxia Lowers Cord Blood Levels of the Brain Injury Marker S-100B. Pediatrics 1224, 350-357.

7. Kaandorp J, Benders M, Rademaker C, Torrance H, Oudijk M, de Haan T, Bloemenkamp K, Rijken M, van Pampus M, Bos A, Porath M, Oetomo S, Willekes C, Gavilanes AD, Wouters M, van Elburg R, Huisjes A, Bakker S, van Meir C, von Lindern J, Boon J, de Boer I, Rijnders R, Jacobs C, Uiterwaal C, Mol BW, Visser G, van Bel F & Derks J. (2010). Antenatal allopurinol for reduction of birth asphyxia induced brain damage (ALLO-Trial); a randomized double blind placebo controlled multicenter study. BMC Pregnancy and Childbirth 10, 8.

 Kaandorp JJ, van Bel F, Veen S, Derks JB, Groenendaal F, Rijken M, Roze E, Venema MMU, Rademaker CM, Bos AF & Benders MJ. (2011b). Long-term neuroprotective effects of allopurinol after moderate perinatal asphyxia: follow-up of two randomized controlled trials. Archives of Disease in Childhood - Fetal and Neonatal Edition.

9. Kaandorp JJ, van Bel F, Veen S, Derks JB, Groenendaal F, Rijken M, Roze E, Venema MMU, Rademaker CM, Bos AF & Benders MJ. (2012). Long-term neuroprotective effects of allopurinol after moderate perinatal asphyxia: follow-up of two randomized controlled trials. Archives of Disease in Childhood - Fetal and Neonatal Edition 97, F162-F166.

10. Van Bel F, Shadid M, Moison R, Dorrepaal C, Fontijn J, Monteiro L, van de Bor & Berger H. (1998). Effect of allopurinol on postasphyxial free radical formation, cerebral hemodynamics, and electrical brain activity. Pediatrics 101, 185 - 193.

11. Derks JB, Oudijk M, Torrance H, Rademaker C, Benders M, Rosen K, Cindrova-Davies T, Thakor AS, Visser G, Burton GJ, van Bel F & Giussani DA. (2010). Allopurinol Reduces Oxidative Stress in the Ovine Fetal Cardiovascular System Following Repeated Episodes of Ischemia-Reperfusion. Pediatric Research 68, 374-380. 12. Masaoka N, Nakajima Y, Hayakawa Y, Ohgame S, Hamano S, Nagaishi M & Yamamoto T. (2005). Transplacental effects of allopurinol on suppression of oxygen free radical production in chronically instrumented fetal lamb brains during intermittent umbilical cord occlusion. The Journal of Maternal-Fetal & amp; Neonatal Medicine 18, 1 - 7.

13. van Dijk A, Parvizil N, Taverne M & Fink-Gremmels J. (2008). Placental transfer and pharmacokinetics of allopurinol in late pregnant sows and their fetuses. Journal of Veterinary Pharmacology and Therapeutics 31, 489-495.

14. Camm EJ, Hansell JA, Kane AD, Herrera EA, Lewis C, Wong S, Morrell NW & Giussani DA. (2010). Partial contributions of developmental hypoxia and undernutrition to prenatal alterations in somatic growth and cardiovascular structure and function. American Journal of Obstetrics and Gynecology 203, 495 Accessed March 497 2013

15. van Kesteren C, Benders M, Groenendaal F, van Bel F, Ververs F & Rademaker C. (2006). Population pharmacokinetics of allopurinol in full-term neonates with perinatal asphyxia. Ther Drug Monit 28, 339 - 344.

16. Boda D, Németh I, Hencz P & Dénes K. (1984). Effect of allopurinol treatment in premature infants with idiopathic respiratory distress syndrome. Dev Pharmacol Ther 7, 357-367.

17. Derks JB, Oudijk M, Rosen K, Thakor AS, Visser G, van Bel F & Giussani DA. (2006). Allopurinol maintains umbilical blood flow following repeated episodes of ischaemia-reperfusion in fetal sheep during late gestation [abstract]. J Soc Gynecol Invest 13, 535.

18. Kaandorp JJ, Derks JB, Oudijk MA, Torrance HL, Harmsen MG, Nikkels PGJ, Bel Fv, Visser GHA & Giussani DA. (2013). Antenatal Allopurinol Reduces Hippocampal Brain Damage After Acute Birth Asphyxia in Late Gestation Fetal Sheep. Reproductive Sciences

19. Boda D, Nemeth P, Kiss P & Orvos H. (1999). Treatment of mothers with allopurinol to produce therapeutic blood levels in newborns. Prenat Neonat Med 4, 130 - 134.

20. Torrance H, Benders M, Derks J, Rademaker C, Bos A, Berg P, Longini M, Buonocore G, Venegas M, Baquero H, Visser G & Van Bel F. (2009a). Maternal allopurinol during fetal hypoxia lowers cord blood levels of the brain injury marker S-100B. Pediatrics 124, 350 - 357.

21. Goodman L, Lazo J, Parker K, Brunton L & Gilman A. (2005). Goodman & Gilman's The Pharmacological Basis of Therapeutics. McGraw-Hill Medical.

22. Parks D & Granger D. (1986). Xanthine oxidase: biochemistry, distribution and physiology. Acta Physiol Scand 548, 87-99.

23. Klein AS, Joh JW, Rangan U, Wang D & Bulkley GB. (1996). Allopurinol: Discrimination of antioxidant from enzyme inhibitory activities. Free Radical Biology and Medicine 21, 713-717.

24. Herrera EA, Kane AD, Hansell JA, Thakor AS, Allison BJ, Niu Y & Giussani DA. (2012). A role for xanthine oxidase in the control of fetal cardiovascular function in late gestation sheep. J Physiol 590, 1825-1837.

Chapter 5 Xanthine oxidase and the fetal cardiovascular defense to hypoxia in late gestation ovine pregnancy

ANDREW D. KANE, JEREMY A. HANSELL, EMILIO A. HERRERA, BETH J. ALLISON, YOUGUO NIU, KIRSTY L. BRAIN, JOEPE J. KAANDORP, JAN B. DERKS, DINO A. GIUSSANI

Accepted by the Journal of Physiology



ABSTRACT

Background

Hypoxia is a common challenge to the fetus, promoting a physiological defense to redistribute blood flow towards the brain and away from peripheral circulation. During acute hypoxia, reactive oxygen species (ROS) interact with nitric oxide (NO) to provide an oxidant tone, which contributes to the mechanisms redistributing the fetal cardiac output, however the source of ROS is unknown. This study investigated whether ROS derived from xanthine oxidase contribute to the fetal peripheral vasoconstrictor response to hypoxia via interaction with NO-dependent mechanisms.

Methods and Results

Fifteen pregnant ewes and their fetuses were surgically prepared for long term recording at 118 days of gestation (term approximately 145 days). After 5 days recovery, mothers were infused i.v. for 30 min with either vehicle, low (30 mg.kg⁻¹) or high (150mg.kg⁻¹) allopurinol with or without fetal NO blockade. Allopurinol inhibited the increase in fetal plasma uric acid and suppressed the fetal femoral vasoconstrictor, glycemic and lactate acidemic responses during hypoxia (all P<0.05), effects that were restored to control levels with fetal NO blockade.

Conclusions

The data provide evidence for the activation of fetal XO in vivo during hypoxia and for XO-derived ROS in contributing to the fetal peripheral vasoconstriction, part of the fetal defense to hypoxia. The data are of significance to the understanding of the physiological control of the fetal cardiovascular system during hypoxic stress. The findings are also of clinical relevance in the context of obstetric trials in which allopurinol is being administered to pregnant women when the fetus shows signs of hypoxic distress.

etal hypoxia can result in marked fetal cardiovascular compromise with subsequent hypoxic-ischemic encephalopathy ^[1], which is associated with cerebral palsy and cognitive disability in later life ^[2] Therefore the prevention and management of fetal hypoxia remain major concerns in obstetric practice today.

The beneficial effects of the xanthine oxidase inhibitor allopurinol in reducing hypoxic damage in adult cardiology and in pediatric and adult cardiothoracic surgery have long been established ^[3-6]. In contrast, the effects of allopurinol in protecting the physiology of the fetus against hypoxia during the developmental period have been less well described. One study reported that treatment with allopurinol of the hypoxic neonate following complicated labor improved neonatal outcome [7]. However, if the time-interval between the hypoxic challenge and treatment was too long, or the hypoxia too severe, no reductions in serious morbidity or mortality were reported ^[8]. Consequently, there has been growing clinical interest in establishing whether perinatal outcome in complicated pregnancy may be improved if the window of treatment with allopurinol is advanced, for instance via maternal treatment to cover the actual period of fetal hypoxia in complicated labor. Maternal treatment with allopurinol crosses the placenta yielding therapeutic levels in the neonatal circulation ^[9] justifying this route of administration for preventative therapy in obstetric practice. However, virtually nothing is known about the effects of maternal treatment with allopurinol on the maternal or fetal physiology, or on maternal and fetal cardiovascular function.

We have discovered that the interaction between the superoxide anion (\cdot O₂-) and nitric oxide (NO) provides an oxidant tone in the fetal vasculature that controls blood flow in several circulations during basal and hypoxic conditions ^[10-13]. Maternal treatment with allo-

purinol under basal conditions increased umbilical blood flow and the gain of the fetal cardiac baroreflex, but it impaired fetal 1 adrenergic mediated pressor and femoral vasopressor responses via increasing NO bioavailability ^[10]. However, the effect of maternal treatment with allopurinol on the fetal cardiovascular defense to acute hypoxia remains unknown. Therefore, in this study we tested the hypothesis that xanthine oxidase has a role in the regulation of fetal cardiovascular function during acute hypoxia. The hypothesis was tested by investigating the in vivo effects of maternal treatment with high and low doses of allopurinol on the fetal cardiovascular responses to hypoxia in the chronically catheterized ewe and fetus during late gestation. To determine whether enhanced NO bioavailability was an involved mechanism mediating the effects of allopurinol on fetal cardiovascular function, maternal treatment with allopurinol was repeated in the presence of fetal in vivo NO blockade with a NO clamp ^[14-16].

Methods

Surgical preparation

Experiments were conducted on pregnant Welsh Mountain ewes using procedures approved by the Local Ethics Review Committee of the University of Cambridge and licensed by the Home Office under the UK Animals (Scientific Procedures) Act, 1986.

Fifteen Welsh Mountain sheep fetuses and their mothers were surgically instrumented under general anesthesia for long-term recording between 118 and 120 days of gestation (term ca. 145 days) using strict aseptic conditions, as described in detail ^[10,17]. Midline abdominal and uterine incisions were made, the fetal hind limbs were exteriorized and the right femoral artery and vein were isolated and catheterized (i.d., 0.86 mm; o.d., 1.52 mm; Critchly Electrical Products, NSW, Australia). A further catheter was anchored onto the fetal hind limb for measurement of amniotic pressure and for administration of antibiotics into the amniotic cavity (600 mg in 2 ml, benzylpenicillin; Crystapen, Schering-Plough, Animal Health Division, Welwyn Garden City, UK). A transonic flow probe was also implanted around the contra-lateral femoral artery (Type 2RS, Transonic Systems Inc.). The dead space of the catheters was filled with heparinized saline (80 i.u. heparin. ml⁻¹ in 0.9% NaCl) and the catheter ends were plugged. A Teflon catheter (i.d. 1.0 mm, o.d. 1.6 mm, Altec, UK) was inserted in the maternal femoral artery and placed in the descending aorta, and a maternal venous catheter placed in the inferior vena cava (i.d., 0.86 mm; o.d., 1.52 mm; Critchly Electrical Products, NSW, Australia). The catheters and flow probe cable were then exteriorized via a keyhole incision in the maternal flank, and placed in a bag sutured to the skin of the ewe. Antibiotics were administered daily to the ewe (0.20-0.25 mg.kg⁻¹ I.M. Depocillin; Mycofarm, Cambridge, UK), to the fetus I.V. and into the amniotic cavity (300 mg Penbritin; SmithKline Beecham Animal Health, Welwyn Garden City, Hertfordshire, UK). Mean fetal arterial blood pressure (corrected for amniotic pressure), femoral blood flow, fetal heart rate (triggered from the arterial blood pressure or femoral blood flow pulse), maternal blood pressure and maternal heart rate (triggered from blood pressure) were continuously recorded using a custom made computerized Data Acquisition System (DAS; Department of Physiology, Development and Neuroscience, Cambridge University, UK).

Experimental protocol

Following 5 days of post-operative recovery, ewes and fetuses were subjected to the acute hypoxia protocol which consisted of 2 h normoxia, 0.5 h hypoxia and 1 h recovery (Figure 5-1). Acute hypoxia in the fetus was induced by maternal inhalational hypoxia ^[17], changing the concentrations of gases breathed by the ewe to 6% O2 in N2 with small amounts of CO2 (15 l.min⁻¹ air, 35 l.min⁻¹ N2, 1.5-2.5 l.min⁻¹ CO2). This mixture was designed to reduce the fetal PaO2 to ca. 10 mmHg whilst maintaining fetal PaCO2. Following the 0.5 h period of hypoxia, the ewe was returned to breathing air for the 1 h recovery period.

Acute hypoxia was induced following maternal I.V. infusion with vehicle (n=11), the low (30 mg.kg⁻¹, n=5) or the high dose of allopurinol (150 mg.kg⁻¹, n=9; Sigma, UK). Allopurinol was dissolved in the minimum volume of 4M sodium hydroxide (NaOH) and made up with saline [10,18] Vehicle was saline treated with 4M NaOH to achieve the same pH of the allopurinol solution ^[10,18] The 30 min infusion period started 20 min following the onset of recording during normoxia. The low dose of allopurinol was adapted from recent studies of allopurinol compatible with human treatment ^{[7,8].} The high dose of allopurinol was 5 times the low dose. The timing of the acute hypoxic challenge coincided with peak concentrations of oxypurinol (the active metabolite of allopurinol) measured in fetal plasma following maternal treatment with a comparable dose of allopurinol in the same breed of sheep ^[19,20].

The experiment with the high dose of allopurinol was repeated following fetal treatment with the NO clamp (n=6). The NO clamp is an established technique that combines fetal treatment with the NO synthase inhibitor NG-nitro-L-arginine methyl ester (L-NAME; 100 mg.kg⁻¹ bolus dissolved in 2 ml saline, fetal i.a; Sigma) with the NO donor sodium nitroprusside (5.1+2.0 μ g.kg⁻¹.min⁻¹, mean+SD dissolved in saline, fetal I.V.; Sigma). The technique blocks de novo synthesis of NO while compensating for the tonic production of the gas and thereby maintaining basal cardiovascular function ^[14-16].

Blood sampling regimen and assays

During any acute hypoxia protocol, descending aortic blood samples (1.0 ml) were taken from the mother and fetus at set time intervals: 0, 50 and 120 min of normoxia, 5, 15 and 30 min of hypoxia, and 15 and 60 min of recovery. Arte-



rial blood gas and acid base status (ABL5 Blood Gas Analyzer, Radiometer; Copenhagen, Denmark; measurements corrected to 39.5 °C for fetal blood and 38 °C for maternal blood), percentage saturation of hemoglobin with oxygen (Sat Hb) and the blood hemoglobin concentration [Hb] (Hemoximeter OSM3, Radiometer; Copenhagen, Denmark), and blood glucose and lactate concentrations (Yellow Springs 2300 Stat Plus Glucose/Lactate Analyzer; YSI Ltd, Farnborough, UK) were determined for each sample. Given xanthine oxidase catalyzes the conversion of hypoxanthine to uric acid, plasma concentrations of urate were also measured using HPLC with electrochemical detection ^[21]. In brief, aliquots of maternal and fetal plasma (acidified 1:1 with ice-cold 10% metaphophoric acid, centrifuged and the supernatant stored at -80°C) were diluted 1:4 with ice-cold 5% metaphophoric acid; final dilution of plasma 1 in 10. To this 100 µl HPLC-grade heptane was added and following vortex mixing for 40 s, the samples were centrifuged (13,000rpm; 5 min) and the lower (aqueous) layer removed and treated with heptane again until the supernatant was clear. This clear supernatant was transferred to a 0.8 ml HPLC vial. An electrochemical detector (EG&G Instruments; Wokingham, Berkshire)

with a working electrode (set at 400 mV and sensitivity of 0.5 µA) was used for detection. Final concentrations for urate were calculated with external standards, which were run simultaneously. The limit of sensitivity for the assay was 0.1 µmol.l⁻¹ for urate, and the inter-assay coefficient of variation was less than 5%. An additional 1 ml of arterial blood was withdrawn at set intervals for fetal plasma catecholamine analyses during acute hypoxia following maternal I.V. infusion with vehicle (n=6) or the high dose of allopurinol (n=6) only to minimize fetal blood loss. These samples were collected under sterile conditions into chilled heparin tubes (2 ml Li+-heparin tubes; LIP Ltd., Shipley, UK) containing reduced glutathione (4 nmol per tube; G-4251; Sigma, UK) and EGTA (5 nmol per tube; E-4378; Sigma, UK). Samples were then centrifuged at 4000 rpm for 4 min at 4°C and stored at -80°C until analysis. Fetal plasma catecholamine concentrations were measured by a commercially available catecholamine radioimmunoassay previously validated for use with sheep plasma (2-CAT RIA, Diasourse) and as previously described in detail^[11].

Data and statistical analyses

Values for blood gas, acid-base and metabolic

status are the mean+S.E.M. for normoxia before (Pre i), during (During i) or after (Post i.) infusion, hypoxia (H, the mean value of 5, 15 and 30 min of hypoxia) and recovery (R, the mean value of 15 and 60 min of recovery). Values for plasma uric acid are the mean+S.E.M. at 0, 50 and 120 min of normoxia, 30 min of hypoxia, and 60 min of recovery. Fetal and maternal cardiovascular variables were recorded continually, and were compiled into line graphs of the mean+S.E.M. for every minute, for all animals. The serial cardiovascular variables were analyzed using summary measures to focus the number of comparisons ^[22]. To determine the effects of allopurinol on basal maternal and fetal cardiovascular function, the mean of minute means were determined before (Pre i, 0-20 min), during (During i, 21-50min) and after (Post i, 51-120 min) infusion. To determine the effects of allopurinol on maternal and fetal cardiovascular function during acute hypoxia, the means ± S.E.M. for the absolute change in the area under the curve (AUC) from normoxic baseline in cardiovascular variables were calculated. These were determined as 30 minute epochs during normoxia (N, 90-120 min), hypoxia (H, 121-150 min) and recovery (R, 151-180 min). For all data, comparisons within (effect of time) and between (effect of treatment) groups were assessed statistically using two-way ANOVA with repeated measures. Where a significant effect of time or treatment was indicated, the post hoc Tukey test was used to isolate the statistical differences. For all comparisons, statistical significance was accepted when P<0.05.

Results

Maternal arterial blood gas, acid base and metabolic status

Pre-infusion values for maternal arterial blood gas, acid base and metabolic status were similar in all ewes and were within the normal range for pregnant Welsh Mountain sheep at this stage of gestation (Table 5-1). Infusion with saline vehicle or the low dose of allopurinol had no effect on arterial blood gas, acid base or metabolic status. In contrast, infusion with the high dose of allopurinol, with or without fetal treatment with the NO clamp, significantly increased maternal pHa for the duration of the protocol. Further, high allopurinol infusion significantly increased maternal acid-base excess (ABE) and decreased maternal PaO2 without affecting maternal hemoglobin saturation with oxygen (Sat Hb; Table 5-1).

In all ewes, acute hypoxia induced significant falls of similar magnitude in maternal PaO2 and Sat Hb without any alteration to PaCO2 (Table 5-1). During recovery, infusion with the high dose of allopurinol, with or without fetal treatment with the NO clamp, maintained the increased maternal pHa. In contrast, all other variables across the groups returned to pre-infusion values.

Fetal arterial blood gas, acid base and metabolic status

Pre-infusion values for fetal arterial blood gas, acid base and metabolic status were similar in all fetuses and were within the normal range for the Welsh Mountain sheep fetus at this stage of gestation (Table 5-2). Infusion with vehicle or allopurinol had no effect on basal arterial blood gas or acid base status. In all fetuses, acute hypoxia induced significant falls of similar magnitude in fetal PaO2 and Sat Hb without any alteration to PaCO₂ (Table 5-2). Acute hypoxia induced a significant decrease in pHa and ABE by the end of the hypoxic challenge in control fetuses only (Table 5-2). In all fetuses, acute hypoxia led to a significant increase in blood lactate. In contrast, a significant increase from baseline in blood glucose during hypoxia only reached significance in the control fetuses and fetuses from mothers treated with the low dose of allopurinol. When blood glucose and lactate were calculated as a change from normoxic baseline, the increments from baseline in blood glucose and lactate were significantly depressed in fetuses from mothers treated with the high

Table 5-1.	Materna	arterial	blood	gas and	acid	base status
------------	---------	----------	-------	---------	------	-------------

			N		н	R
			During i	Post i		
pH ₂	Vehicle	7.53±0.01	7.51±0.01	7.54±0.01	7.52±0.01	7.52±0.01
	Low Allopurinol	7.53±0.01	7.53±0.01	7.52±0.02	7.52±0.03	7.53±0.02
	High Allopurinol	7.50±0.02	7.61±0.02*,†	7.57±0.01*	7.56±0.01*	7.56±0.01*
	High Allopurinol + NO Clamp	7.49±0.01	7.61±0.02*,†	7.57±0.01*	7.57±0.02*	7.57±0.01*
P _a CO ₂	Vehicle	34.5±0.8	35.3±1.3	33.5±0.9	34.8±0.8	34.2±0.8
(mmHg)	Low Allopurinol	35.8±1.6	35.8±1.1	36.8±0.4	35.7±0.8	35.5±1.3
	High Allopurinol	36.1±1.7	33.5±1.1	33.8±0.8	34.6±0.8	34.2±0.7
	High Allopurinol + NO Clamp	37.7±0.8	34.7±0.8	37.0±1.0	34.9±0.9	35.5±1.0
P ₈ O ₂	Vehicle	106±3	100±3	109±4	33±1*	101±2
(mmHg)	Low Allopurinol	104±3	101±4	101±3	35±2*	105±4
	High Allopurinol	105±2	86±5*,†	107±2	37±1*	107±3
	High Allopurinol + NO Clamp	104±2	77±3*,†	100±2	37±3*	102±3
Sat Hb	Vehicle	97.3±0.7	96.5±0.8	96.7±0.7	54.1±2.5*	95.9±0.6
(%)	Low Allopurinol	95.6±0.4	95.5±0.6	95.8±0.5	60.4±3.9*	96.1±0.4
	High Allopurinol	98.5±1.5	92.6±1.3	99.0±1.3	59.4±2.8*	99.1±1.3
	High Allopurinol + NO Clamp	99.2±0.6	97.2±1.2	98.8±1.0	57.8±4.9*	99.1±0.5
ABE (meq.l ⁻¹)	Vehicle	5.8±0.5	5.7±0.4	5.8±0.6	6.0±0.5	5.4±0.5
	Low Allopurinol	6.4±1.3	7.2±1.3	6.8±1.4	7.2±1.4	6.6±1.3
	High Allopurinol	5.9±0.7	11.7±1.3*,†	9.5±1.0*, ⁺	8.8±1.2	8.6±1.1
	High Allopurinol + NO Clamp	6.2±1.4	12.7±1.3*,†	8.8±1.7	8.3±1.6	8.7±1.4
[Glucose]	Vehicle	2.95±0.24	3.08±0.25	3.08±0.24	3.10±0.20	3.19±0.29
(mmol.l ⁻¹)	Low Allopurinol	3.38±0.30	3.69±0.35	3.73±0.41	3.43±0.36	3.54±0.42
	High Allopurinol	2.49±0.18	2.56±0.20	2.53±0.23	2.63±0.24	2.80±0.23
	High Allopurinol + NO Clamp	2.89±0.13	2.91±0.15	2.89±0.16	2.89±0.17	3.31±0.21
[Lactate]	Vehicle	0.39±0.07	0.45±0.08	0.47±0.10	0.56±0.09	0.44±0.05
(mmol.l-1)	Low Allopurinol	0.50±0.11	0.52±0.11	0.54±0.11	0.55±0.11	0.52±0.13
	High Allopurinol	0.44±0.04	0.69±0.12	0.68±0.16	0.95±0.20	0.66±0.15
	High Allopurinol + NO Clamp	0.40±0.03	0.57±0.03	0.56±0.06	0.97±0.23	0.75±0.18

Values represent the means \pm S.E.M. at 0 (Pre i), 50 (During i) and 115 (Post i) min of normoxia, at 30 min of hypoxia (H) and at 60 min of recovery (R) for mothers exposed to 0.5 hour of hypoxia either during saline vehicle infusion (n=11), treatment with the low dose of allopurinol (30 mg.kg⁻¹; n=5), treatment with the high dose allopurinol (150 mg.kg⁻¹; n=9) or treatment with the high dose of allopurinol during fetal blockade of NO synthase (NOS) with the NO clamp (n=6). Significant difference (P<0.05) are: *, within group with respect to time period Pre i. †, between groups with respect to saline vehicle infusion (Two-way ANOVA with post hoc Tukey test).

Table 5-2.	Fetal arterial blood	gas and acid base status
------------	----------------------	--------------------------

			N		н	R
			During i	Post i		
pH ₂	Vehicle	7.36±0.01	7.37±0.01	7.37±0.01	7.30±0.02*	7.26±0.02*
	Low Allopurinol	7.36±0.00	7.36±0.00	7.36±0.01	7.32±0.01	7.30±0.01*
	High Allopurinol	7.36±0.01	7.37±0.01	7.36±0.01	7.32±0.02	7.30±0.02*
	High Allopurinol + NO Clamp	7.33±0.02	7.34±0.01	7.35±0.01	7.35±0.01	7.27±0.03*
P _a CO ₂	Vehicle	53.1±1.5	51.5±1.2	51.5±1.0	53.8±1.2	52.9±1.3
(mmHg)	Low Allopurinol	57.0±1.2	54.3±1.8	55.8±0.9	55.0±1.0	53.5±1.4
	High Allopurinol	53.1±0.8	51.3±1.2	52.8±1.2	52.9±1.0	52.7±0.8
	High Allopurinol + NO Clamp	55.8±0.7	53.0±0.8	53.5±0.9	52.6±1.2	52.6±0.9
P ₈ O ₂	Vehicle	20±1	20±1	20±1	9±1*	20±1
(mmHg)	Low Allopurinol	19±1	19±1	19±1	9±0*	18±1
	High Allopurinol	21±1	19±1	20±1	10±1*	20±1
	High Allopurinol + NO Clamp	22±1	20±1	21±1	11±1*	22±1
Sat Hb	Vehicle	56.4±3.0	52.9±3.3	51.5±2.3	17.3±1.5*	50.8±3.3
(%)	Low Allopurinol	51.3±1.4	50.7±2.5	52.9±2.1	19.1±1.4*	49.4±2.1
	High Allopurinol	55.0±3.2	49.5±3.5	52.5±3.1	20.3±2.0*	52.9±3.1
	High Allopurinol + NO Clamp	58.8±2.9	52.5±3.1	54.0±3.7	21.5±3.6*	54.3±4.4
ABE (meq.l-1)	Vehicle	3.6±0.5	3.4±0.4	3.6±0.5	-1.4±1.2*	-3.8±1.3*
	Low Allopurinol	4.6±0.7	4.2±0.6	4.2±0.6	0.7±1.2	-1.5±1.1*
	High Allopurinol	3.0±0.6	0.9±2.6	3.4±0.5	0.1±1.3	-0.9±1.4*
	High Allopurinol + NO Clamp	1.7±1.0	1.3±0.8	1.7±0.6	-1.3±1.3	-3.0±1.8*
[Glucose]	Vehicle	0.91±0.10	0.97±0.25	0.96±0.09	1.40±0.18*	1.38±0.18*
(mmol.l-1)	Low Allopurinol	1.15±0.12	1.17±0.35	1.16±0.14	1.52±0.23	1.40±0.24
	High Allopurinol	0.97±0.09	0.84±0.20	0.82±0.08	1.09±0.12	1.08±0.16
	High Allopurinol + NO Clamp	0.91±0.08	0.89±0.15	0.88±0.04	1.09±0.12	1.21±0.07
[Lactate]	Vehicle	0.87±0.06	0.87±0.08	1.10±0.06	3.46±0.51*	4.97±0.49*
(mmol.l-1)	Low Allopurinol	1.02±0.11	1.02±0.12	1.07±0.17	2.88±0.59*	3.80±0.51*,†
	High Allopurinol	0.85±0.06	0.98±0.09	1.00±0.06	2.63±0.50*	3.30±0.51*,†
	High Allopurinol + NO Clamp	0.95±0.10	1.16±0.11	1.36±0.10	2.79±0.62*	3.95±0.72*

Values represent the means \pm S.E.M. at 0 (Pre i), 50 (During i) and 115 (Post i) min of normoxia, at 30 min of hypoxia (H) and at 60 min of recovery (R) for fetuses exposed to 0.5 hour of hypoxia either during maternal saline vehicle infusion (n=11), maternal treatment with the low dose of allopurinol (30 mg.kg⁻¹; n=5), maternal treatment with the high dose allopurinol (150 mg.kg⁻¹; n=9) or maternal treatment with the high dose of allopurinol during fetal blockade of NO synthase (NOS) with the NO clamp (n=6). Significant difference (P<0.05) are: *, within group with respect to time period Pre i. +, between groups with respect to saline vehicle infusion (Two-way ANOVA with post hoc Tukey test).

dose of allopurinol relative to control (Δ [glucose]: 0.49±0.15 vs. 0.12±0.13 mmol.l⁻¹, Δ [lactate]: 2.59±0.51 vs. 1.59±0.48 mmol.l⁻¹, P<0.05 for saline vs. high allopurinol). Fetal treatment with the NO clamp during maternal infusion with the high dose of allopurinol restored the increment in fetal blood glucose and lactate towards control levels (Δ [glucose]: 0.29±0.11 mmol.l⁻¹, Δ [lactate]: 1.85±0.61 mmol.l⁻¹).

During recovery, PaO2 and Sat Hb returned to pre-hypoxic levels in all fetuses whilst PaCO2 remained unaltered (Table 5-2). There was a significant decrease in pHa and ABE in all fetuses (Table 5-2). All fetuses continued to show a significant increase in blood lactate during recovery and blood glucose remained significantly elevated from normoxic baseline only in control fetuses (Table 5-2). The increments from baseline in blood glucose and lactate during recovery were again significantly depressed in fetuses from mothers treated with the high dose of allopurinol relative to control (Δ [glucose]: 0.47±0.15 vs. 0.10±0.16mmol.l⁻¹, Δ [lactate]: 4.10±0.50 vs. 2.45±0.48 mmol.l⁻¹, P<0.05 for saline vs. high allopurinol). Fetal treatment with the NO clamp during maternal infusion with the high dose of allopurinol restored the increment in fetal blood glucose but not lactate towards control levels during recovery (Δ [glucose]: 0.31±0.10 mmol.l⁻¹, Δ [lactate]: 3.01±0.71 mmol.l⁻¹, P<0.05 for saline vs. high allopurinol + NO clamp).

Effects of allopurinol on maternal and fetal basal cardiovascular function

Pre-infusion values for maternal arterial blood



Values represent the means \pm S.E.M. calculated every minute for arterial blood pressure and heart rate during 2 hours of normoxia, 0.5 hour of hypoxia (dashed lines) and 1 hour of recovery for mothers either during saline vehicle infusion (n=11), treatment with the low dose of allopurinol (30 mg.kg⁻¹; n=5), treatment with the high dose allopurinol (150 mg.kg⁻¹; n=9) or treatment with the high dose of allopurinol during fetal blockade of NO synthase (NOS) with the NO clamp (n=6). pressure and heart rate were similar in all ewes (Figure 5-2). Infusion with saline or the low dose of allopurinol had no effect on basal maternal cardiovascular function. In contrast, infusion with the high dose of allopurinol led to a significant decrease in maternal basal arterial blood pressure and a significant increase in maternal basal heart rate (Figure 5-2, 5-4A).

Pre-infusion values for fetal arterial blood pressure, heart rate and femoral vascular resistance were similar in all fetuses (Figure 5-3). Maternal infusion with the low or high dose of allopurinol, with or without the NO clamp, significantly decreased basal fetal arterial blood pressure but only infusion with the high dose of allopurinol, with or without fetal treatment with the NO clamp, significantly increased basal fetal heart rate. Allopurinol treatment at either dose did not affect basal fetal femoral blood flow or fetal femoral vascular resistance (Figure 5-3, 5-4B).

Effects of allopurinol on maternal and fetal cardiovascular function during hypoxia

Maternal arterial blood pressure and heart rate increased significantly during acute hypoxia following maternal infusion with vehicle (Fig-



Values represent the means \pm S.E.M. calculated every minute for fetal arterial blood pressure, fetal heart rate, fetal femoral blood flow and fetal femoral vascular resistance during 2 hours of normoxia, 0.5 hour of hypoxia (dashed lines) and 1 hour of recovery for fetuses either during maternal saline vehicle infusion (n=11), maternal treatment with the low dose of allopurinol (30 mg.kg⁻¹; n=5), maternal treatment with the high dose allopurinol (150 mg.kg⁻¹; n=9) or maternal treatment with the high dose of allopurinol during fetal blockade of NO synthase (NOS) with the NO clamp (n=6).



Values are the means \pm S.E.M. for the percent change in mean cardiovascular variables in the mother (A) and in the fetus (B). These were calculated before (Pre i, 0-20 min), during (During i, 21-50 min) and after (Post i, 51-120 min) maternal infusion with saline vehicle (\odot ; n=11), maternal treatment with the low dose of allopurinol (30 mg.kg⁻¹; \odot ; n=5), maternal treatment with the high dose of allopurinol (30 mg.kg⁻¹; \odot ; n=5), maternal treatment with the high dose of allopurinol during fetal blockade of NO synthase (NOS) with the NO clamp (\Box ; n=6). Significant difference (P<0.05) are: *, within group with respect to time period Pre i. **†**, between groups with respect to saline vehicle infusion (Two-way ANOVA with *post hoc* Tukey test).



Values are the means \pm S.E.M. for the absolute change in the area under the curve (AUC) from normoxic baseline in cardiovascular variables in the mother (A) and in the fetus (B). These were calculated as 30 minute epochs during normoxia (N, 90-120 min), hypoxia (H, 121-150 min) and recovery (R, 151-180 min) after maternal infusion with saline vehicle (\odot ; n=11), maternal treatment with the low dose of allopurinol (30 mg.kg⁻¹; \odot ; n=5), maternal treatment with the high dose allopurinol (150 mg.kg⁻¹; \odot ; n=9) or maternal treatment with the high dose of allopurinol during fetal blockade of NO synthase (NOS) with the NO clamp (\Box ; n=6). Significant difference (P<0.05) are: *, within group with respect to time period Pre i. **†**, between groups with respect to saline vehicle infusion (Two-way ANOVA with *post hoc* Tukey test).



ure 5-2, 5-5A). Although treatment with either dose of allopurinol tended to diminish the increment in maternal arterial blood pressure during hypoxia, this failed to reach significance (Figure 5-5A). Both maternal arterial blood pressure and heart rate returned to pre-hypoxic levels during recovery across all groups.

In all groups, acute hypoxia led to a significant increase in fetal arterial blood pressure and femoral vascular resistance and a signifiValues represent the means \pm S.E.M. for fetal plasma adrenaline and noradrenaline at 0 (Pre i), 50 (During i) and 115 (Post i) min of normoxia (N), at 30 min of hypoxia (H) and at 60 min of recovery (R) for fetuses exposed to 0.5 hour of hypoxia either during maternal saline vehicle infusion (white dots, n=5) or maternal treatment with the high dose allopurinol (150 mg.kg⁻¹; black dots, n=5).

cant decrease in fetal heart rate and fetal femoral blood flow (Figure 5-3, 5-5B). The increase in arterial blood pressure and the decrease in heart rate were similar across all groups whilst the increase in femoral vascular resistance was markedly diminished in fetuses treated with maternal infusion with allopurinol (Figure 5-3, 5-5B). Fetal treatment with the NO clamp returned the femoral vascular resistance response to control levels in fetuses from mothers infused with the high dose of allopurinol. The increase



Figure 5-7. Maternal and fetal plasma uric acid concentrations during the hypoxia protocol

Values are the means \pm S.E.M. for maternal and fetal plasma uric acid during the hypoxia protocol after maternal infusion with saline vehicle (\odot ; n=5), maternal treatment with the low dose of allopurinol (30 mg.kg⁻¹; \oplus ; n=5), maternal treatment with the high dose allopurinol (150 mg.kg⁻¹; \oplus ; n=5) or maternal treatment with the high dose of allopurinol during fetal blockade of NO synthase (NOS) with the NO clamp (\Box ; n=5). Significant difference (P<0.05) are: *, within group; **†**, between groups with respect to saline vehicle infusion (Two-way ANOVA with *post hoc* Tukey test). in fetal arterial blood pressure was sustained in all groups during the recovery period (Figure 5-3, 5-5B). In contrast, fetal heart rate, femoral blood flow and femoral vascular resistance returned to pre-hypoxic values.

Fetal plasma catecholamines

Since increases in fetal plasma catecholamines contribute to the femoral vasoconstrictor response to acute hypoxia ^[17], it was of interest to determine if maternal allopurinol affected this response in the fetal circulation. Plasma noradrenaline and adrenaline showed a significant increase from baseline concentrations during acute hypoxia. These elevations in fetal plasma catecholamines were not affected by maternal treatment with the high dose of allopurinol (Figure 5-6).

Maternal and fetal plasma uric acid

Pre-infusion maternal uric acid levels were similar in all groups and were not significantly altered by allopurinol infusion or hypoxia. Fetal plasma uric acid levels were higher than maternal levels during the pre-infusion period. Hypoxia increased fetal plasma uric acid levels and there was a continued increase during the recovery period. In contrast, maternal plasma uric acid remained unaltered from baseline during acute hypoxia. Maternal allopurinol treatment led to a decrease in fetal plasma basal uric acid levels when given at 150 mg.kg⁻¹, with and without the fetal NO clamp. Both low and high doses of maternal allopurinol led to a dose-dependent inhibition of the increase in fetal plasma uric acid measured during acute hypoxia and recovery (Figure 5-7).

Discussion

This study tested the hypothesis that xanthine oxidase has a role in the regulation of fetal cardiovascular function during acute hypoxia. The principle findings of the study show that maternal treatment with allopurinol significantly diminished the rise in fetal plasma uric acid and the fetal femoral vasoconstrictor, hyperglycemic and lactic acidemic responses to acute hypoxia. The effects of maternal allopurinol on fetal femoral vascular resistance, glucose and lactate concentrations in fetal blood were prevented by fetal in vivo NO blockade. Therefore, the data support the hypothesis tested to imply that enhanced NO bioavailability is an involved mechanism mediating the effects of maternal allopurinol treatment on fetal cardiovascular function during acute hypoxia.

The fetal defense to acute hypoxia is largely contingent on fetal cardiovascular responses, which have been well characterized. This fetal cardiovascular defense includes bradycardia and peripheral vasoconstriction ^[23,24]. The latter aids the redistribution of the fetal combined ventricular output away from less essential vascular beds to maintain oxygen and nutrient delivery to the brain; the so-called brain sparing effect ^[25,26]. The physiology underlying this response is also well delineated. The fetal bradycardia and peripheral vasoconstriction are triggered exclusively by a carotid body chemoreflex ^[24,27]. Release of hormones, such as catecholamines, into the fetal circulation maintain the neurally-triggered peripheral vasoconstriction ^[17] and return fetal heart rate back to basal levels, opposing the enhanced vagal tone [24]. The neural and endocrine peripheral vasoconstriction is further fine-tuned by an oxidant tone, created by the interaction between •O²⁻ and NO during acute hypoxia, whereby a fall in the ratio favors dilatation and an increase enhances constriction ^[10-13,15]. Data in the present study show that maternal treatment with allopurinol markedly diminished the fetal peripheral vasoconstrictor response to acute hypoxia without affecting fetal bradycardia and that this hemodynamic effect was prevented by fetal in vivo NO blockade. Therefore, the data in this study support that activation of xanthine oxidase contributes to the femoral vasoconstrictor response during acute hypoxia by altering the peripheral vascular oxidant tone. Hence, maternal treatment with allopurinol shifts the ratio between $\cdot O_2^{-}$ and NO towards dilatation, opposing chemoreflex and endocrine vasoconstrictor influences on the fetal femoral vascular bed. The normalization of the magnitude of the femoral vasoconstrictor response to acute hypoxia in the presence of fetal in vivo NO blockade during maternal treatment with allopurinol confirms this as an involved mechanism.

The increase in fetal blood glucose concentrations during acute hypoxia results from an inhibition in glucose uptake and utilization by peripheral tissues coupled with stimulation of hepatic glucose production ^[28,29]. The fetal lactic acidemia response to acute hypoxia principally arises from the anaerobic metabolism of glucose in hypoxic fetal tissues, in particular the hind limbs in which blood flow is markedly reduced ^[30]. As well as helping in the redistribution of blood flow away from less essential vascular beds towards the fetal brain, the peripheral vasoconstriction also markedly decreases oxygen consumption in the fetus, as the latter is exquisitely coupled to oxygen delivery [30]. Since fetal treatment with phentolamine prevented the glycemic response but enhanced insulin secretion during acute hypoxia ^[31], and since infusion of catecholamines increased glucose output in the sheep fetus ^[32], both the reduction in insulin-dependent glucose uptake and the increase in glucose production by the fetal tissues may be mediated via neural and endocrine adrenergic pathways. Depression of the glycemic response to acute hypoxia following fetal exposure to allopurinol in this study may therefore represent an effect on insulin release and/or on the glucogenic pathways mediated either via the neural sympathetic or plasma amine activities. Indeed, allopurinol has been reported to depresses the hyperglycemia of hemorrhagic shock via similar mechanisms [33]. The depressed circulating lactate concentrations during acute hypoxia in the fetus following exposure to allopurinol in the present study may have resulted from the diminished increased in blood glucose availability

tate by the fetal hind limbs ^[90]. Reversal of the depressive effects of allopurinol on the fetal lactic acidemic response to acute hypoxia following fetal NO blockade may thus result from the restoration towards control levels of the hyper-glycemic response and/or the femoral vasoconstrictor response during acute hypoxia in the fetus. One could argue that inhibition of the fetal pe-

^[34] and/or from the decreased production of lac-

ripheral vasoconstrictor response to acute hypoxia following exposure to XO inhibition may be mediated via depressed chemo-transduction mechanisms within the carotid body and/or due to reduced activation of endocrine constrictor responses, such as the increase in plasma catecholamine levels in the fetus during acute hypoxia. However, in the present study, we further show that maternal treatment with even very high doses of allopurinol did not affect the magnitude of the increase in fetal plasma catecholamine concentrations during acute hypoxia. Inhibition of the fetal femoral vasoconstriction and depression of the fetal glycemic response during acute hypoxia as a consequence of an effect of allopurinol on plasma amine activities is therefore not supported. Similarly, dissociation between a lack of an effect of allopurinol on the fetal bradycardia, which persists during acute hypoxia, and inhibition of the fetal femoral vasoconstriction during the same time period does not support an effect of allopurinol at the level of the carotid chemoreflex, since both fetal bradycardia and fetal femoral vasoconstriction are triggered by the same carotid body chemoreflex ^[24,27]. Rather, these additional findings further support an affect of allopurinol acting to alter the local oxidant tone at the level of the fetal peripheral vasculature. Accordingly, we have reported that fetal treatment with other antioxidants, such as vitamin C, or other agents that increase NO bioavailability, such as statins, has a similar effect on the fetal peripheral vascular oxidant tone, shifting the ratio towards dilatation via NO-dependent pathways, and impairing the redistribution of blood flow away from peripheral circulations during acute hypoxia in the fetus ^[11,13].

A hypotensive effect of allopurinol during basal conditions supports the idea that the cellular oxidant milieu also plays a tonic contribution to peripheral vascular resistance and that, under basal conditions, XO-derived $\cdot O_{n}^{-1}$ is involved in arterial blood pressure maintenance. The mechanisms mediating the tachycardic responses to allopurinol in either the mother or fetus are less clear. The dissociation between the magnitude and timing of the depressor and cardiac responses both suggest that baroreflex activation is an unlikely contributing mechanism increasing heart rate. A more likely explanation is that allopurinol has direct chronotropic effects. Studies have reported that allopurinol increases myocardial contractility [35-37]. We have also reported that the tachycardic response to allopurinol during basal conditions in late gestation fetal sheep can be prevented by fetal treatment with the ß1-adrenergic antagonist atenolol, suggesting that allopurinol may enhance sympathetic influences on the heart.

In summary, data in the present study show that maternal treatment with low and high doses of

allopurinol induces significant effects on maternal and fetal cardiovascular function not only during basal conditions but also in response to acute hypoxia. The data are not only of significance to the understanding of the physiological control of the fetal cardiovascular system during acute hypoxic stress, but they are also of particular clinical relevance in the context of ongoing trials in which allopurinol is being administered to pregnant women when their unborn child shows signs of hypoxic distress [38,39] as highlighted in a recent editorial ^[40]. Collectively, past and present evidence on the effects of allopurinol on the fetus are of mixed clinical implications. Beneficial effects of maternal treatment with allopurinol are supported by its protective effects on umbilical blood flow $^{[10,20]}$, and on the fetal heart $^{[20]}$ and fetal brain $^{[41]}$ during and following periods of ischemia and reperfusion. Detrimental consequences on the fetus of maternal treatment with allopurinol are supported by its effects in impairing fetal peripheral vascular reactivity to constrictor agonists ^[10] and on the fetal cardiovascular defense to acute hypoxia (present work). Clearly, further work is warranted. Until then, the therapeutic use of allopurinol in clinical obstetric practice should be approached with caution.

ACKNOWLEDGMENTS

This work was supported by the British Heart Foundation, the Biotechnology and Biological Sciences Research Council, UK. We are thankful to Mr Scott Gentle and Mrs Sue Nicholls for their invaluable help with the maintenance of the animals. Dino Giussani is the Professor of Cardiovascular Developmental Physiology & Medicine at the Department of Physiology Development & Neuroscience at the University of Cambridge, Professorial Fellow and Director of Studies in Medicine at Gonville & Caius College, a Lister Institute Fellow and a Royal Society Wolfson Research Merit Award Holder.

REFERENCES OF CHAPTER 5

1. Primhak R, Jedeikin R, Ellis G, Makela S, Gillan J, Swyer P, Rowe R. Myocardial ischaemia in asphyxia neonatorum. Electrocardiographic, enzymatic and histological correlations. Acta Paediatr Scand. 1985;74:595-600

2. Cowan F, Rutherford M, Groenendaal F, Eken P, Mercuri E, Bydder GM, Meiners LC, Dubowitz LM, de Vries LS. Origin and timing of brain lesions in term infants with neonatal encephalopathy. Lancet. 2003;361:736-742

3. Johnson W, Kayser K, Brenowitz J, Saedi S. A randomized controlled trial of allopurinol in coronary bypass surgery. Am Heart J. 1991;121:20-24

4. Coghlan J, Flitter W, Clutton S, Panda R, Daly R, Wright G, Ilsley C, Slater T. Allopurinol pretreatment improves postoperative recovery and reduces lipid peroxidation in patients undergoing coronary artery bypass grafting. J Thorac Cardiovasc Surg. 1994;107:248-256

5. Castelli P, Condemi A, Brambillasca C, Fundarò P, Botta M, Lemma M, P V, Santoli C, Gatti S, Riva E. Improvement of cardiac function by allopurinol in patients undergoing cardiac surgery. J Cardiovasc Pharmacol. 1995;25:119-125

6. Clancy RR, McGaurn SA, Goin JE, Hirtz DG, Norwood WI, Gaynor JW, Jacobs ML, Wernovsky G, Mahle WT, Murphy JD, Nicolson SC, Steven JM, Spray TL. Allopurinol neurocardiac protection trial in infants undergoing heart surgery using deep hypothermic circulatory arrest. Pediatrics. 2001;108:61-70

7. Van Bel F, Shadid M, Moison R, Dorrepaal C, Fontijn J, Monteiro L, van de Bor, Berger H. Effect of allopurinol on postasphyxial free radical formation, cerebral hemodynamics, and electrical brain activity. Pediatrics. 1998;101:185 - 193

8. Benders M, Bos A, Rademaker C, Rijken M, Torrance H, Groenendaal F, van Bel F. Early postnatal allopurinol does not improve short term outcome after severe birth asphyxia. Arch Dis Child Fetal Neonatal Ed. 2006;91:163 - 165

9. Boda D, Nemeth P, Kiss P, Orvos H. Treatment of mothers with allopurinol to produce therapeutic blood levels in newborns. Prenat Neonat Med. 1999;4:130 - 134

10. Herrera EA, Kane AD, Hansell JA, Thakor AS, Allison BJ, Niu Y, Giussani DA. A role for xanthine oxidase in the control of fetal cardiovascular function in late gestation sheep. J Physiol. 2012;590:1825-1837

11. Kane AD, Herrera EA, Hansell JA, Giussani DA. Statin treatment depresses the fetal defence to acute hypoxia via increasing nitric oxide bioavailability. J Physiol. 2012;590:323-334

12. Thakor AS, Herrera EA, Serón-Ferré M, Giussani DA. Melatonin and vitamin c increase umbilical blood flow via nitric oxide-dependent mechanisms. Journal of Pineal Research. 2010;49:399-406

13. Thakor AS, Richter HG, Kane AD, Dunster C, Kelly FJ, Poston L, Giussani DA. Redox modulation of the fetal cardiovascular defence to hypoxaemia. J Physiol. 2010;588:4235-4247

14. Gardner DS, Giussani DA. Enhanced umbilical blood flow during acute hypoxemia after chronic umbilical cord compression: A role for nitric oxide. Circulation. 2003;108:331-335

15. Morrison S, Gardner DS, Fletcher AJ, Bloomfield MR, Giussani DA. Enhanced nitric oxide activity offsets peripheral vasoconstriction during acute hypoxaemia via chemoreflex and

adrenomedullary actions in the sheep fetus. The Journal of Physiology. 2003;547:283-291

16. Thakor A, Giussani D. Nitric oxide reduces vagal baroreflex sensitivity in the late gestation fetus. Pediatr Res. 2009;65:269-273

17. Fletcher AJ, Edwards CM, Gardner DS, Fowden AL, Giussani DA. Neuropeptide y in the sheep fetus: Effects of acute hypoxemia and dexamethasone during late gestation. Endocrinology. 2000;141:3976-3982

18. van Dijk A, Parvizil N, Taverne M, Fink-Gremmels J. Placental transfer and pharmacokinetics of allopurinol in late pregnant sows and their fetuses. Journal of Veterinary Pharmacology and Therapeutics. 2008;31:489-495

19. Masaoka N, Nakajima Y, Hayakawa Y, Ohgame S, Hamano S, Nagaishi M, Yamamoto T. Transplacental effects of allopurinol on suppression of oxygen free radical production in chronically instrumented fetal lamb brains during intermittent umbilical cord occlusion. The Journal of Maternal-Fetal & amp; Neonatal Medicine. 2005;18:1 - 7

20. Derks J, Oudijk M, Torrance H, Rademaker C, Benders M, Rosen K, Cindrova-Davies T, Thakor A, Visser G, Burton G, van Bel F, Giussani D. Allopurinol reduces oxidative stress in the ovine fetal cardiovascular system following repeated episodes of ischemia-reperfusion. Pediatric Research. 2010;68:374-380

21. Iriyama K, Yoshiura M, Iwamoto T, Ozaki Y. Simultaneous determination of uric and ascorbic acids in human serum by reversed-phase high-performance liquid chromatography with electrochemical detection. Analytical Biochemistry. 1984;141:238-243

22. Matthews J, Altman D, Campbell M, Royston P. Analysis of serial measurements in medical research. BMJ. 1990;300:230-235

23. Cohn HE, Sacks EJ, Heymann MA, Rudolph AM. Cardiovascular responses to hypoxemia and acidemia in fetal lambs. Am J Obstet Gynecol. 1974;120:817-824

24. Giussani DA, Spencer JA, Moore PJ, Bennet L, Hanson MA. Afferent and efferent components of the cardiovascular reflex responses to acute hypoxia in term fetal sheep. J Physiol. 1993;461:431-449

25. Giussani DA, Davidge ST. Developmental programming of cardiovascular disease by prenatal hypoxia. Journal of Developmental Origins of Health and Disease. 2013.

26. Rudolph A. Distribution and regulation of blood flow in the fetal and neonatal lamb. Circ Res. 1985;57:811-821

27. Bartelds B, van Bel F, Teitel D, Rudolph A. Carotid, not aortic, chemoreceptors mediate the fetal cardiovascular response to acute hypoxemia in lambs. Pediatr Res. 1993;34:51-55

28. Jones CT. The development of some metabolic responses to hypoxia in the fetal sheep. The Journal of Physiology. 1977;265:743-762

29. Jones CT, Ritchie J. The effects of adrenergic blockade on fetal response to hypoxia. J Dev Physiol. 1983;5:211-222

30. Boyle D, Hirst K, Zerbe G, Meschia G, Wilkening R. Fetal hind limb oxygen consumption and blood flow during acute graded hypoxia. Pediatr Res. 1990;28:94-100

31. Jones C, Ritchie J, Walker D. The effects of hypoxia on glucose turnover in the fetal sheep. J Dev Physiol. 1983;5:223-235

32. Apatu RS, Barnes RJ. Release of glucose from the liver of fetal and postnatal sheep by portal vein infusion of catecholamines or glucagon. The Journal of Physiology. 1991;436:449-468

33. Salles M, Tabatabai M, Shahidi H. Blood glucose levels in dogs pretreated with allopurinol during hemorrhagic shock. Am J Physiol. 1972;223:679-681

34. Lawrence GF, Brown VA, Parsons RJ, Cooke ID. Feto-maternal consequences of high-dose glucose infusion during labor. Br J Obstet Gynaecol. 1982;89:27-32

35. Ekelund UEG, Harrison RW, Shokek O, Thakkar RN, Tunin RS, Senzaki H, Kass DA, Marbán E, Hare JM. Intravenous allopurinol decreases myocardial oxygen consumption and increases mechanical efficiency in dogs with pacing-induced heart failure. Circulation Research. 1999;85:437-445

36. Cappola TP, Kass DA, Nelson GS, Berger RD, Rosas GO, Kobeissi ZA, Marbán E, Hare JM. Allopurinol improves myocardial efficiency in patients with idiopathic dilated cardiomyopathy. Circulation. 2001;104:2407-2411

37. Kögler H, Fraser H, McCune S, Altschuld R, Marbán E. Disproportionate enhancement of myocardial contractility by the xanthine oxidase inhibitor oxypurinol in failing rat myocardium. Cardiovascular Research. 2003;59:582-592 **38.** Kaandorp J, Benders M, Rademaker C, Torrance H, Oudijk M, de Haan T, Bloemenkamp K, Rijken M, van Pampus M, Bos A, Porath M, Oetomo S, Willekes C, Gavilanes AD, Wouters M, van Elburg R, Huisjes A, Bakker S, van Meir C, von Lindern J, Boon J, de Boer I, Rijnders R, Jacobs C, Uiterwaal C, Mol BW, Visser G, van Bel F, Derks J. Antenatal allopurinol for reduction of birth asphyxia induced brain damage (allo-trial); a randomized double blind placebo controlled multicenter study. BMC Pregnancy and Childbirth. 2010;10:8

39. Kaandorp JJ, van Bel F, Veen S, Derks JB, Groenendaal F, Rijken M, Roze E, Venema MMU, Rademaker CM, Bos AF, Benders MJ. Long-term neuroprotective effects of allopurinol after moderate perinatal asphyxia: Follow-up of two randomized controlled trials. Archives of Disease in Childhood - Fetal and Neonatal Edition. 2012;97:F162-F166

40. Peebles D. Radical change for the fetus. The Journal of Physiology. 2012;590:1773

41. Kaandorp JJ, Derks JB, Oudijk MA, Torrance HL, Harmsen MG, Nikkels PGJ, Bel Fv, Visser GHA, Giussani DA. Antenatal allopurinol reduces hippocampal brain damage after acute birth asphyxia in late gestation fetal sheep. Reproductive Sciences. 2013

Chapter 6 Antenatal allopurinol reduces hippocampal brain damage after acute birth asphyxia in late gestation fetal sheep

JOEPE J. KAANDORP, JAN B. DERKS, MARTIJN A. OUDIJK, HELEN L. TORRANCE, MARLINE G. HARMSEN, PETER G. J. NIKKELS, FRANK VAN BEL, GERARD H. A. VISSER, DINO A. GIUSSANI

Reproductive Sciences, 2013 Jun 21. [Epub ahead of print]



ABSTRACT

Free radical induced reperfusion injury is a recognized cause of brain damage in the newborn after birth asphyxia. The xanthine-oxidase inhibitor allopurinol reduces free radical synthesis and crosses the placenta easily. Therefore, allopurinol is a promising therapeutic candidate. This study tested the hypothesis that maternal treatment with allopurinol during fetal asphyxia limits ischemia-reperfusion (I/R) damage to the fetal brain in ovine pregnancy. The I/R challenge was induced by 5 repeated measured compressions of the umbilical cord each lasting 10 min, in chronically-instrumented fetal sheep at 0.8 of gestation. Relative to control fetal brains, the I/R challenge induced significant neuronal damage in the fetal hippocampal cornu ammonis zone 3 and 4. Maternal treatment with allopurinol during the I/R challenge restored the fetal neuronal damage towards control scores. Maternal treatment with allopurinol offers potential neuroprotection to the fetal brain in the clinical management of perinatal asphyxia.
uring labor and delivery, the most common challenge to the fetus is repeated compression of the umbilical cord. If severe or prolonged enough, this may result in perinatal asphyxia, leading to marked fetal acidosis and cardiovascular compromise with subsequent hypoxic-ischemic encephalopathy (HIE), predictive for future cerebral palsy and cognitive disability ^[1]. The incidence of severe perinatal asphyxia is estimated 4-9 per 1000 full-term neonates, of whom 15-20% will die and about 25% of the survivors will permanently suffer from neuropsychological deficits ^[2].

To date, the only established treatment for HIE is therapeutic hypothermia of the neonate ^[3,4]. However, only mild to moderately asphyxiated infants seem to significantly benefit and only a small reduction of death or disability is seen ^[3,5,6]. Therefore, there is a great need for additional treatment options to help prevent or diminish cerebral damage in the infant due to perinatal asphyxia.

Repeated compressions of the umbilical cord not only induce fetal asphyxia and acidosis but also episodes of ischemia-reperfusion (I/R), promoting the excessive generation of reactive oxygen species (ROS) such as the superoxide (O_2^{-1}) and hydroxyl (OH) anions ^[7-9]. Several potential antioxidant pharmacologic options to intervene in the putative neurotoxic cascade have already been studied ^[6,10]. Drugs like tetrahydrobiopterin, melatonin, nNOS inhibitors, xenon and vitamin C showed promising results in experimental studies, but have not yet been translated to clinical use ^[6,11].

One of the main pro-oxidant pathways stimulated by I/R is the xanthine oxidase pathway ^[12,13]. Allopurinol, a xanthine oxidase inhibitor and FDA-approved drug, inhibits the conversion of hypoxanthine into xanthine and uric acid, thereby limiting the toxic overproduction of ROS. In high doses, it also functions as a chelator of Non Protein Bound Iron (NPBI) and a direct scavenger of the hydroxyl radical ^[14]. Furthermore, substantial knowledge on the use of allopurinol in terms of dose and in vivo safety profiles is already available ^[15-21]. Collectively, these properties make allopurinol a strong potential candidate for cerebral neuroprotection arising from I/R damage.

A recent follow-up study of two prospective randomized controlled trials determining the effects of postnatally administered allopurinol in term asphyxiated human neonates showed a beneficial effect of treatment on mortality and severe disabilities at 4-8 years of age, but only in the moderately asphyxiated group [20]. It was further reported that treatment with allopurinol of the asphyxiated neonate improved neonatal outcome ^[15], however if the time-interval between I/R and treatment had been prolonged, or again when asphyxia had been too severe, no reduction in serious morbidity or mortality was observed ^[16]. Therefore, there has been accumulating interest in establishing whether perinatal cerebral outcome may be improved if the window of treatment with allopurinol is advanced, for instance via maternal treatment to cover the actual period of fetal asphyxia and of I/R in complicated labor. Maternal treatment with allopurinol crosses the placenta, it suppresses O_ production in the fetus ^[22] and yields therapeutic levels in the neonatal circulation [23], justifying this route of administration for preventative therapy in obstetric practice.

Therefore, this study tested the hypothesis that maternal treatment with allopurinol during repeated episodes of fetal asphyxia would limit I/R damage to the fetal brain. The hypothesis was tested in chronically instrumented fetal sheep in late gestation subjected to an I/R challenge involving repeated, measured compression of the umbilical cord with or without maternal treatment with allopurinol.

Materials and Methods

Ethical Approval

The study was approved by the Cambridge University Ethical Review Committee. All procedures were performed under the UK Animals (Scientific Procedures) 1986 Act and conducted under the authority of the appropriate project and personal licenses.

Animals and Surgical Procedures

Eleven Welsh Mountain Sheep were surgically instrumented for long-term recording at 124 days of gestation (term is ~ 145 days) using strict aseptic conditions, as previously described in detail ^[19,24,25]. Under general anesthesia (1.5-2.0% halothane in 50:50 O₂:N₂O), midline abdominal and uterine incisions were made, the fetal hind limbs were exteriorized and, on one side, fetal arterial (i.d., 0.86 mm; o.d., 1.52 mm; Critchly Electrical Products, NSW, Australia) and venous (i.d., 0.56 mm; o.d., 0.96 mm) catheters were inserted. Another catheter was anchored onto the fetal hind limb for recording of the reference amniotic pressure. In addition, a transit-time flow transducer was implanted around the left umbilical artery close to the common umbilical artery inside the fetal abdominal cavity (4SB; Transonic Systems Inc., Ithaca, NY, USA) [24,26]. An inflatable occluder cuff (In Vivo Metrics) was positioned around the proximal end of the umbilical cord, as described previously in detail. ^[19,26] Ewes were instrumented with arterial and venous catheters placed in the left femoral artery and vein, respectively. All incisions were closed in layers. The catheters, occluder cable and flow probe leads were then exteriorized via a keyhole incision in the maternal flank and kept inside a plastic pouch sewn onto the maternal skin.

Post-Operative Care and Experimental Protocol

Antibiotics were administered daily to the ewe (0.20-0.25 mg.kg⁻¹ i.m. Depocillin; Mycofarm, Cambridge, UK) and fetus i.v. and into the am-

niotic cavity (150 mg.kg⁻¹ Penbritin; SmithKline Beecham Animal Health, Welwyn Garden City, Hertfordshire, UK). Following at least five days of post-operative recovery, all fetuses were submitted to an I/R challenge, produced by 5 x 10 minutes inflations of the cord occluder with sterile saline at 10-minute intervals. Each cord compression was designed to reduce umbilical blood flow by 80-90% from baseline, and to lead to a progressive fall in fetal arterial pH to 6.9 by the end of the fifth compression. In five fetuses, the I/R challenge was induced during maternal I.V. treatment with allopurinol (Sigma Ltd., 20 mg.kg⁻¹ maternal weight, dissolved in buffered saline and infused over a twenty minute period). In the remaining six fetuses, the I/R challenge was induced during maternal infusion with buffered saline at the same rate. Infusion of either allopurinol or vehicle started 10 minutes before the fourth umbilical cord compression and finished immediately after the end of it. The dosing regimen of allopurinol was adopted from the only study that used the drug in women undergoing uncomplicated labor^[23].

Blood sampling regimen

To determine arterial blood gas, acid base and metabolic status, maternal and fetal arterial blood samples (0.3 mL) were drawn into sterile syringes one hour prior to the I/R challenge, at 10 minute intervals during and for 48 hours following the I/R challenge (ABL5 Blood Gas Analyzer, Radiometer, Copenhagen, Denmark; see Figure 6-1). Values for percentage saturation of hemoglobin with oxygen (SatHb) were determined using a hemoximeter (OSM3; Radiometer). Blood glucose and lactate concentrations were measured by an automated analyser (Yellow Springs 2300 Stat Plus; YSI Ltd., Farnborough, UK). Additional paired maternal and fetal blood samples (1 mL) were taken in the allopurinol treated pregnancies at varying set intervals, starting at the onset of the infusion period and up to five hours following the end of infusion, to compile a comprehensive serial profile of maternal and fetal plasma concentrations of allopurinol and oxypurinol without affecting materno-fetal concentrations of hemoglobin. Reversed-phase high-performance liquid chromatography (HPLC) with UV-detection at 254 nm was used for the quantification of allopurinol and oxypurinol in both the fetal and maternal plasma. The method was linear between 0.5 and 25 mg.L⁻¹ with a lower limit of detection of 0.2 mg.L⁻¹ for both compounds ^[27].

Histological Evaluation

Forty-eight hours after the end of the experi-



At least 5 days after surgery, all fetuses were subjected to an ischemia-reperfusion (I/R) challenge (gray bar). After 1 hour of basal recording, the I/R challenge consisted of 5 compressions of the umbilical cord, each of 10 minutes' duration (black bars) with a 10-minute interval. Maternal infusion with allopurinol or vehicle started 10 minutes before the fourth umbilical cord compression and finished immediately after the end of it (white bar). Fetal arterial blood samples (arrows) were taken for analysis of blood gas and metabolic status. B, Data are umbilical blood flow (UBF) mean+standard error of the mean (SEM), before and after each umbilical cord compression in 5 I/R allopurinol pregnancies (black bars) and in 6 I/R vehicle pregnancies (white bars). Significant differences *P < .05 versus baseline (2-way repeated measures analysis of variance with t Newman-Keuls test).

ANTENATAL ALLOPURINOL REDUCES HIPPOCAMPAL BRAIN DAMAGE AFTER ACUTE BIRTH ASPHYXIA IN LATE GESTATION FETAL SHEEP

mental protocol, ewes and fetuses were subjected to humane euthanasia using a lethal dose of sodium pentobarbitone (200 mg.kg⁻¹ I.V. Pentoject; Animal Ltd., York, UK) for tissue collection. Tissues were also harvested from 6 un-instrumented fetal sheep at 0.8 gestation, which served as age-matched controls. Immediately after euthanasia, a Gaesarean section was performed and fetal brains were perfusion-fixed with formaldehyde delivered through the carotid arteries under constant pressure. Fixed brains were embedded in paraffin. Four-micrometer coronal sections were cut at the level of the dorsal hippocampus, caudate nucleus and cerebellum and stained with H and E and a combined staining procedure using acid fuchsin-thionin (0.1% fuchsin, Gurr/BDH; 0.25% thionin, Chroma). The latter stain highlights (acidophilic) damaged neurons; these neurons have a bright pink cytoplasm on a blue Nissl background. Gells were considered to be necrotic if they had acidophilic cytoplasm, nuclear pyknosis or loss of nuclear detail. The proportion of necrotic cells was determined in the hippocampus (Cornu Ammonis (CA), areas 1+2 and 3+4, dentate gyrus and parahippocampal cortex), thalamus and caudate nucleus by light microscopy (Zeiss standard 25, Germany) at 100x and 400x magnification. Examination was performed using a 6-point scale: 0 (0%), 1 (>0-10%), 2 (>10-50%), 3

	Experimental Group	Baseline	At start infusion	After end of infusion	+48 h post
рНа	I/R vehicle	7.36±0.01	7.11±0.03*	6.97±0.03*	7.34±0.01
	I/R allo	7.36±0.01	7.07±0.02*	6.97±0.03*	7.35±0.01
ABE (mEq.L ⁻¹)	I/R vehicle	4.8±0.5	-6.7±2.1*	-15.0±1.7*	3.0±0.8
	I/R allo	3.6±0.8	-9.0±0.98*	-15.0±1.3*	2.8±0.5
PaO2 (mmHg)	I/R vehicle	21.3±2.0	14.2±1.9*	14.0±1.4*	23.8±2.0
	I/R allo	20.8±1.9	13.6±2.4*	13.4±3.7*	21.4±1.3
PaCO2 (mmHg)	I/R vehicle	58.0±1.2	81.7±3.4*	90.8±5.5*	57.0±0.91
	I/R allo	55.2±2.0	84.4±5.4*	90.6±5.4*	54.8±1.2
SatHb (%)	I/R vehicle	55.3±5.0	24.0±3.4*	21.8±2.1*	58.8±5.7
	I/R allo	57.9±7.4	24.6±5.0*	22.8±5.9*	58.1±5.5
Lactate (mmol.L ⁻¹)	I/R vehicle	1.13±0.20	6.11±1.17*	9.79±1.24*	0.92±0.14
	I/R allo	1.05±0.20	6.09±0.94*	9.09±0.99*	1.19±0.27
Glucose (mmol.L ⁻¹)	I/R vehicle	0.97±0.09	1.66±0.29*	1.72±0.10*	1.04±0.16
	I/R allo	0.80±0.11	1.65±0.38*	1.18±0.30*	1.05±0.23

Table 6-1. Fetal arterial blood gases and metabolic status

Fetal physiologic parameters (arterial pH, blood gases, glucose and lactate values) 60 min before umbilical cord compression (baseline), immediately before and immediately after the end of infusion of allopurinol or buffered saline and at 48 hours after the I/R challenge (see Materials and Methods and Figure 6-1). Significant differences within groups are *p < 0.05 vs baseline (two-way RM ANOVA with post hoc t-Newman-Keuls test). There were no significant differences between the two treatment groups.

(>50-90%), 4 (>90-99%) and 5 (100%) ^[28-30]. Sections were randomly numbered and scored simultaneously using a dual viewing attachment by two investigators (JK and MH) who were unaware of the experimental groups.

To assess apoptosis, another set of four-micrometer coronal sections of the hippocampus, thalamus, basal nuclei, cortex and white matter was stained with a cleaved caspase-3 stain. The degree of apoptosis was determined using a 3-point scoring system as previously described ^[31]. Sections were randomly numbered and scored simultaneously using a dual viewing attachment by two investigators (HT and PN) who were unaware of the experimental groups.

Data and statistical analyses

Cardiovascular data were recorded continually at 1 s intervals using a computerized Data Acquisition System (Department of PDN, University of Cambridge, UK). Summary measures analysis was applied to the cardiovascular serial data to focus the number of comparisons and areas under the curve were calculated for statistical comparison, as previously described in detail [32]. For all variables, values are expressed as mean ± S.E.M. Comparisons between groups were assessed using one or two-way ANOVA with an appropriate post hoc test. The relationships between indices of fetal brain damage and values for fetal arterial blood gases or acid base and metabolic status were assessed using the Pearson Product Moment correlation. For all comparisons, P <0.05 was considered significant.

Results

Plasma levels of allopurinol

Administration of allopurinol to pregnant ewes led to elevations in fetal plasma levels of allopurinol between 4 and 7 mg.L⁻¹ within 90 minutes of the start of the infusion. These fetal plasma concentrations of allopurinol are within the human therapeutic range for xanthine oxidase inhibition ^[23]. Plasma levels of the active metabolite oxypurinol rose more gradually to levels between 1 and 1.5 mg.L⁻¹ at two hours after maternal administration and these elevations lasted much longer, returning towards baseline 5 hours after treatment.

Fetal arterial blood gas, acid base and metabolic status

Repeated compressions of the umbilical cord transiently decreased fetal PaO_2 , SatHb, pH, and arterial base excess (ABE) and increased $PaCO_2$, glucose and lactate concentrations (Table 6-1).

Changes in the vehicle and allopurinol treated groups, respectively, before and at the end of the I/R challenge are summarized as follows: fetal pH was reduced from 7.36 ± 0 at baseline to 6.97 ± 0.03 versus 7.36 ± 0 to 6.97 ± 0.03 by the end of the fifth compression (both groups P < 0.05), arterial base excess was reduced from 4.8 \pm 0.5 to -15 \pm 1.7 mEq.L⁻¹ versus 3.6 \pm 0.8 to -15 \pm 1.3 mEq.L⁻¹ (both groups P < 0.05), PaO, was reduced from 21 ± 2 to 14 ± 1 mmHg versus 21 ± 2 to 13 ± 2 mmHg (both groups P<0.05), SatHb was reduced from 55 \pm 5 to 22 \pm 2 % versus 58 \pm 7 to 23 ± 6 % (both groups P < 0.05), PaCO, was increased from 58 ± 1 to 91 ± 6 mmHg versus 55 ± 2 to 91 \pm 5 mmHg (both groups P < 0.05), lactate levels rose from 1.1 \pm 0.2 to 9.8 \pm 1.7 mmol.L⁻¹ versus 1.1 ± 0.2 to 9.9 ± 1.2 mmol.L⁻¹ (both groups P < 0.05) and glucose levels rose from 1.0 ± 0.1 to $1.7 \pm 0.1 \text{ mmol.L}^{-1}$ versus 0.8 ± 0.1 to 1.2 ± 0.3 mmol.L⁻¹ (both groups P < 0.05). There were no differences in the magnitude of any change in arterial blood gas, acid base excess or metabolic status between groups.

Assessment of Neuronal Necrosis

Relative to controls, the I/R challenge induced significantly greater neuronal damage in the hippocampal cornu ammonis zone 3 and 4 of fetal sheep brains (Figures 6-2 and 6-3). Maternal treatment with allopurinol during the I/R challenge restored fetal cerebral neuronal dam-



Representative images of the histopathological appearance of fetal sheep brains at 0.8 of gestation in controls (A, D, and G), the ischemia-reperfusion (I/R) vehicle group (B, E, and H), and the I/R allopurinol group (C, F, and I). Panels A to C represent zone 4 of the cornu ammonis; panels D to F are images of the parahippocampal cortex; and panels G, H, and I represent the thalamus. An acid fuchsin and thionin staining was used to detect neuronal loss (acidophilic positive neurons).

age in these areas towards control scores (Figures 2 and 3). Differences in fetal brain neuronal damage between control and I/R groups, and relative protection following I/R by maternal allopurinol treatment, was also prominent, although outside statistical significance (P=0.08), in the fetal dentate gyrus and thalamic regions (Figure 6-3).

Assessment of Neuronal Apoptosis

Relative to controls, significantly greater apoptosis in the basal nuclei and cortex was measured in fetuses subjected to the I/R challenge. However, the magnitude of apoptosis in these fetal brain regions was similar between vehicle and allopurinol treated groups (Figure 6-4). Differences in the extent of neuronal apoptosis between control and fetuses subjected to the I/R challenge, independent of allopurinol treatment, were also apparent in other fetal brain regions. However, these differences fell outside statistical significance (P=0.08, Figure 6-4).

When variables representing fetal arterial blood gases, acid base and metabolic status were correlated to indices of cerebral necrosis or apoptosis in all brain regions within individual animals for all groups, no significant relationships were found.

Fetal Cardiovascular Variables

Data concerning fetal cardiovascular variables



are described in detail in a previous article by our group^[19]. These data show that maternal treatment with allopurinol helped maintain umbilical blood flow and reduced fetal cardiac oxidative stress after I/R.

Discussion

This study was designed to investigate the potential neuroprotective effects of the FDA-approved drug allopurinol, when administered to the mother during intra-uterine asphyxia at term. To test the hypothesis, we developed an experimental model to simulate perinatal asphyxia by using 5 intermittent umbilical cord occlusions leading to clinically relevant metabolic acidosis with subsequent brain damage in late gestation fetal sheep. At the start of maternal allopurinol infusion, mean pH was 7.07±0.02 and 7.11±0.03 in the I/R allopurinol and I/R vehicle groups respec-



Values are mean+standard error of the mean (SEM). *P < .05 versus ischemia–reperfusion (I/R) v hicle and I/R allopurinol (1-way analysis of variance & Bonferroni).

tively, indicating severe fetal asphyxia. In the human clinical setting, this would be the timepoint to start preparing an emergency Cesarean section or instrumented vaginal delivery, a window of time in which maternal infusion with allopurinol could be incorporated.

The data show that in the late gestation ovine fetus repeated, intermittent, measured compressions of the umbilical cord led to neuronal cell damage and apoptosis in most brain areas examined. Ischemia-reperfusion induced significantly greater neuronal damage in the hippocampal cornu ammonis zone 3 and 4, as compared to non-ischemic controls. Differences between groups were prominent, although outside statistical significance, in the dentate gyrus and thalamus regions. Maternal administration with allopurinol during I/R restored fetal neuronal damage towards control scores, confirming a possible role for xanthine oxidase in the pathway leading to neuronal damage and supporting a potential neuroprotective effect of antenatal antioxidant treatment.

Several experimental models to induce fetal asphyxia have previously been performed; either partial occlusion of the umbilical cord for 30 minutes to several hours, repetitive infrequent occlusions lasting 5 minutes, occlusions of increasing length and/or frequency and (nearly) total occlusion for 10 minutes, as in our model ^[28,29,33-40]. Although a broad variation of distribution patterns of neuronal damage and apoptosis have been identified after using these different techniques, the observed (para)hippocampal vulnerability has frequently been described ^[28,34,35,41]. In our study, the hippocampus and the parahippocampal cortex, areas involved in memory and spatial orientation and navigation, were also the most severely damaged brain areas, results in keeping with the characteristics of such studies.

Apparent differences in histological brain damage between the I/R allo and I/R vehicle groups that did not reach statistical significance in the

preset study are most likely due to the small number of subjects used. Alternatively, a lack of an effect of allopurinol treatment could reflect brain damage triggered via pro-oxidant pathways other than via the activation of xanthine oxidase. Several other pathways may promote oxidative stress, such as the glutamate-induced excitotoxicity which proceeds via NMDA receptor activation, producing Ca2+ influx and thereby activation of Ca2+-dependent NOS, particularly nNOS. At high concentrations, NOreacts with superoxide (O,) to produce peroxynitrite (ONOO-), which in turn induces lipid peroxidation and mitochondrial nitrosylation. Consequently, mitochondrial dysfunction and membrane depolarization develop with further release of O₂.

An increased influx of calcium also leads to the activation of cytosolic phospholipases, increasing eicosanoid release and inflammation with concomitant accumulation of neutrophils. A combined therapeutic approach using drugs, which influence different parts of the neurotoxic cascade, like melatonin ^[42] or tetrahydrobiopterin, might therefore further improve the outcome of antioxidant therapy.

In the present study, analysis of indices of apoptosis also revealed significant programmed cell death in most brain areas of fetal sheep that underwent ischemia-reperfusion. However, maternal treatment with allopurinol during the I/R challenge did not have a significant effect on reducing the amount of apoptotic brain cells. This may be because histological assessment was performed at 48 hours after ischemia reperfusion, while the process of apoptosis normally reaches its maximum 72 hours after a hypoxic ischemic event ^[30].

Previous papers have reported on effects of allopurinol in post-asphyxial reperfusion brain damage in both animals and humans. Although allopurinol has been administered both before and after hypoxic-ischemic insults in these studies, most of the experiments have been carried out postnatally. Only five other studies in the literature have reported on antenatally administered allopurinol, but none of them have described any effects on the histology of the brain ^[18,19,22,23,43].

In the present study, therapeutic ranges of allopurinol and its metabolite oxypurinol have shown to be attainable after antenatal maternal treatment with allopurinol, while no adverse side effects were observed. These findings are in line with previous studies in both animals and humans ^[18,22,23,43]. Masaoka *et al.* were the first to investigate antenatally administered allopurinol during intermittent umbilical cord occlusion in fetal sheep, and described a significantly reduced production of superoxide in the allopurinol treated animals ^[22]. This indicates either a direct effect of allopurinol in preventing the production of free radicals, in scavenging free radicals once formed, or both. These results were confirmed in the ovine fetuses used in the present study, in which we found reduced indices of oxidative stress in the cardiovascular system after maternal treatment with allopurinol ^[19]. A randomized controlled human clinical pilot study performed by our group has also

shown a significant reduction of Non Protein Bound Iron (NPBI) in the cord blood of allopurinol treated infants, confirming the oxidative stress-reducing effect of allopurinol in humans. Furthermore, an inverse correlation between levels of allopurinol and levels of S-100ß, a marker of brain damage, in umbilical cord blood was seen, further indicating a potential neuroprotective effect ^[18].

In summary, this is the first report investigating the protective value of antenatal administration of allopurinol during birth asphyxia on brain damage in the late gestation sheep fetus. The data support the hypothesis tested that maternal treatment with allopurinol during repeated episodes of fetal asphyxia would limit I/R damage to the fetal brain. Maternal treatment with allopurinol therefore remains a promising therapeutic option to complement neonatal head cooling. A large prospective multicenter placebo controlled trial in the Netherlands investigating the effect of maternal administration of allopurinol on markers of brain damage and neonatal outcome in humans has recently completed recruitment [44].

ACKNOWLEDGMENTS

We thank Mr Scott Gentle and Mrs Sue Nicholls for their help with the animal maintenance

REFERENCES OF CHAPTER 6

1. Cowan F, Rutherford M, Groenendaal F, *et al.* Origin and timing of brain lesions in term infants with neonatal encephalopathy. Lancet. 2003;361(9359):736-742.

2. Vannucci RC, Perlman JM. Interventions for perinatal hypoxic-ischemic encephalopathy. Pediatrics. 1997;100(6):1004-1014.

3. Cluckman PD, Wyatt JS, Azzopardi D, et al. Selective head cooling with mild systemic hypothermia after neonatal encephalopathy: Multicentre randomized trial. Lancet. 2005;365(9460):663-670.

4. Gunn AJ, Thoresen M. Hypothermic neuroprotection. NeuroRx. 2006;3(2):154-169.

5. Edwards AD, Brocklehurst P, Gunn AJ, *et al.* Neurological outcomes at 18 months of age after moderate hypothermia for perinatal hypoxic ischaemic encephalopathy: Synthesis and meta-analysis of trial data. BMJ. 2010;340:c363.

6. Robertson NJ, Tan S, Groenendaal F, *et al.* Which neuroprotective agents are ready for bench to bedside translation in the newborn infant? J Pediatr. 2012;160(4):544-552.e4.

7. Clapp JF, Peress NS, Wesley M, Mann Ll. Brain damage after intermittent partial cord occlusion in the chronically instrumented fetal lamb. Am J Obstet Gynecol. 1988;159(2):504-509.

8. Masaoka N, Hayakawa Y, Ohgame S, Sakata H, Satoh K, Takahashi H. Changes in purine metabolism and production of oxygen free radicals by intermittent partial umbilical cord occlusion in chronically instrumented fetal lambs. J Obstet Gynaecol Res. 1998;24(1):63-71.

9. Rogers MS, Wang CC, Lau TK, et al. Relationship between isoprostane concentrations, metabolic acidosis, and morbid neonatal outcome. Clin Chem. 2005;51(7):1271-1274.

10. Fan X, van Bel F. Pharmacological neuroprotection after perinatal asphyxia. J Matern Fetal Neonatal Med. 2010.

 Miller SL, Wallace EM, Walker DW. Antioxidant therapies: A potential role in perinatal medicine. Neuroendocrinology. 2012;96(1):13-23.

12. Saugstad OD. Role of xanthine oxidase and its inhibitor in hypoxia: Reoxygenation injury. Pediatrics. 1996;98(1):103-107.

13. Ono T, Tsuruta R, Fujita M, *et al.* Xanthine oxidase is one of the major sources of superoxide anion radicals in blood after reperfusion in rats with forebrain ischemia/reperfusion. Brain Res. 2009;1305:158-167.

14. Pacher P, Nivorozhkin A, Szabo C. Therapeutic effects of xanthine oxidase inhibitors: Renaissance half a century after the discovery of allopurinol. Pharmacol Rev. 2006;58(1):87-114.

15. Van Bel F, Shadid M, Moison RM, *et al.* Effect of allopurinol on postasphyxial free radical formation, cerebral hemodynamics, and electrical brain activity. Pediatrics. 1998;101(2):185-193.

16. Benders MJ, Bos AF, Rademaker CM, *et al*. Early postnatal allopurinol does not improve short term outcome after severe birth asphyxia. Arch Dis Child Fetal Neonatal Ed. 2006;91(3):F163-5.

17. Gunes T, Ozturk MA, Koklu E, Kose K, Gunes I. Effect of allopurinol supplementation on nitric oxide levels in asphyxiated newborns. Pediatr Neurol. 2007;36(1):17-24.

18. Torrance HL, Benders MJ, Derks JB, *et al*. Maternal allopurinol during fetal hypoxia lowers cord blood levels of the brain injury marker S-100B. Pediatrics. 2009;124(1):350-357.

19. Derks JB, Oudijk MA, Torrance HL, *et al.* Allopurinol reduces oxidative stress in the ovine fetal cardiovascular system after repeated episodes of ischemia-reperfusion. Pediatr Res. 2010;68(5):374-380.

20. Kaandorp JJ, van Bel F, Veen S, *et al.* Long-term neuroprotective effects of allopurinol after moderate perinatal asphyxia: Follow-up of two randomized controlled trials. Arch Dis Child Fetal Neonatal Ed. 2012;97(3):F162-6. doi: 10.1136/archdischild-2011-300356. 10.1136/archdischild-2011-300356.

21. Herrera EA, Kane AD, Hansell JA, *et al.* A role for xanthine oxidase in the control of fetal cardiovascular function in late gestation sheep. J Physiol. 2012;590(Pt 8):1825-1837.

22. Masaoka N, Nakajima Y, Hayakawa Y, *et al*. Transplacental effects of allopurinol on suppression of oxygen free radical production in chronically instrumented fetal lamb brains during intermittent umbilical cord occlusion. J Matern Fetal Neonatal Med. 2005;18(1):1-7.

23. Boda D, Németh P, Kiss P, Orvos H. Treatment of mothers with allopurinol to produce therapeutic blood levels in newborns. Prenat Neonat Med. 1999;4:130-132, 133, 134.

24. Thakor AS, Herrera EA, Seron-Ferre M, Giussani DA. Melatonin and vitamin C increase umbilical blood flow via nitric oxide-dependent mechanisms. J Pineal Res. 2010;49(4):399-406.

25. Kane AD, Herrera EA, Hansell JA, Giussani DA. Statin treatment depresses the fetal defence to acute hypoxia via increasing nitric oxide bioavailability. J Physiol. 2012;590(Pt 2):323-334.

26. Gardner DS, Fletcher AJ, Fowden AL, Giussani DA. A novel method for controlled and reversible long term compression of the umbilical cord in fetal sheep. J Physiol. 2001;535(Pt 1):217-229.

27. van Kesteren C, Benders MJ, Groenendaal F, van Bel F, Ververs FF, Rademaker CM. Population pharmacokinetics of allopurinol in full-term neonates with perinatal asphyxia. Ther Drug Monit. 2006;28(3):339-344.

28. Gunn AJ, Parer JT, Mallard EC, Williams CE, Gluckman PD. Cerebral histologic and electrocorticographic changes after asphyxia in fetal sheep. Pediatr Res. 1992;31(5):486-491.

29. De Haan HH, Gunn AJ, Williams CE, Gluckman PD. Brief repeated umbilical cord occlusions cause sustained cytotoxic cerebral edema and focal infarcts in near-term fetal lambs. Pediatr Res. 1997;41(1):96-104.

30. Fraser M, Bennet L, Gunning M, *et al.* Cortical electroencephalogram suppression is associated with post-ischemic cortical injury in 0.65 gestation fetal sheep. Brain Res Dev Brain Res. 2005;154(1):45-55.

31. Groenendaal F, Lammers H, Smit D, Nikkels PG. Nitrotyrosine in brain tissue of neonates after perinatal asphyxia. Arch Dis Child Fetal Neonatal Ed. 2006;91(6):F429-33.

32. Matthews JN, Altman DG, Campbell MJ, Royston P. Analysis of serial measurements in medical research. BMJ. 1990;300(6719):230-235.

33. Goni-de-Cerio F, Alvarez A, Caballero A, *et al.* Early cell death in the brain of fetal preterm lambs after hypoxic-ischemic injury. Brain Res. 2007;1151:161-171.

34. Mallard EC, Gunn AJ, Williams CE, Johnston BM, Gluckman PD. Transient umbilical cord occlusion causes hippocampal dam-

age in the fetal sheep. Am J Obstet Gynecol. 1992;167(5):1423-1430.

35. Mallard EC, Williams CE, Johnston BM, Gluckman PD. Increased vulnerability to neuronal damage after umbilical cord occlusion in fetal sheep with advancing gestation. Am J Obstet Gynecol. 1994;170(1 Pt 1):206-214.

36. de Haan HH, Van Reempts JL, Vles JS, de Haan J, Hasaart TH. Effects of asphyxia on the fetal lamb brain. Am J Obstet Gynecol. 1993;169(6):1493-1501.

37. Ikeda T, Choi BH, Yee S, Murata Y, Quilligan EJ. Oxidative stress, brain white matter damage and intrauterine asphyxia in fetal lambs. Int J Dev Neurosci. 1999;17(1):1-14.

38. Falkowski A, Hammond R, Han V, Richardson B. Apoptosis in the preterm and near term ovine fetal brain and the effect of intermittent umbilical cord occlusion. Brain Res Dev Brain Res. 2002;136(2):165-173.

39. Rocha E, Hammond R, Richardson B. Necrotic cell injury in the preterm and near-term ovine fetal brain after intermittent umbilical cord occlusion. Am J Obstet Gynecol. 2004;191(2):488-496.

40. Berger R, Lehmann T, Karcher J, Schachenmayr W, Jensen A. Relation between cerebral oxygen delivery and neuronal cell damage in fetal sheep near term. Reprod Fertil Dev. 1996;8(3):317-321.

41. Castillo-Melendez M, Chow JA, Walker DW. Lipid peroxidation, caspase-3 immunoreactivity, and pyknosis in late-gestation fetal sheep brain after umbilical cord occlusion. Pediatr Res. 2004;55(5):864-871.

42. Miller SL, Yan EB, Castillo-Melendez M, Jenkin G, Walker DW. Melatonin provides neuroprotection in the late-gestation fetal sheep brain in response to umbilical cord occlusion. Dev Neurosci. 2005;27(2-4):200-210.

43. van Dijk AJ, Parvizi N, Taverne MA, Fink-Gremmels J. Placental transfer and pharmacokinetics of allopurinol in late pregnant sows and their fetuses. J Vet Pharmacol Ther. 2008;31(6):489-495.

44. Kaandorp JJ, Benders MJ, Rademaker CM, *et al*. Antenatal allopurinol for reduction of birth asphyxia induced brain damage (ALLO-trial); a randomized double blind placebo controlled multicenter study. BMC Pregnancy Childbirth. 2010;10:8.

Chapter 7 Antenatal allopurinol for reduction of birth asphyxia induced brain damage (ALLO-Trial); a randomized double blind placebo controlled multicenter study (study protocol)

JOEPE J. KAANDORP, MANON J.N.L. BENDERS, CARIN M.A. RADEMAKER, HELEN L. TORRANCE, MARTIJN A. OUDIJK, TIMO R. DE HAAN, KITTY W.M. BLOEMENKAMP, MONIQUE RIJKEN, MARIA G. VAN PAMPUS, ARIE F. BOS, MARTINA M. PORATH, SIDARTO BAMBANG OETOMO, CHRISTINE WILLEKES, A.W. DANILO GAVILANES, MAURICE G.A.J. WOUTERS, RUURD M. VAN ELBURG, ANJOKE J.M. HUISJES, SASKIA C.M.J.E.R. BAKKER, CLAUDIA A. VAN MEIR, JEANNETTE VON LINDERN, JANINE BOON, INGE P. DE BOER, ROBBERT J.P. RIJNDERS, CORRIE J.W.F.M. JACOBS, CUNO S.P.M. UITERWAAL, BEN WILLEM J. MOL, GERARD H.A. VISSER, FRANK VAN BEL, JAN B. DERKS

BMC Pregnancy Childbirth. 2010 Feb 18;10:8



ABSTRACT

Background

Hypoxic-ischemic encephalopathy is associated with the development of cerebral palsy and cognitive disability later in life and is therefore one of the fundamental problems in perinatal medicine. The xanthine-oxidase inhibitor allopurinol reduces the formation of free radicals, thereby limiting the amount of hypoxia-reperfusion damage. In case of suspected intra-uterine hypoxia, both animal and human studies suggest that maternal administration of allopurinol immediately prior to delivery reduces hypoxic-ischemic encephalopathy.

Methods/Design

The proposed trial is a randomized double blind placebo controlled multicenter study in pregnant women at term in whom the fetus is suspected of intra-uterine hypoxia.

Allopurinol 500 mg IV or placebo will be administered antenatally to the pregnant woman when fetal hypoxia is suspected. Fetal distress is being diagnosed by the clinician as an abnormal or non-reassuring fetal heart rate trace, preferably accompanied by either significant ST-wave abnormalities (as detected by the STAN-monitor) or an abnormal fetal blood scalp sampling (pH < 7.20). Primary outcome measures are the amount of S100ß (a marker for brain tissue damage) and the severity of oxidative stress (measured by isoprostane, neuroprostane, non protein bound iron and hypoxanthine), both measured in umbilical cord blood. Secondary outcome measures are neonatal mortality, serious composite neonatal morbidity and long-term neurological outcome. Furthermore pharmacokinetics and pharmacodynamics will be investigated. We expect an inclusion of 220 patients (110 per group) to be feasible in an inclusion period of two years. Given a suspected mean value of S100ß of 1.05 ug/L (SD 0.37 ug/L) in the placebo group this trial has a power of 90% (alpha 0.05) to detect a mean value of S100ß of 0.89 ug/L (SD 0.37 ug/L) in the 'allopurinol-treated' group (z-test2-sided). Analysis will be by intention to treat and it allows for one interim analysis.

Discussion

In this trial we aim to answer the question whether antenatal allopurinol administration reduces hypoxic-ischemic encephalopathy in neonates exposed to fetal hypoxia. ypoxic-ischemic encephalopathy is associated with development of cerebral palsy and cognitive disability later in life ^[1]. The recognition, prevention and treatment of intra-uterine hypoxia is therefore one of the main priorities in perinatal medicine. Animal and human studies have shown that brain damage not only occurs during the hypoxic-ischemic event, but continues for hours up to days upon and after reoxygenation and reperfusion and is caused by production of free radicals ^[2].

Free radical formation due to conversion of hypoxanthine into xanthine by xanthine-oxidase is very important during this process ^[3]. Administration of the xanthine-oxidase inhibitor allopurinol (ALLO) reduces the production of free radicals, thereby limiting the amount of hypoxia-reperfusion damage ^[4,5]. Furthermore, ALLO also has a non-protein bound iron (pro-radical) chelating and direct free radical (hydroxyl) scavenging effect. Animal research in asphyxiated pigs demonstrated beneficial effects of postnatally administrated ALLO on cerebral energy status and cytotoxic edema ^[6].

A prospective randomized study in human neonates, examining the effects of ALLO in term asphyxiated neonates, showed an improvement of electrocortical brain activity and a reduction in free radical formation after neonatal ALLO administration ^[7]. A more recent paper by Gunes *et al*.^[8] reports an improved neurological outcome after postnatal ALLO administration compared to a placebo in term asphyxiated neonates. Benders *et al* however demonstrated that ALLO was not effective if administrated 3 to 4 hours after the hypoxic incident to severely asphyxiated neonates ^[9]. However, when the most severely asphyxiated children were excluded from the study, a beneficial effect of ALLO was seen on neurological development. This is in line with previous studies by Gluckman et al. on neonatal head cooling ^[10]. They also demonstrated a beneficial effect of their treatment after exclusion of the most severely asphyxiated neonates. Apparently, no advantage of neonatal treatment is seen anymore, when the interval to the initiation of treatment is too long or when the brain damage is too severe. This has probably been the major disadvantage of late post neonatal treatment with ALLO on the Neonatal Intensive Care Unit (NICU). ALLO administrated at the NICU is likely to be given too late to provide adequate neuroprotection during the early period of reoxygenation in which the vast amount of free radicals is being produced.

Apparently, when the asphyxia has been too severe, the inflicted brain damage can no longer be reversed. It is conceivable that earlier ALLO treatment, i.e. the use of ALLO during labor in case of suspected fetal hypoxia, provides the opportunity to start earlier with the treatment, thereby limiting the amount of hypoxia-reperfusion injury and improving neurological outcome.

Pharmacological data have been published by Boda *et al*, who showed that pharmacological plasma levels were reached in the human fetus after ALLO administration to the mother ^[11]. A systematic review in Pubmed, searching for [allopurinol, fetus, neonate, asphyxia, hypoxia] did not provide us with any additional studies performed in humans. Some animal studies have been performed using piglets (neonatal administration), sheep and sows (maternal administration) ^[6,12-14]. These studies showed improved outcome after allopurinol treatment on cerebral energy status and recovery of umbilical blood flow respectively.

We recently performed a prospective randomized placebo controlled pilot study, in which we administered ALLO to the pregnant woman when fetal asphyxia was imminent. Data from this pilot study show an inverse correlation between levels of ALLO and the amount of S100ß, a biomarker for brain tissue damage, in cord blood ^[15]. In addition, we performed a study in the chronically instrumented fetal sheep, in which we showed evidence of cardio- and neuroprotection after antenatal ALLO administration to the pregnant ewe during repeated periods of ischemia ^[13,16].

Incidence and financial impact of birth asphyxia

Birth asphyxia carries a high incidence, 4-9 per 1000 live born neonates. On estimate 1-4 of these neonates will die or develop a severe handicap ^[17,18]. With almost 200.000 deliveries in the Netherlands annually, this implicates that each year approximately 800 neonates will die or suffer from a handicap due to birth asphyxia. Moreover, there is evidence that individuals suffering from moderate birth asphyxia, develop behavioral problems later in life ^[19].

The health care costs of asphyxia are enormous. Many of these children are admitted to a neonatal intensive care unit at a cost of \notin 1.500, per day, with an average duration of admission of 10 days. Estimated costs of disabled children are \notin 80.000, - and \notin 20.000, - yearly for severely and moderately handicapped children respectively. The costs of neonatal care are annually 20 million Euros, whereas the costs of care for disabled children born in one year are more than one hundred million Euros.

As the costs of ALLO and its administration are relatively low (100 Euros per pregnant woman), a small treatment effect will already make the intervention cost-effective.

Safety of allopurinol

Allopurinol has been used in internal medicine for many years for the treatment of gout. Side effects of ALLO are rare and most commonly include symptoms of hypersensitivity like skin rashes ^[20]. Stevens-Johnson syndrome has been reported, but only after prolonged treatment (> 10 days) in case of gout ^[21]. During previously performed studies in pregnant women and neonates, no adverse reactions were seen ^[11,15,22,23].

Methods / Design

Aims

The objective of the study is to test the hypothesis that intra-uterine treatment with ALLO in case of suspected fetal hypoxia will reduce brain damage and with that may improve neonatal outcome.

Participants/eligibility criteria

Pregnant women with a gestational age of at least 36 weeks and suspected intra-uterine hypoxia (fetal distress) during labor can be included in the trial. Fetal distress is being diagnosed by the clinician as an abnormal or non-reassuring fetal heart rate trace, preferably accompanied by either significant ST-wave abnormalities (as detected by the STAN-monitor) or an abnormal fetal blood scalp sampling (pH < 7.20). Neonates suspected of chromosomal or congenital anomalies will be excluded.

Procedures, recruitment, randomization and collection of baseline data

The study will be a randomized double blind placebo controlled multicenter study. Hospitals which participate in the Dutch Consortium for Studies in Women's health and reproductivity will participate in the trial. The study will be staffed by research midwives and -nurses, who will counsel patients at the outpatient clinic or at the obstetric ward.

Before entry into the study, subjects will be informed about the aims, methods, reasonably anticipated benefits and potential hazards of the study. Subjects will be informed that their participation is voluntary and that they may withdraw consent to participate at any time during the study. They will be informed that choosing not to participate will not affect their care. In every center an independent gynecologist will be available for more detailed information both for patients and colleagues if required. After giving sufficient information, written informed consent will be obtained. The consent form must be signed before performance of any study-related activity.

In case of suspected fetal distress and written informed consent, the patient will be randomized immediately prior to delivery (Figure 7-1).

All details of delivery are recorded in the case record form that is accessible through a website (www.studies-obsgyn.nl/allo). In case of admittance of one or more children to the neonatal intensive care unit, details of this admittance are also recorded.

Data will be collected using Oracle Clinical Remote Data Capture (RDC), which is a new generation of application systems that enable collection and cleanup of clinical trial data using the Internet. For detailed information on Oracle RDC see http://www.ctc-g.co.jp/~CTCLS/ opa/en/. The expertise for this technology is already used in the study group and applied in the ZonMw funded consortium studies that are currently running.

Interventions

All women who are identified as having suspected fetal distress and have given informed consent prior to delivery will be randomized in two groups: One group of pregnant women will receive one IV dosage of 500 mg ALLO. The other group will receive one IV dosage of placebo antenatally.

Outcome measures

Primary outcome measures are related to biochemical brain damage markers (S100ß, enolase) and the severity of oxidative stress as measured by isoprostane, neuroprostane, non protein bound iron and hypoxanthine in umbilical cord blood.

The main primary outcome measure will be the protein S100ß as this is, at present, the most validated biochemical marker for brain tissue damage. Secondary outcome parameters are neonatal mortality and short term neonatal morbidity, such as hypoglycemia, convulsions and post-hypoxic neonatal encephalopathy, length of admission at the neonatal intensive care unit and placental transfer, pharmacodynamics and -kinetics of ALLO. Neonatal encephalopathy will be assessed by Sarnat- and Thompson scores ^[24,25].

Follow up of women and infants

Blood sampling will be performed to assess the amount of tissue damage (i.e. S100ß, enolase, troponin), oxidative stress (i.e hypoxanthine, isoprostane, neuroprostane and non protein bound iron) and ALLO plasma levels. Immediately after delivery, samples of cord blood and maternal blood are obtained. One hour and 18-24 hours after birth neonatal blood samples are obtained in all children with clinically indicated blood sampling.

Long-term follow-up using neuropsychological tests and validated neurodevelopmental questionnaires at the age of 4-5 and 8 years and additional MRI at the age of 8 years is desirable, but is depending on future funding.

Statistical methods

Sample size

The sample size is calculated based on the primary outcome measure cord-S100ß as a marker of brain damage. Our own pilot study showed a mean value of S100ß of 1.05 ug/L (SD 0.37 ug/L) in the 'non-treated' group with suspected intra-uterine hypoxia. Based on this same pilot study we expect the mean value of S100ß to be lower in the 'allopurinol-treated' group compared to the 'non-treated group'.

Based on the accrual of patients in our pilot study we consider an inclusion of 220 patients (110 ALLO, 110 placebo) to be feasible in an inclusion period of 2 years. Given a suspected mean value of S100ß of 1.05 ug/L (SD 0.37 ug/L) in the placebo group this trial has a power of 90% with an alpha of 0.05 to detect a mean value of S100ß of 0.89 ug/L (SD 0.37 ug/L) in the 'allopurinol-treated' group (z-test2-sided)^[15].



Data analysis

Data will be analyzed according to the intention to treat principle. Outcome measures will be analyzed with (non)parametric tests where continuous variables are concerned. Proportional data will be reported as relative risks with 95% confidence intervals. Multivariable regression techniques will be applied to correct for any important differences in prognostic baseline characteristics, despite of randomization, by adding all prognostic variables as independent variables. Both corrected and uncorrected group differences will be reported. Numbers needed to treat (NNT) will be reported for both continuous and discrete data.

Interim analysis

An interim analysis will be performed at t=0.5 using O'Brien –Fleming alpha spending function. An interim analysis will be performed after the inclusion of 110 women. This analysis will be done by an independent person that will be unaware of the allocation of treatment or placebo when they judge data on effectiveness.

Data safety monitoring committee

Serious Adverse Events (SAEs) and Suspected Unexpected Serious Adverse Reactions (SU-SARs) will be reported to a Data Safety Monitoring Committee (DSMC). The DSMC can order to perform an extra interim analysis and, if indicated, terminate the trial prematurely.

Ethical considerations

This study is approved by the National Central Committee on Research involving Human Subjects (CCMO – NL26516.000.09).

Discussion

In conclusion, hypoxic-ischemic encephalopathy is associated with development of cerebral palsy and cognitive disability later in life and is therefore one of the fundamental problems in perinatal medicine. Prevention of brain damage and developing adequate therapy is therefore of big importance. The xanthine-oxidase inhibitor allopurinol (ALLO) reduces free radical formation, thereby limiting the amount of hypoxia-reperfusion damage. Animal and human studies suggest that administration of ALLO immediately prior to delivery in case of suspected fetal asphyxia might reduce hypoxic-ischemic encephalopathy. We designed a randomized placebo controlled multicenter trial to determine whether intra-uterine treatment with allopurinol reduces hypoxic-ischemic encephalopathy in neonates exposed to fetal hypoxia.

To our knowledge no similar studies are or will shortly be performed in the Netherlands or abroad.

ACKNOWLEDGEMENTS

This study is funded by ZonMw, a Dutch organization for Health Research and Development. Project number 170991001.

Trial registration number

Clinical Trials, protocol registration system: NCT00189007

REFERENCES OF CHAPTER 7

1. Fellman V, Raivio KO. Reperfusion injury as the mechanism of brain damage after perinatal asphyxia. Pediatr Res 1997; 41:599-606.

2. Rosenberg AA, Murdaugh E, White CW. The role of oxygen free radicals in postasphyxia cerebral hypoperfusion in newborn lambs. Pediatr Res 1989; Sep;26:215-219.

3. Warner DS, Sheng H, Batinic-Haberle I. Oxidants, antioxidants and the ischemic brain. J Exp Biol 2004; 207:3221-3231.

4. Palmer C, Vannucci RC, Towfighi J. Reduction of perinatal hypoxic-ischemic brain damage with allopurinol. Pediatr Res 1990; 27:332-336.

 Whitelaw A. Systematic review of therapy after hypoxic-ischaemic brain injury in the perinatal period. Semin Neonatol 2000; 5:33-40.

6. Peeters-Scholte C, Braun K, Koster J, Kops N, Blomgren K, Buonocore C, van Buul-Offers S, Hagberg H, Nicolay K, van Bel F, Groenendaal F. Effects of allopurinol and deferoxamine on reperfusion injury of the brain in newborn piglets after neonatal hypoxia-ischemia. Pediatr Res 2003; 54:516-522.

7. Van Bel F, Shadid M, Moison RM, Dorrepaal CA, Fontijn J, Monteiro L, van de Bor, Berger HM. Effect of allopurinol on postasphyxial free radical formation, cerebral hemodynamics, and electrical brain activity. Pediatrics 1998; 101:185-193.

8. Gunes T, Ozturk MA, Koklu E, Kose K, Gunes I. Effect of allopurinol supplementation on nitric oxide levels in asphyxiated newborns. Pediatr Neurol 2007; 36:17-24.

9. Benders MJ, Bos AF, Rademaker CM, Rijken M, Torrance HL, Groenendaal F, van Bel F. Early postnatal allopurinol does not improve short term outcome after severe birth asphyxia. Arch Dis Child Fetal Neonatal Ed 2006; 91:163-165.

10. Gluckman PD, Wyatt JS, Azzopardi D, Ballard R, Edwards AD, Ferriero DM, Polin RA, Robertson CM, Thoresen M, Whitelaw A, Gunn AJ. Selective head cooling with mild systemic hypothermia after neonatal encephalopathy: multicentre randomized trial. Lancet 2005; 365:663-670.

11. Boda D, Németh P, Kiss P, Orvos H. Treatment of mothers with allopurinol to produce therapeutic blood levels in newborns. Prenat Neonat Med 1999;4: 130-134.

12. van Dijk AJ, Parvizi N, Taverne MA, Fink-Gremmels J. Placental transfer and pharmacokinetics of allopurinol in late pregnant sows and their fetuses. J Vet Pharmacol Ther 2008; 31:489-495.

13. Derks JB, Oudijk MA, Rosén KG, Thakor AS, Visser GHA, van Bel F, Giussani DA. Allopurinol maintains umbilical blood flow following repeated episodes of ischaemia-reperfusion in fetal sheep during late gestation [abstract]. J Soc Gynecol Invest 2006;13:s35.

14. Masaoka N, Nakajima Y, Hayakawa Y, Ohgame S, Hamano S, Nagaishi M, Yamamoto T. Transplacental effects of allopurinol on suppression of oxygen free radical production in chronically instrumented fetal lamb brains during intermittent umbilical cord occlusion. J Matern Fetal Neonatal Med 2005; 18:1-7.

15. Torrance HL, Benders MJ, Derks JB, Rademaker CM, Bos AF, Van Den Berg P, Longini M, Buonocore G, Venegas M, Baquero H, Visser GH, Van Bel F. Maternal allopurinol during fetal hypoxia lowers cord blood levels of the brain injury marker S-100B. Pediatrics 2009; 124:350-357.

16. Torrance H, Derks JB, Oudijk MA, Thakor AS, Cindrova-Davies T, van Bel F, Visser GHA, Burton GJ, Giussani DA.: Maternal treatment with allopurinol diminishes fetal cardiac oxidative stress following repeated episodes of ischaemia-reperfusion in sheep. In Proceedings of the 34th annual meeting of the Fetal and Neonatal Physiological Society; 26-29 august 2007; Sendai, Japan. 2007;90.

17. Ferriero DM. Neonatal brain injury. N Engl J Med 2004; 351:1985-1995.

18. Hagberg B, Hagberg G, Olow I. The changing panorama of cerebral palsy in Sweden. VI. Prevalence and origin during the birth year period 1983-1986. Acta Paediatr 1993; 82:387-393.

19. van Handel M, Swaab H, de Vries LS, Jongmans MJ. Long-term cognitive and behavioral consequences of neonatal encephalopathy following perinatal asphyxia: a review. Eur J Pediatr 2007; 166:645-654.

20. Summary of Product Characteristics Acepurin®. [http://db.cbg-meb.nl/Bijsluiters/h09974.pdf]

21. Halevy S, Ghislain PD, Mockenhaupt M, Fagot JP, Bouwes Bavinck JN, Sidoroff A, Naldi L, Dunant A, Viboud C, Roujeau JC; EuroSCAR Study Group. Allopurinol is the most common cause of Stevens-Johnson syndrome and toxic epidermal necrolysis in Europe and Israel. J Am Acad Dermatol 2008; 58:25-32.

22. Chaudhari T, McGuire W. Allopurinol for preventing mortality and morbidity in newborn infants with suspected hypoxic-ischaemic encephalopathy. Cochrane Database Syst Rev 2008; Apr 16;(2):CD006817.

23. van Kesteren C, Benders MJ, Groenendaal F, van Bel F, Ververs FF, Rademaker CM. Population pharmacokinetics of allopurinol in full-term neonates with perinatal asphyxia. Ther Drug Monit 2006; 28:339-344.

24. Sarnat HB, Sarnat MS. Neonatal encephalopathy following fetal distress. A clinical and electroencephalographic study. Arch Neurol 1976; 33:696-705.

25. Thompson CM, Puterman AS, Linley LL, Hann FM, van der Elst CW, Molteno CD, Malan AF. The value of a scoring system for hypoxic ischaemic encephalopathy in predicting neurodevelopmental outcome. Acta Paediatr 1997; 86:757-761.

Chapter 8 Rapid target allopurinol concentrations in the hypoxic fetus after maternal administration during labor

JOEPE J. KAANDORP^{*}, MARCEL P.H. VAN DEN BROEK^{*}, MANON J.N.L. BENDERS, MARTIJN A. OUDIJK, MARTINA M. PORATH, SIDARTO BAMBANG OETOMO, MAURICE G.A.J. WOUTERS, RUURD M. VAN ELBURG, MAUREEN F.M. FRANSSEN, ARIE F. BOS, BEN-WILLEM J. MOL, GERARD H.A. VISSER, FRANK VAN BEL, CARIN M.A. RADEMAKER, JAN B. DERKS FOR THE ALLO-TRIAL STUDY GROUP *both authors contributed equally to this manuscript

Accepted by the Archives of Diseases in Childhood, Fetal and Neonatal Edition



ABSTRACT

Design

We used data from a randomized double blind multicenter trial comparing maternal allopurinol versus placebo in case of imminent fetal hypoxia (NCT00189007).

Patients

We studied 58 women in labor at term, with suspected fetal hypoxia prompting immediate delivery, in the intervention arm of the study.

Setting

Delivery rooms of 11 Dutch hospitals.

Intervention

500 mg allopurinol, intravenously to the mother, immediately prior to delivery.

Main outcome measures

Drug disposition (maternal plasma concentrations, cord blood concentrations) and drug safety (maternal and fetal adverse events).

Results

Within 5 minutes after the end of maternal allopurinol infusion target plasma concentrations of allopurinol of ≥ 2 mg/L were present in cord blood. Of all analyzed cord blood samples, 95% (52/55) had a target allopurinol plasma concentration at the moment of delivery. No adverse events were observed in the neonates. Two mothers had a red and/or painful arm during infusion.

Conclusions

A dose of 500 mg intravenous allopurinol rapidly crosses the placenta, and provides target concentrations in 95% of the fetuses at the moment of delivery, which makes it potentially useful as a neuroprotective agent in perinatology with very little side-effects.

espite advances in maternal and neonatal care, limited progress has been achieved in reducing the incidence of fetal asphyxia and subsequent neurodevelopmental disabilities ^[1]. Neuronal damage following birth asphyxia is most likely caused by the formation of free radicals in the reperfusion phase starting directly after birth ^[2]. Currently, the only established treatment for free radical induced hypoxic-ischemic encephalopathy is postnatal mild hypothermia, often started several hours after the ischemic and reperfusion phases ^[3,4]. Therefore, there is need to optimize therapeutic options to further prevent or reduce hypoxia-induced brain damage during labor, preferably in an earlier stage in the process of (or even before) free radical production.

Several potential antioxidant pharmacologic options to intervene in the putative neurotoxic cascade have already been studied ^[5,6]. Drugs like tetrahydrobiopterin, melatonin, nNOS inhibitors, xenon and vitamin C showed promising results in experimental studies, but have not yet been translated to clinical use ^[7,8].

Several pathways may promote oxidative stress during labor, such as the glutamate-induced excitotoxicity which proceeds via N-methyl-D-aspartate (NMDA) receptor activation, producing Ca2+ influx and thereby activation of Ca2+-dependent nitric oxide synthase (NOS), particularly neuronal NOS [9-12]. At high concentrations, nitric oxide (NO-) reacts with superoxide (O_{1}^{-}) to produce peroxynitrite (ONOO-), which in turn induces lipid peroxidation and mitochondrial nitrosylation [13]. Consequently, mitochondrial dysfunction and membrane depolarization develop with further release of (O_{2}) . An increased influx of calcium also leads to the activation of cytosolic phospholipases, increasing eicosanoid release and inflammation with concomitant accumulation of neutrophils^[13,14].

One of the main pro-oxidant pathways stimulated by ischemia and reperfusion is the xan-

thine oxidase pathway ^[15,16]. Allopurinol, a FDA-approved xanthine oxidase inhibitor for lowering uric acid concentrations in patients with gout and neoplastic diseases, inhibits the conversion of hypoxanthine into xanthine and uric acid, thereby inhibiting the production of the reactive oxygen species hydrogen peroxide (H₂O₂) and superoxide ^[17-20]. Furthermore, in high dosages, it functions as a chelator of non-protein bound iron and as a direct scavenger of hydroxyl radicals. All these mechanisms could potentially result in neuroprotective effects. Allopurinol is metabolised into the active metabolite oxypurinol, which blocks uric acid production. Its plasma concentrations are therefore used in the monitoring of gout patients ^[21,22].

A recent follow-up study of two prospective randomised controlled trials in human neonates in which the effects of *postnatally* administered allopurinol in term asphyxiated neonates were studied, showed a beneficial effect of allopurinol administration on mortality and severe disabilities at 4-8 years of age, but only in the moderately asphyxiated group ^[23]. These beneficial effects are in line with previous findings on neonatal head cooling after acute fetal hypoxia at term ^[3]. Most likely no advantage of treatment occurs anymore when the interval to the initiation of treatment is too long or when brain damage is already too severe. It is conceivable that earlier allopurinol treatment, i.e. maternally administered allopurinol during labor in case of suspected fetal hypoxia, may earlier limit the amount of free radical production and subsequent hypoxia-reperfusion injury.

For successful intrauterine neuroprotection of the hypoxic fetus during the perinatal reperfusion and reoxygenation phases, a 'candidate' drug such as allopurinol should, besides being effective in the fetus, readily cross the placenta and should not be harmful to mother and fetus.

The aim of our study was to investigate the

pharmacological applicability of allopurinol for intrauterine neuroprotection after maternal administration, by studying drug disposition (maternal plasma concentrations, cord blood concentrations) as well as drug safety (maternal and neonatal adverse events). We studied the required minimal time that was needed to obtain intrauterine target allopurinol concentrations after the end of maternal allopurinol administration, to identify the possible therapeutic time window for administration of allopurinol to the mother in labor in case hypoxia occurs.

Methods

Setting and study population

The study population was extracted from women participating in a randomized double blind placebo controlled multicenter trial performed by our group (NCT00189007) ^[5]. The trial focused on the effect of maternally administered allopurinol during fetal hypoxia on lowering brain specific biomarkers of neonatal brain damage. The study was conducted in 11 hospitals participating in the Dutch Consortium for Studies in Women's Health and Reproductivity. Pregnant women with a gestational age of at least 36 weeks and suspected intra-uterine fetal hypoxia, prompting immediate delivery, were eligible to participate in the trial. Fetal hypoxia was suspected, being an abnormal or non-reassuring fetal heart rate trace (according to FI-GO-criteria), significant ST-wave abnormalities (detected by the STAN-monitor) or abnormal fetal blood scalp sampling (pH < 7.20) ^[24]. Women bearing a child suspected of chromosomal or congenital anomalies were not eligible ^[12]. For this study, only singletons were used in the assessment of the transplacental pharmacokinetics.

Intervention

Women were randomly assigned in a double-blind fashion to receive either a single intravenous dose of 500 mg allopurinol (ALLO) in 50mL water for injection (Acepurin, AcePharmaceuticals, Zeewolde, the Netherlands) or placebo (CONT, 500mg Mannitol/50mL water for injection) administered in 10 minutes immediately prior to delivery ^[5]. The dose of allopurinol was based on a study in healthy pregnancies, performed by Boda *et al.* ^[25], and on a previously pilot-study performed by our group ^[26]. Currently the maximum allowed dose of allopurinol in a single dose in clinical use is 500 mg (as stated in the SPC of the product) ^[27].

Study drugs were stored at 20-25°C and dissolved immediately before administration. Allopurinol and placebo had exactly the same appearance; freeze dried white powder in glass vials with a pH of 11 after dilution with water for injection.

Bioanalysis

Venous cord blood and maternal blood samples were obtained immediately after birth according to protocol. Allopurinol and oxypurinol plasma concentrations were determined by the Clinical Pharmaceutical and Toxicological laboratory of the University Medical Center Utrecht. This method is used for research and routinely monitoring of patients with gout on allopurinol therapy ^[28]. Blood samples were collected in heparinized tubes and centrifuged at 3000 rpm for10 minutes. Plasma concentrations were determined using reversed- phase high-performance liquid chromatography (RP-HPLC) with UV-detection at 254 nm. The method was linear between 0.5 and 25 mg/l with a lower limit of quantification (LLoQ) of 0.2 mg/l for both compounds.

Transplacental transport

Transplacental transport of allopurinol was studied by comparing obtained maternal and cord plasma concentrations immediately after birth. The transplacental ratio is expressed as the plasma concentration in the fetus divided by the plasma concentration in the mother. Only maternal and cord blood samples that were drawn within a 30 minute interval were used to calculate the transplacental ratio, otherwise the ratio could be distorted due to ongoing drug clearance in the mothers blood after collection of the cord blood. Transplacental transport can occur via passive (e.g. diffusion), active (e.g. ATP-requiring transporters) or combined transport mechanisms (e.g. facilitated diffusion)^[29]. Only in presence of active transport, fetal drug concentration can exceed the concentrations in the mother (transplacental ratio >1), since diffusion depends on the concentration-gradient. Without active transport, transplacental ratios can only reach a maximum of 1 (e.g. equal concentration in fetus and mother).

Safety

The occurrence of (serious) adverse events and obtained plasma concentrations of allopurinol and oxypurinol were used to study maternal and fetal safety. For maternal safety of the drug only oxypurinol plasma concentrations were assessed, since only for oxypurinol a target window for uric acid lowering therapy in patients



with gout (target >5 mg/L) is known ^[30].

Ethical considerations

This study was approved by the National Central Committee on Research involving Human Subjects (CCMO – NL26516.000.09) and by Scientific Boards and Ethical Committees of the 11 participating hospitals (the University Medical Center Utrecht, the Máxima Medical Center Veldhoven, the University Medical Center Veldhoven, the University Medical Center Amsterdam, the Academic Medical Center Amsterdam, the Diakonessenhuis Utrecht, the Jeroen Bosch Hospital 's Hertogenbosch, the Maastricht University Medical Center, the Leiden University Medical Center, the Groene Hart Hospital Gouda, the Gelre Hospital Apeldoorn).

Informed consent was obtained from both future parents, antenatally at the outpatient clinic or at the labor ward. Randomization only occurred if fetal hypoxia was suspected.

Trial registration

The study is registered in the Dutch Trial Register (NTR1383) and the Clinical Trials protocol registration system (NCT00189007).

Results

From October 2009 until December 2011, 222 eligible mothers and their 224 infants were enrolled in the randomized trial. We randomly allocated 111 women (two were pregnant of a dichorionic twin) to the group that received 500 mg allopurinol; 111 women received a placebo drug (figure 8-1). In total 70 mothers received the full allopurinol dose of 500 mg. Forty-one of the 111 women did not receive the full dose because the child was born before the whole dose was administered.

Table 8-1. Characteristics of the analyzed pregnant women and neonates

Characteristic	Value (range)ª	
Mothers (n=58)		
Age (years)	33.1 (19.0 – 41.0)	
Gravida	2 (1 – 8)	
Parity	0 (0 – 3)	
Time between end of allopurinol administration and blood withdrawal (minutes)	57 (5 – 309)	
Measured allopurinol plasma concentration (mg/L)	4.3 (0.9 – 14.0)	
Measured oxypurinol plasma concentration (mg/L)	3.7 (0.8 – 8.0)	
Neonates (n=55)		
Birth weight (kg)	3.22 (1.66 – 4.38)	
Gestational age (weeks+days)	40+1 (36+1 - 42+0)	
Time between end of allopurinol administration and blood withdrawal (minutes)	6 (1 - 44)	
Measured allopurinol umbilical cord concentration (mg/L)	3.7 (0.7 – 16.5)	
Measured oxypurinol umbilical cord concentration (mg/L)	0.9 (<llq<sup>b – 3.91)</llq<sup>	

^a values are expressed as median unless specified otherwise

 $^{\rm b}{<}{\rm LLQ}$ = below lower limit of quantification, < 0.2 mg/L for allopurinol and oxypurinol



In 58 mothers who received the full dose blood samples for the analysis of allopurinol and oxypurinol plasma concentrations were available as well as all relevant information (administration times, blood withdrawal times) (figure 8-1). Allopurinol and oxypurinol plasma concentrations in cord blood were available, and all relevant information was known in 55 neonates of the mothers that received the full dose. Characteristics of the studied women and neonates are displayed in table 8-1.

Fetal drug disposition and transplacental transport

Within 5 minutes after the end of maternal allopurinol infusion target plasma concentrations of allopurinol were present in cord blood. Of all analyzed cord blood samples, 95% (52/55) had a target allopurinol plasma concentration of ≥ 2 mg/L at the time of delivery. No concentrations below the target concentration were found if the time interval between the end of allopurinol infusion and delivery was at least 7 minutes. Umbilical cord allopurinol plasma concentrations are illustrated in figure 8-2.

In 20 cases maternal and cord blood samples were drawn within half an hour of each other. The observed transplacental ratio (cord blood / mother) for allopurinol ranged from 0.23 to 0.91. In none of the cases the allopurinol concentration in cord blood exceeded the maternal allopurinol plasma concentration (for all cases the transplacental ratio was < 1). Transplacental ratios are shown in figure 8-3.

Maternal and neonatal safety

No adverse events were observed in the neonates in the allopurinol group and no serious adverse events or suspected unexpected serious adverse reactions occurred in the mothers who had received allopurinol. This is in line with the detected oxypurinol plasma levels in the maternal blood samples. Of the mothers who received allopurinol, 14% (8/58) had a target oxypurinol plasma concentration (≥5 mg/L for lowering



uric acid concentrations) corresponding with a target level for obtaining a hypo-uricemic effect (figure 8-4). These target concentrations were observed between 27 and 155 minutes after the end of the allopurinol administration. Five of the 111 women (4.5%) had a red and/or painful arm after the infusion of allopurinol, which recovered within several hours after the end of administration.

Discussion

After administration of a single intravenous dose of 500 mg allopurinol to the mother, target plasma concentrations were observed in 95% of the umbilical cord samples at the moment of delivery. Since target plasma concentrations in umbilical cord samples were already observed within 5 minutes after the end of the allopurinol administration, maternal distribution and transplacental transport occur rapidly.

Allopurinol is most likely transferred to the fetus via passive transport since transplacental ratios of >1 were not found in our studied cases. An active transplacental transport seems unlikely given the consistency of this ratio.

A dose of 500 mg allopurinol seems to be a safe dose for both mother and fetus, since no adverse events were observed except some transient redness and pain around the IV cannula in 4.5% of the participating mothers. Allopurinol hypersensitivity was not observed in our patients.

In preclinical and clinical studies, allopurinol has shown to have neuroprotective properties. An animal study in chronically instrumented fetal sheep has shown evidence of cardio- and neuroprotection after antenatal allopurinol administration to the pregnant ewe during repeated periods of ischemia ^[31,32]. In a prospective randomized placebo controlled pilot study in which allopurinol was administered to the pregnant woman when fetal asphyxia was imminent, we found an inverse correlation between levels of allopurinol and the amount of S100ß, a biomarker for brain tissue damage, in cord blood ^[25].

In an animal model of asphyxia it has been shown that the production of reactive oxygen

species after ischemia and reperfusion peaks within 10-20 minutes after the end of the ischemia-reperfusion period, depending on the duration of the ischemic period ^[2]. Pre-ischemia administration of allopurinol to rats resulted in a decreased amount of superoxide anion radicals production during the reperfusion phase and suppressed the free radical peak ^[16]. These observations emphasize that allopurinol is immediately effective as a neuroprotectant and target levels should ideally be present right at the start of ischemia and reperfusion. The fact that allopurinol has a rapid transplancental passage favors its administration as soon as signs of fetal hypoxia occur.

A previous study examining transplacental transport of 600 mg of allopurinol administered orally to women after the onset of labor showed target allopurinol levels in the cord blood of all 68 participating women, with a minimum time of 23 minutes after administration ^[26]. Corrected for bioavailability, an oral dose of 600 mg allopurinol is equal to about 500 mg of an intravenous dose. During labor gastric emptying is known to be delayed and laboring women are prone to vomiting, so oral administration of allopurinol may result in a delayed and unpredictable concentration in the fetus ^[33].

It seems, therefore, preferable to choose the intravenous administration route. The fact that in the present study target allopurinol levels were detected in cord blood within 5 minutes after the end of administration further emphasizes this point.

In one third of the participating women we did not succeed to administer the whole 500 mg dose of allopurinol before birth. In cases like these, an "escape" dose of allopurinol, administered via the umbilical cord at the resuscitation table, might be considered in future studies.

In conclusion, our study showed that intravenously administered allopurinol to the moth-



er, rapidly crosses the placenta, resulting in target neonatal allopurinol concentrations at birth and is safe to both mother and neonate, making this a promising approach for early fetal neuroprotective therapy during labor.

REFERENCES OF CHAPTER 8

1. Ferriero DM. Neonatal brain injury. N Engl J Med. 2004;351(19):1985-1995.

2. Dirnagl U, Lindauer U, Them A, et al. Global cerebral ischemia in the rat: Online monitoring of oxygen free radical production using chemiluminescence in vivo. J Cereb Blood Flow Metab. 1995;15(6):929-940.

3. Gluckman PD, Wyatt JS, Azzopardi D, et al. Selective head cooling with mild systemic hypothermia after neonatal encephalopathy: Multicentre randomized trial. Lancet. 2005;365(9460):663-670.

4. Shankaran S. Neonatal encephalopathy: Treatment with hypothermia. J Neurotrauma. 2009;26(3):437-443.

5. Whitelaw A, Thoresen M. Clinical trials of treatments after perinatal asphyxia. Curr Opin Pediatr. 2002;14(6):664-668.

 Whitelaw A. Systematic review of therapy after hypoxic-ischaemic brain injury in the perinatal period. Semin Neonatol. 2000;5(1):33-40.

7. Palmer C, Vannucci RC, Towfighi J. Reduction of perinatal hypoxic-ischemic brain damage with allopurinol. Pediatr Res. 1990;27(4 Pt 1):332-336.

8. Palmer C, Towfighi J, Roberts RL, Heitjan DF. Allopurinol administered after inducing hypoxia-ischemia reduces brain injury in 7-day-old rats. Pediatr Res. 1993;33(4 Pt 1):405-411.

9. Masaoka N, Nakajima Y, Hayakawa Y, et al. Transplacental effects of allopurinol on suppression of oxygen free radical production in chronically instrumented fetal lamb brains during intermittent umbilical cord occlusion. J Matern Fetal Neonatal Med. 2005;18(1):1-7.

10. Graham S, Day RO, Wong H, *et al.* Pharmacodynamics of oxypurinol after administration of allopurinol to healthy subjects. Br J Clin Pharmacol. 1996;41(4):299-304.

11. Kaandorp JJ, van Bel F, Veen S, *et al.* Long-term neuroprotective effects of allopurinol after moderate perinatal asphyxia: Follow-up of two randomized controlled trials. Arch Dis Child Fetal Neonatal Ed. 2012;97(3):F162-6.

12. Kaandorp JJ, Benders MJ, Rademaker CM, et al. Antenatal allopurinol for reduction of birth asphyxia induced brain damage (ALLO-trial); a randomized double blind placebo controlled multicenter study. BMC Pregnancy Childbirth. 2010;10:8. **13.** FIGO News. Guidelines for the use of fetal monitoring. Int J Gynecol Obstet. 1987;25:159-167.

14. Torrance HL, Benders MJ, Derks JB, *et al*. Maternal allopurinol during fetal hypoxia lowers cord blood levels of the brain injury marker S-100B. Pediatrics. 2009;124(1):350-357.

15. Boda D, Németh P, Kiss P, Orvos H. Treatment of mothers with allopurinol to produce therapeutic blood levels in newborns. Prenat Neonat Med. 1999;4:130-132, 133, 134.

16. Summary of product characteristics acepurin®. http://db.cbg-meb.nl/Bijsluiters/h09974.pdf. Updated 2012.

17. Syme MR, Paxton JW, Keelan JA. Drug transfer and metabolism by the human placenta. Clin Pharmacokinet. 2004;43(8):487-514.

18. Stamp LK, Barclay ML, O'Donnell JL, et al. Relationship between serum urate and plasma oxypurinol in the management of gout: Determination of minimum plasma oxypurinol concentration to achieve a target serum urate level. Clin Pharmacol Ther. 2011;90(3):392-398.

19. Ono T, Tsuruta R, Fujita M, *et al.* Xanthine oxidase is one of the major sources of superoxide anion radicals in blood after reperfusion in rats with forebrain ischemia/reperfusion. Brain Res. 2009;1305:158-167.

20. Davison JS, Davison MC, Hay DM. Gastric emptying time in late pregnancy and labor. J Obstet Gynaecol Br Commonw. 1970;77(1):37-41.

21. Derks JB, Oudijk MA, Torrance HL, *et al.* Allopurinol reduces oxidative stress in the ovine fetal cardiovascular system after repeated episodes of ischemia-reperfusion. Pediatr Res. 2010;68(5):374-380.

22. Kaandorp JJ, Derks JB, Oudijk MA, *et al.* Antenatal allopurinol reduces hippocampal brain damage after acute birth asphyxia in late gestation fetal sheep. Reprod Sci. 2013.

Chapter 9 Maternal allopurinol administration during suspected fetal hypoxia; a novel neuroprotective intervention? A multicenter randomized placebo controlled trial (the ALLO-trial)

JOEPE J. KAANDORP, MANON J.N.L. BENDERS, EWOUD SCHUIT, CARIN M.A. RADEMAKER, MARTIJN A. OUDIJK, MARTINA M. PORATH, SIDARTO BAMBAND OETOMO, MAURICE G.A.J. WOUTERS, RUURD M. VAN ELBURG, MAUREEN T.M. FRANSSEN, ARIE F. BOS, TIMO R. DE HAAN, JANINE BOON, INGE P. DE BOER, ROBBERT J.P. RIJNDERS, CORRIE J.W.F.M. JACOBS, LIESBETH H.C.J. SCHEEPERS, DANILO A.W. GAVILANES, KITTY W.M. BLOEMENKAMP, MONIQUE RIJKEN, CLAUDIA A. VAN MEIR, JEANNETTE S. VON LINDERN, ANJOKE J.M. HUISJES, SASKIA C.M.J.E.R. BAKKER, BEN WILLEM J. MOL, GERARD H.A. VISSER, FRANK VAN BEL, JAN B. DERKS

Submitted



ABSTRACT

Objective

To determine whether maternal allopurinol treatment during suspected fetal hypoxia would reduce the release of biomarkers associated with neonatal brain damage.

Study design

We performed a randomized double blind placebo controlled multicenter trial among laboring women with clinical indices of fetal hypoxia in 11 participating hospitals. Fetal hypoxia was diagnosed by an abnormal fetal heart rate, significant ST-changes on fetal ECG and/or fetal scalp pH<7.20. When immediate delivery was foreseen based on these symptoms, women were allocated to receive allopurinol 500 mg IV (ALLO) or placebo IV (CONT). Primary endpoint was S100ß in cord blood, a tissue-specific biomarker for brain damage. Analyses were performed according to the intention-to-treat principle. The difference in median (DiM) S100ß between the treatment groups was estimated with quantile regression. We also examined the difference in infants with values >75th percentile (>p75) between the groups, which is reported as Risk Ratios (RR).

Results

222 women were randomized to receive allopurinol (ALLO, n=111) or placebo (CONT, n=111). Cord S100ß was not significantly different between the two groups; 44.5 pg/mL (20.2 to 71.4) in the ALLO-group vs 54.9 pg/mL (26.8 to 94.7) in the CONT-group (DiM -7.69 (95%CI -24.9 to 9.52). The distribution of infants with an S100ß value >p75 was not significantly different either (ALLO, n=23 (23%)) vs CONT, n=25 (27%); RR_{S1008>p75} 0.85 (95%CI 0.53 to 1.4)). Post-hoc subgroup-analysis showed a marked gender difference in treatment effect of allopurinol in favor of girls. There were significantly less girls with an S100ß value >p75 in the ALLO-group (ALLO, n=5 (12%)) as compared to the placebo-treated group (CONT, n=10 (31%); RR_{S1008>p75} 0.37 (95%CI 0.14 to 0.99)). This difference was not found in boys (ALLO, n= 18 (32%) vs CONT, n=15 (25%); RR_{S1008>p75} 1.4 (95%CI 0.84 to 2.3)).

Conclusions

Maternal treatment with allopurinol during fetal hypoxia did not significantly lower neuronal damage markers in cord blood. Post-hoc analysis revealed a potential beneficial treatment effect in girls.

H ypoxic-ischemic encephalopathy is strongly associated with cerebral palsy and cognitive disability ^[1]. The prevention and treatment of fetal hypoxia is therefore an important priority in perinatal medicine. Animal and human studies have shown that brain damage not only occurs during hypoxic-ischemic events, but continues after reoxygenation and reperfusion due to production of free radicals ^[2].

At present, the only established treatment for hypoxic-ischemic encephalopathy in the clinical setting mild hypothermia, but only moderately asphyxiated infants seem to benefit with a small reduction of death and disability ^[3]. There is, therefore, a need for additional treatment options to prevent or reduce cerebral damage after perinatal asphyxia.

Allopurinol, a xanthine oxidase inhibitor and FDA-approved drug for gout, inhibits conversion of hypoxanthine into xanthine and uric acid, thereby limiting the excessive production of reactive oxygen species. Furthermore, at high dosages, the combined functions of chelating Non Protein Bound Iron (NPBI) and scavenging hydroxyl radicals might collectively result in a neuroprotective effect ^[4]. Substantial knowledge on the use, dosage, safety profile and placental transfer of allopurinol is already available, making it very interesting for translation to clinical use in perinatology. When administered postnatally to asphyxiated rats, allopurinol preserved cerebral metabolism and reduced both cytotoxic edema and perinatal hypoxic-ischemic brain damage ^[5]. Clinical trials in neonates at the NICU (neonatal intensive care unit), however, could not entirely confirm these results ^[6,7]. No advantage of neonatal treatment was seen in human infants when the interval to initiation of treatment had been too long or when brain damage was too severe ^[6,7]. It is therefore conceivable that advancing the window of treatment forward, by administering allopurinol to the mother and fetus during labor in case of suspected fetal hypoxia, may provide the opportunity to limit the generation of

excess free radical formation and subsequent reperfusion injury in the infant.

A recent study in chronically instrumented fetal sheep provided evidence of cardiac and neuroprotection following antenatal allopurinol administration to pregnant ewes during repeated periods of fetal ischemia ^[8,9]. Furthermore, a randomized placebo controlled pilot study in which we administered allopurinol to pregnant women with imminent fetal asphyxia, showed an inverse correlation between allopurinol levels and S100ß, a clinically used biomarker for brain tissue damage, in cord blood ^[10].

In the present multicenter randomized placebo controlled clinical trial we evaluated our hypothesis that maternal allopurinol treatment during fetal hypoxia reduces markers of neonatal brain damage in cord blood.

Methods

Recruitment / Eligibility criteria

This study was conducted at the delivery wards of 11 hospitals participating in the Dutch Consortium for Studies in Women's health and Reproductivity, as a double blind placebo controlled randomized trial. Pregnant women with a gestational age \geq 36 weeks and suspected intra-uterine fetal hypoxia, prompting immediate delivery, were eligible to participate in the trial. Women bearing a child suspected of chromosomal or congenital anomalies were not eligible. Informed consent was obtained antenatally at the outpatient clinic or at the labor ward. Randomization only occurred if the clinician decided to end labor and deliver the child immediately because of the suspicion of fetal hypoxia (e.g. an abnormal or non-reassuring fetal heart rate trace (according to FIGO-criteria) [11], significant ST-wave abnormalities (detected by the STAN-monitor)^[12] or abnormal fetal blood scalp sampling (pH < 7.20).

Intervention

Women were randomly assigned in a double-blind fashion to receive either a single in-

travenous dose of 500 mg allopurinol (ALLO) in 50mL water for injection (Acepurin®, AcePharmaceuticals) or placebo (CONT, 500mg Mannitol/50mL water for injection) administered, in 10 minutes, immediately prior to delivery. The dose of allopurinol was based on a study in healthy pregnancies ^[19] and on our pilot-study ^[10]. This dose of 500 mg is the maximum dosage approved in adults as a single intravenous injection, as stated in the SPC of the product [14].

AcePharmaceuticals (Zeewolde, the Netherlands) prepared the study drugs and generated the random allocation sequence in blocks of four (2:2), using a computer program ("RandList", version 1.2 (Windows XP), clinical trial version) with stratification according to participating hospitals. Every site was supplied with sequentially numbered study drug packages, labeled in a blind manner. Allocation was performed by an attending nurse who administered the contents of the first in line sequentially numbered study drug package, as soon as a consenting patient met the inclusion criteria. The nurse carrying out the study-protocol was not directly involved in the care of the woman in labor to ensure no delay in delivering the child.

Study drugs were stored at 20-25°C and dissolved immediately before administration. Allopurinol and placebo had exactly the same appearance; both were freeze-dried as white powder in glass vials with a pH of 11 after dilution. Participants, care providers and those who assessed outcomes remained blinded for allocation until the statistical analysis of the primary outcome parameter.

Outcome parameters

Primary outcome measure was S100ß in cord blood, an extensively studied biochemical marker for brain-tissue damage and subsequent neurodevelopmental disabilities ^[15-17]. S100ß is present in high concentrations in glial cells and Schwann cells. During a hypoxic-ischemic insult, this protein is released from damaged tissue in the systemic circulation. Several studies have shown that elevated plasma levels of S100ß are directly related to the development of hypoxic-ischemic encephalopathy ^[15-17].

Secondary outcome parameters were indirect markers of free radical production (8-isoprostane, neuroketal), neonatal mortality and short-term morbidity. Isoprostanes and neuroprostanes are prostaglandin-like compounds produced by free radical-induced peroxidation of arachidonic acid and docosahexaenoic acid respectively, which are highly enriched in the brain ^[18,19]. As a next step highly reactive -ketoaldehydes (neuroketals) are formed as products of free radical-induced peroxidation of neuroprostanes ^[18,20].

Data collection

Fetal venous cord blood and maternal blood samples were obtained immediately after birth. To assess the severity of fetal hypoxia, venous cord lactate and arterial cord pH were measured at the participating site. Blood samples for assessment of brain damage markers, free radical markers and plasma concentrations of allopurinol and its active metabolite oxypurinol were collected in heparinized tubes and centrifuged at 3000 rpm/10 minutes within 30 minutes after collection. The supernatant was stored at a temperature of -70°C until batch analysis at the end of the recruitment period.

All details of delivery and neonatal admission were recorded by trained research nurses and midwives at the participating site in a case record form that was accessible through a website (www.studies-obsgyn.nl/allo). Data were collected using Oracle Clinical Remote Data Capture (RDC, http://www.ctc-g.co.jp/~CTCLS/ opa/en/).

Laboratory techniques

Brain damage markers

Brain damage markers were analyzed by Haemoscan (Groningen, the Netherlands). The test for detecting S100ß is based on sandwich formation of immobilized anti-S-100 antibody, S100ß from the test sample and biotin-labeled
S100ß antibody. Label is measured after streptavidin-HRP binding and subsequent conversion of *o*-phenylenediamine (OPD) (Abnova, Taipei, Taiwan). 8-Isoprostane was determined by means of a competitive enzyme immunoassay (Cayman, Ann Arbor, MI). Neuroketals were measured by immunoassay (EIA) by competition with a specific antibody for binding to neuroketal coated microtiter plates (Haemoscan, Groningen, The Netherlands).

Allopurinol / oxypurinol

Allopurinol (ALLO) and oxypurinol (OXY) plasma concentrations were determined by the laboratory of clinical pharmacology in the University Medical Center Utrecht using reversed-phase high-performance liquid chromatography with UV-detection at 254 nm for quantification of ALLO and OXY in plasma. The method was linear between 0.5 and 25 mg/l with a lower limit of quantification of 0.2 mg/l for both compounds ^[21].

Statistical methods

Sample size

The sample size was calculated based on the primary outcome measure cord S100ß as a marker of brain damage. Based on our pilot study we expected mean values of S100ß to be lower in the 'ALLO-treated' group compared to the 'non-treated group'. Given a suspected mean value of S100ß of 1.05 ug/L (SD 0.37 ug/L) in the placebo group ^[10], this trial needed 220 patients (110 ALLO, 110 placebo) to detect a mean value of S100ß of 0.89 ug/L (SD 0.37 ug/L) in the 'AL-LO-treated' group (z- test_{2-sided}) with a power of 90% and an alpha of 0.05.

Data analysis

An independent data safety monitoring committee monitored the trial and reviewed interim results. An interim analysis was performed after including 110 women using O'Brien –Fleming alpha spending function ^[22]. This analysis was done by an independent person who was unaware of the allocation of treatment or placebo when judging the data on significant differences in neonatal mortality, neonatal NICU-admission and serious adverse events. At interim analysis a nominal value p-value < 0.005, and at the final analysis a nominal p-value < 0.049 was considered to be statistically significant.

All analyses were based on the intention-to-treat principle. We compared the primary outcome, in women exposed to allopurinol as compared with placebo. Since the distribution of S100ß was right-skewed, we calculated a difference in medians (DiM) with 95% confidence intervals (95% CIs) by using linear quantile mixed models. To account for the stratified randomization, the analysis was adjusted for center by fitting a random intercept for each center.

For continuous outcomes that were normally distributed with or without log-transformation, linear mixed models were used to calculate difference in means or difference in geometric means, respectively. Differences in dichotomous outcomes were assessed using a log-binomial mixed model to estimate a risk ratio (RR) with 95% CI, while accounting for the stratified randomization. All statistical analyses were conducted in R for Windows, version 2.15.0.

Post-hoc analyses

Three post-hoc analyses were performed. Because both S100ß and neuroketal approached physiological values, we examined the effect of allopurinol in infants with S100ß- and neuroketal values > p75. We also performed a subgroup analysis in infants with a cord pHa < 7.15. Finally, because of gender-specific treatment effects in previous studies investigating neuroprotective strategies ^[23-26], we performed a post-hoc subgroup analysis based on the sex of the infant. The treatment effect was investigated using an interaction term between the treatment and gender in the regression model. When the interaction term was significant (p < 0.05), the treatment effect was also estimated in boys and girls separately.

Results

From October 2009 through December 2011, 222 eligible mothers and their 224 infants were enrolled in the trial. We randomly allocated 111 women (two had a dichorionic twin pregnancy) to the ALLO-group and 111 to the placebo (CONT) group (figure 9-1). One mother with a twin pregnancy received the medication before birth of both children. In this case, no blood sampling was performed due to logistic problems. The other mother bearing a twin pregnancy received the drug after the delivery of the first child. Only cord blood of the second child was obtained for analysis of the primary outcome.

Of all participants 98 (87%) and 92 (83%) cord blood samples were collected in the ALLO-group and CONT-group respectively to assess the primary (S100ß) and secondary outcome parameters (neuroketal, 8-isoprostane and allopurinol levels) (figure 9-1). Missing cord blood samples,



Table 9-1. Baseline Characteristics

	ALLO	CONT
	N=111	N=111
Maternal Age, mean ± SD	32.6 ± 4.9	32.7 ± 4.3
Gestational Age, median (IQR)	40.1 (38.6-41.0)	40.6 (39.1-41.3)
Nulliparity, n (%)	84 (76)	82 (74)
Multiple gestation, n (%)	2 (1.8)	0 (0)
Diabetes Mellitus / Gestational n (%)	12 (11)	9 (8)
PIH / PE, n (%)	33 (30)	22 (20)
Smoking, n (%)	10 (9)	10 (9)
Termination of labor based on*		
Abnormal fetal heart rate trace, n (%)	103 (93)	103 (93)
Significant ST-wave abnormalities, n (%)	24 (22)	33 (30)
Fetal blood scalp sampling with pH < 7.20, n (%)	15 (14)	22 (20)
Time between infusion and delivery, median (IQR)	14.0 (7.3-26.8)	11.5 (7.0-28.5)
Tocolytics administered, n (%)	76 (68)	84 (76)
Infants [†]	113	111
Mode of delivery		
Spontaneous, n (%)	18 (16)	11 (10)
Instrumented vaginal delivery, n (%)	44 (39)	57 (52)
Cesarean Section, n (%)	49 (43)	42 (38)
Male gender child, n (%)	66 (58)	72 (65)
Birth weight (gram), mean ± SD	3217 ± 567	3329 ± 536

ALLO: Allopurinol group; CONT: Placebo group; SD: Standard Deviation; IRQ: interquartile range; PIH: Pregnancy Induced Hypertension; PE: Preeclampsia. * Numbers for this variable exceed the total number of included women, since some patients had more than one reason to terminate labor. + The ALLO-group contained two dichorionic twin pregnancies.

and subsequent missing values of brain damage markers, were mainly due to logistic problems (no appropriate blood sampling material present or a prolonged (> 30 min) time between blood sampling and centrifuging the blood).

Clinical outcome measures

Baseline characteristics of the groups were comparable (table 9-1). Overall more boys (61.6%) than girls (38.4%) were included in the trial. Boys and girls were equally distributed between the two groups. There were no significant differences in Apgar score, umbilical artery pH and Base Excess, or hypoglycemia between the groups (table 9-2). The mean umbilical artery pH in the ALLO- and CONT-group was 7.19 \pm 0.084 and 7.19 \pm 0.080 respectively. None of the children developed convulsions or hypoxic-ischemic encephalopathy, and there were no differences between the two groups in the amount of NICU-admissions (ALLO n=11 (10%) vs CONT n=9 (8%) (table 9-2)).

	ALLO N=113*	CONT N=111	Risk Ratio (95% CI) ⁺ ALLO vs CONT
Apgar score < 7 at 5 min, n (%)	8 (7.2)	6 (5.4)	1.33 (0.48 to 3.73)
Umbilical artery pH, mean ± SD	7.19 ± 0.084	7.19 ± 0.080	-0.002 (-0.025 to 0.021)§
Umbilical artery BE, mean ± SD	-6.7 ± 3.9	-6.8 ± 3.8	0.043 (-0.95 to 1.03)§
Umbilical artery pH (< 7.05) and BE (< -12) , n (%)	3 (2.7)	1 (0.9)	3.00 (0.31 to 28.7)
NICU admission, n (%)	11 (10)	9 (8)	1.22 (0.52 to 2.86)
Length of stay > 2 days, n (%)	3 (3)	4 (4)	0.75 (0.17 to 3.30)
Hypoglycemia, n (%)	18 (16)	16 (15)	1.13 (0.60 to 2.10)
Convulsions, n (%)	0 (0)	0 (0)	NA
Hypoxic Ischemic Encephalopathy, n (%)	0 (0)	0 (0)	NA

Table 9-2. Neonatal Data as a Function of Treatment Group

ALLO: Allopurinol group; CONT: Placebo group; Cl: Confidence Interval; SD: Standard Deviation; IRQ: interquartile range; BE: Base Excess; NICU: Neonatal Intensive Care Unit. * Two dichorionic twin pregnancies. + Associations are reported as risk ratios, unless stated otherwise; § Mean difference (95% Cl).

Maternal and cord allopurinol and oxypurinol concentrations

was 500 mg (interquartile range (IQR) 370 to 500 mg). Time from start of maternal administration of ALLO or placebo to birth ranged from 0 to 59 minutes (median 14 minutes (IQR

The median total dose of the study drug received by women before delivery in the ALLO-group

Figure 9-2. Maternal allopurinol/oxypurinol and cord allopurinol/oxypurinol concentrations (mg/L) at birth



Data shown as box-whisker plots (presented as median and IQR).



Cord concentrations of the primary outcome measure S100ß (pg/mL) in the allopurinol-treated and placebo-treated groups shown as box-whisker plots (presented as median and IQR).

7.3 to 27 minutes)) and 0 to 99 minutes (median 12 minutes (IQR 7.0 to 29 minutes)) respectively. There were 73 (33%) cases where the mother did not receive the total amount of study medication, mostly due to fast delivery of the child. Maternal and umbilical cord ALLO and OXY concentrations are shown in figure 9-2. Maternal ALLO and OXY concentrations ranged between 0 µg/mL and 15 µg/mL (median: 3.9 (IQR 2.1 to 5.5) μg/mL) and between 0 μg/mL and 8.0 µg/mL (median: 3.2 (IQR 2.1 to 4.5) µg/mL), respectively. Umbilical cord concentrations of ALLO and OXY ranged between 0 µg/mL and 16 μg/mL (median: 3.3 μg/mL) and between 0 μg/ mL and 6.8 µg/mL (median: 0.82 µg/mL) respectively. Target blood levels of ALLO ($\geq 2.0 \, \mu g/ml$) in cord blood were reached in 64 (72%) of the measured samples.

Cord S100ß, Neuroketal and 8-isoprostane concentrations

The primary outcome measure S100ß did not significantly differ between the two groups;

44.5 pg/mL (20.2 to 71.4) in the ALLO-group vs 54.9 pg/mL (26.8 to 94.7) in the CONT-group (DiM -7.69 (95%CI -24.9 to 9.52, figure 9-3), nor did 8-isoprostane and neuroketal (table 9-3). Since S100ß and neuroketal approached physiological values, we also examined the distribution of infants with values >p75 between the two groups (table 9-3). The number of infants with an S100ß-value >p75 did not differ between the ALLO- and placebo group (ALLO, n=23 (23%) vs CONT, n=25 (27%); (RR_{S100B2075} 0.85 (95%CI 0.53 to 1.4)). This also applied for neuroketal (ALLO, n= 22 (23%) vs CONT, n=24 (27%); (RR_{neuroketal>p75} 0.85 (0.51 to 1.4)). Post-hoc subgroup-analysis indicated a significant treatment-interaction for infants with an arterial cord pH < 7.15 for neuroketal-values >p75. Stratified analysis showed a treatment effect in this group, in which significantly fewer infants had neuroketal-values >p75 in the ALLO-group (n=3 (14%)) as compared to the placebo-group (n=11 (46%); p=0.033). No interactions were found for S100ß or 8-isoprostane.

Post-hoc analysis of gender differences

Post-hoc subgroup-analysis showed a significant treatment-gender interaction for S100ß >p75 (p=0.049) and neuroketal (p=0.027). Stratified analysis indicated a marked gender difference in treatment effect of allopurinol in favor of girls for both S100ß and neuroketal (table 9-3, figure 9-5). There were significantly less girls with an S100ß value >p75 in the ALLO-group as compared to the CONT-group (n=5 (12%) vs n=10 (31%); RR 0.37 (95%CI 0.14 to 0.99), table 9-3). This difference was not found in boys (ALLO, n=18 (32%) vs CONT, n=15 (25%); RR_{S100B>p75} 1.4 (95%CI 0.84 to 2.3)). Furthermore, the mean neuroketal value in cord blood of allopurinol-treated girls was significantly lower than in placebo-treated girls (ALLO, 18.0 pg/mL (12.1 to 26.9) vs CONT, 32.2 pg/mL (22.7 to 45.7); GMD -16.4 (95%CI -24.6 to -1.64), table 9-3, figure 9-5). Again, this difference was not found in boys (ALLO, 32.4 pg/mL (24.4 to 43.1) vs CONT, 32.1 pg/mL (24.7 to 41.8); GMD 2.38 (95%CI -10.3 to 21.3))

9 MATERNAL ALLOPURINOL ADMINISTRATION DURING SUSPECTED FETAL HYPOXIA; A NOVEL NEUROPROTECTIVE INTERVENTION? A MULTICENTER RANDOMIZED PLACEBO CONTROLLED TRIAL





Serious Adverse Events and Suspected Unexpected Severe Adverse Reactions

One Serious Adverse Event (severe postpartum hemorrhage resulting in IC-admission of the mother) and one Suspected Unexpected Severe Adverse Reaction (anaphylactic reaction after the administration of study medication and preoperative medication prior emergency Cesarean section) were reported during the study, both for patients in the placebo-group.

Discussion

In this trial, maternal administration of allopurinol during suspected fetal hypoxia did not significantly lower brain damage markers (S100ß, neuroketal and 8-isoprostane) in cord blood. Post-hoc analysis, however, revealed a potential neuroprotective effect in girls as indicated by the lower S100ß and neuroketal values in the treatment group. There is increasing evidence that gender difference plays a role in the effectiveness of pharmacologic neuroprotection after perinatal ischemia-reperfusion ^[27]. This gender difference in treatment effect might be explained by different pathways for programmed cell-death or by hormonal influences ^[23-26].

This was the first multicenter randomized clinical trial that evaluated whether a maternally administered neuroprotective drug during *acute* fetal hypoxia could reduce neuronal damage markers in cord blood. The primary outcome, S100ß, is a surrogate measure of brain damage. Whereas clinical outcome parameters would have provided us with more optimal answers to our hypothesis, the used biomarker has extensively been investigated and has proven to be directly related to the amount of neuronal damage and clinical outcome ^[15-17,28]. A follow-up study on neurodevelopmental outcome at 2 years of age using ASQ- en CBCL-questionnaires is currently running.

Despite the complexity of the study, we were able to obtain 86% of all cord samples. Levels of cord S100ß measured in this trial were slightly lower than those obtained in the pilot study ^[10]. This can be explained by the fact that perinatal hypoxia was less severe in our multicenter trial as compared to the pilot study (mean pHa Pilot: 7.16 and mean lactate Pilot 8.0 mmol/L; mean pHa RCT 7.19 and mean lactate RCT 6.9 mmol/L)^[10]. The total group of infants with imminent fetal hypoxia during labor, including the non-randomized patients, in participating hospitals had a mean cord pHa of 7.15, confirming the relatively mild hypoxia in our studied population. The treatment effect of allopurinol might potentially be larger among fetuses with more severe fetal hypoxia. This is partly confirmed by the post-hoc analysis we performed in infants with an arterial pH lower than 7.15, in which we found significantly less infants with high neuroketal values in the ALLO group, compared to the placebo group.

In contrast to postnatal neuroprotection of infants with proven asphyxia, the present trial recruited mothers with suspected fetal hypoxia. Fetal surveillance during labor remains difficult and is often false positive in detecting fetal hypoxia, which might result in substantial over-treatment. However, although the arterial pH of the infants was within the normal range at birth, elevated lactate levels did suggest there had been earlier periods of fetal hypoxia. A finding that is supported by data from Derks et al, in which arterial cord pH of fetal sheep restored earlier than lactate levels in cord blood, after a period of controlled asphyxia ^[8]. Finally, our study was not designed to examine the difference in treatment effect between genders, which limits the strength of our conclusions.

Allopurinol remains promising as a neuroprotective drug in perinatology. Our recent follow-up study of two small prospective randomized controlled trials in neonates, investigating effects of postnatal allopurinol in asphyxiated neonates, showed beneficial effects of allopurinol on mortality and severe disabilities at 4-8 years of age in moderately asphyxiated neonates ^[29]. Furthermore, maternal allopurinol treatment helped to maintain umbilical blood

	ALLO	CONT	
	(n=98)	(n=92)	
General blood measures, umbilical cord			
Uric Acid, mean ± SD	0.34 ± 0.076	0.35 ± 0.096	
Lactate, mean ± SD	6.55 ± 2.45	7.14 ± 2.91	
Troponin, geometric mean (IQR)	0.017 (0.013 to 0.021)	0.018 (0.014 to 0.024)	
Biomarkers brain damage, umbilical cord			
S100ß in pg/ml, median (IQR)	44.5 (20.2 to 71.4)	54.9 (26.8 to 94.7)	
S100ß girls	32.8 (15.3 to 53.3)	48.4 (20.3 to 109.1)	
S100ß boys	48.6 (25.2 to 96.6)	31.8 (14.4 to 83.0)	
S100ß > 75th percentile (85 pg/ml), n (%)	23 (23)	25 (27)	
S100ß girls, n (%)	5 (12)	10 (31)	
S100ß boys, n (%)	18 (32)	15 (25)	
Neuroketal in pg/ml, geometric mean (95% Cl)	25.2 (19.8 to 32.1)	32.1 (26.1 to 39.6)	
Neuroketal girls	18.0 (12.1 to 26.9)	32.2 (22.7 to 45.7)	
Neuroketal boys	32.4 (24.4 to 43.1)	32.1 (24.7 to 41.8)	
Neuroketal > 75th percentile (62.5 pg/ml), n (%)	22 (23)	24 (27)	
Neuroketal girls, n (%)	7 (17)	8 (26)	
Neuroketal boys, n (%)	15 (27)	16 (28)	
lsoprostane in pg/ml, geometric mean (95% Cl)	128.4 (108.0 to 152.7)	117.1 (93.8 to 146.3)	

Table 9-3. Important Chemical Markers Umbilical Blood

ALLO: Allopurinol group; CONT: Placebo group; CI: Confidence Interval; SD: Standard Deviation; IRQ: interquartile range. * The difference in medians was estimated using linear quantile mixed models. Quantile regression was used because for S100ß the distribution of S100ß was right-skewed and could not be normalized using e.g. a log-transformation. + P < 0.05.

flow and reduced fetal cardiac oxidative stress after ischemia-reperfusion in fetal sheep ^[8]. Fetal sheep treated with allopurinol also had significantly less neuronal damage in the hippocampal brain area ^[9]. All results from previous research indicate cardio- and neuroprotective effects of allopurinol administered to neonates or fetuses during or after perinatal hypoxia ^[5,6,8-10,29,30]. Although the present trial could only confirm (potential) neuroprotection in girls, it is remarkable that even in these very mildly hypoxic fetuses a treatment effect of allopurinol might have been revealed. Whether the neuroprotective effect is really gender specific with a direct effect of allopurinol on the brain or secondary to a cardioprotective effect remains to be elucidated.

In conclusion, our study indicates that allopurinol is safe and potentially effective in preventing hypoxic damage to the brain of newborn girls. The latter, however, is based on post-hoc

Difference in mean (95%CI) ALLO vs CONT 	Difference in geometric mean (95% CI) ALLO vs CONT	Difference in median (95% CI)* ALLO vs CONT	Risk Ratio (95% CI) ALLO vs CONT
-0.003 (-0.026 to 0.021)			
-0.60 (-1.35 to 0.16)			
	-0.001 (-0.004 to 0.003)		
		-7.69 (-24.9 to 9.52)	
		-12.5 (-50.0 to 25.0)	
		-3.10 (-21.1 to 14.9)	
			0.85 (0.53 to 1.36)
			0.37 (0.14 to 0.99) †
			1.38 (0.84 to 2.28)
	-7.53 (-15.5 to 3.62)		
	-16.4 (-24.6 to -1.64) †		
	2.38 (-10.3 to 21.3)		
			0.85 (0.51 to 1.41)
			0.66 (0.27 to 1.63)
			0.99 (0.54 to 1.81)
	6.5 (-23.7 to 47.1)		

analyses and should be more intensively investigated before being implemented in perinatal medicine. Furthermore, future research should combine fetal allopurinol treatment with postnatal neuroprotective interventions if the newborn infant appears to be really asphyxiated at birth (i.e. Apgar score < 7 at 5 minutes, pH umbilical artery < 7.05). Alternatively, in case of unexpected birth asphyxia, allopurinol treatment via the umbilical cord at the resuscitation table might result in a more optimal selection of infants needing treatment. This approach bypasses any placental problems, which may result in suboptimal placental transfer of allopurinol. And finally, any future research focusing on perinatal neuroprotection should always take gender differences into account.

ACKNOWLEDGEMENTS

We thank all (research) nurses and midwives, residents, gynecologists and neonatologists in the participating centers for the collection of data.

Trial registration

Clinical Trials protocol registration system (NCT00189007), Dutch Trial Register (NTR1383).

Funding

The trial was funded by a grant from ZonMw, the Dutch Organization for Health Research and Development (Grant number: 170991001).

Study Protocol

The study protocol was published ^[31].

REFERENCES OF CHAPTER 9

1. Low JA. Determining the contribution of asphyxia to brain damage in the neonate. J Obstet Gynaecol Res. 2004 Aug;30(4):276-8.

2. Rosenberg AA, Murdaugh E, White CW. The role of oxygen free radicals in postasphyxia cerebral hypoperfusion in newborn lambs. Pediatr Res. 1989 Sep;26(3):215-9.

3. Cluckman PD, Wyatt JS, Azzopardi D, Ballard R, Edwards AD, Ferriero DM, *et al.* Selective head cooling with mild systemic hypothermia after neonatal encephalopathy: Multicentre randomised trial. Lancet. 2005 Feb 19-25;365(9460):663-70.

4. Palmer C, Vannucci RC, Towfighi J. Reduction of perinatal hypoxic-ischemic brain damage with allopurinol. Pediatr Res. 1990 Apr;27(4 Pt 1):332-6.

5. Palmer C, Towfighi J, Roberts RL, Heitjan DF. Allopurinol administered after inducing hypoxia-ischemia reduces brain injury in 7-day-old rats. Pediatr Res. 1993 Apr;33(4 Pt 1):405-11.

6. Benders MJ, Bos AF, Rademaker CM, Rijken M, Torrance HL, Groenendaal F, *et al*. Early postnatal allopurinol does not improve short term outcome after severe birth asphyxia. Arch Dis Child Fetal Neonatal Ed. 2006 May;91(3):F163-5.

7. Van Bel F, Shadid M, Moison RM, Dorrepaal CA, Fontijn J, Monteiro L, *et al*. Effect of allopurinol on postasphyxial free radical formation, cerebral hemodynamics, and electrical brain activity. Pediatrics. 1998 Feb;101(2):185-93.

8. Derks JB, Oudijk MA, Torrance HL, Rademaker CM, Benders MJ, Rosen KG, *et al.* Allopurinol reduces oxidative stress in the ovine fetal cardiovascular system after repeated episodes of ischemia-reperfusion. Pediatr Res. 2010 Nov;68(5):374-80.

9. Kaandorp JJ, Derks JB, Oudijk MA, Torrance HL, Harmsen MG, Nikkels PG, et al. Antenatal allopurinol reduces hippocampal brain damage after acute birth asphyxia in late gestation fetal sheep. Reprod Sci. 2013 Jun 21. Epub ahaed of print.

10. Torrance HL, Benders MJ, Derks JB, *et al.* Maternal allopurinol during fetal hypoxia lowers cord blood levels of the brain injury marker S-100B. Pediatrics. 2009 Jul;124(1):350-7.

11. FIGO News. Guidelines for the use of fetal monitoring. Int J Gynecol Obstet. 1987;25:159-67.

12. Amer-Wahlin I, Arulkumaran S, Hagberg H, Marsal K, Visser GH. Fetal electrocardiogram: ST waveform analysis in intrapartum surveillance. BJOG. 2007 Oct;114(10):1191-3.

13. Boda D, Németh P, Kiss P, Orvos H. Treatment of mothers with allopurinol to produce therapeutic blood levels in newborns. Prenat Neonat Med. 1999;4:130,132, 133, 134.

14. Summary of product characteristics acepurin® [homepage on the Internet]. . 2012.

15. Gazzolo D, Visser GH, Lituania M, Sarli R, Bruschettini M, Michetti F, *et al.* S100B protein cord blood levels and development of fetal behavioral states: A study in normal and small-for-dates fetuses. J Matern Fetal Neonatal Med. 2002 Jun;11(6):378-84.

16. Michetti F, Corvino V, Geloso MC, Lattanzi W, Bernardini C, Serpero L, *et al.* The S100B protein in biological fluids: More than a lifelong biomarker of brain distress. J Neurochem. 2012 Mar;120(5):644-59.

17. Beharier O, Kahn J, Shusterman E, Sheiner E. S100B - a potential biomarker for early detection of neonatal brain damage following asphyxia. J Matern Fetal Neonatal Med. 2012 Sep;25(9):1523-8.

18. Bernoud-Hubac N, Davies SS, Boutaud O, Montine TJ, Roberts LJ,2nd. Formation of highly reactive gamma-ketoaldehydes (neuroketals) as products of the neuroprostane pathway. J Biol Chem. 2001 Aug 17;276(33):30964-70.

19. Rogers MS, Wang CC, Lau TK, *et al.* Relationship between isoprostane concentrations, metabolic acidosis, and morbid neonatal outcome. Clin Chem. 2005 Jul;51(7):1271-4.

20. Seet RC, Lee CY, Chan BP, Sharma VK, Teoh HL, Venketasubramanian N, *et al.* Oxidative damage in ischemic stroke revealed using multiple biomarkers. Stroke. 2011 Aug;42(8):2326-9.

21. van Kesteren C, Benders MJ, Groenendaal F, van Bel F, Ververs FF, Rademaker CM. Population pharmacokinetics of allopurinol in full-term neonates with perinatal asphyxia. Ther Drug Monit. 2006 Jun;28(3):339-44.

22. O'Brien PC, Fleming TR. A multiple testing procedure for clinical trials. Biometrics. 1979 Sep;35(3):549-56.

23. Nijboer CH, Kavelaars A, van Bel F, Heijnen CJ, Groenendaal F. Gender-dependent pathways of hypoxia-ischemia-induced cell death and neuroprotection in the immature P3 rat. Dev Neurosci. 2007;29(4-5):385-92.

24. Wen Y, Yang S, Liu R, *et al*. Estrogen attenuates nuclear factor-kappa B activation induced by transient cerebral ischemia. Brain Res. 2004;22;1008(2):147-54.

25. Renolleau S, Fau S, Charriaut-Marlangue C. Gender-related differences in apoptotic pathways after neonatal cerebral ischemia. Neuroscientist. 2008;14(1):46-52.

26. Spence RD, Hamby ME, Umeda E, Itoh N, Du S, Wisdom AJ, et al. Neuroprotection mediated through estrogen receptor-alpha in astrocytes. Proc Natl Acad Sci U S A. 2011 24;108(21):8867-72.

27. Hurn PD, Vannucci SJ, Hagberg H. Adult or perinatal brain injury: Does sex matter? Stroke. 2005 Feb;36(2):193-5.

28. Michetti F, Gazzolo D. S100B protein in biological fluids: A tool for perinatal medicine. Clin Chem. 2002 Dec;48(12):2097-104.

29. Kaandorp JJ, van Bel F, Veen S, Derks JB, Groenendaal F, Rijken M, *et al.* Long-term neuroprotective effects of allopurinol after moderate perinatal asphyxia: Follow-up of two randomised controlled trials. Arch Dis Child Fetal Neonatal Ed. 2012 May;97(3):F162-6.

30. Marro PJ, Hoffman D, Schneiderman R, Mishra OP, Delivoria-Papadopoulos M. Effect of allopurinol on NMDA receptor modification following recurrent asphyxia in newborn piglets. Brain Res. 1998 Mar 16;787(1):71-7.

31. Kaandorp JJ, Benders MJ, Rademaker CM, Torrance HL, Oudijk MA, de Haan TR, et al. Antenatal allopurinol for reduction of birth asphyxia induced brain damage (ALLO-trial); a randomized double blind placebo controlled multicenter study. BMC Pregnancy Childbirth. 2010 Feb 18;10:8.

Chapter 10 General discussion



I n this thesis we have investigated the use of the xanthine oxidase inhibitor allopurinol as a neuroprotective and cardioprotective agent during acute fetal hypoxia in three different species. This current chapter will discuss and summarize the results of our studies that focused on the pharmacological, neuroprotective and cardiovascular effects of allopurinol on the fetus in the late gestational rat, sheep and human pregnancy.

As explained in more detail in **Chapter 1 to 3**, current postnatal (pharmacologic) treatment of birth asphyxia-related damage to the newborn brain has a limited neuroprotective effect, even when therapy has been started within the "window of opportunity", being within 6 hours after birth. This might be due to the fact that maximal formation of reactive oxygen species (ROS) occurs within the first 30-to-60 minutes after reperfusion and reoxygenation. Moreover, already during fetal hypoxia substantial amounts of ROS are being produced.

The FDA-approved drug allopurinol is known to inhibit the formation of ROS, thereby potentially limiting free radical induced hypoxic ischemic encephalopathy. Our randomized clinical trial (ALLO trial), described in Chapter 7 and Chapter 9, investigated whether biomarkers for brain damage would be reduced after complicated hypoxic labor if the window of treatment with allopurinol would be initiated earlier, via maternal treatment, to cover the actual period of fetal hypoxia and reperfusion. Further research conducted in this thesis investigated the mechanisms of action of allopurinol in terms of xanthine oxidase inhibition (Chapter 4 and 5), placental transfer (Chapter 8) and perinatal outcome after maternal (Chapter 6 and 9) or postnatal (Chapter 3) allopurinol treatment.

Adequate placental passage of allopurinol is necessary to obtain therapeutic fetal plasma levels and subsequent potential beneficial effects on the fetus. Furthermore, the route and timing of administration, and dosage of the drug are important to obtain therapeutic levels of allopurinol at the start of reperfusion and reoxygenation, without having major negative side effects on both mother and fetus. Since allopurinol is not registered for neuroprotection during perinatal hypoxia it is difficult to define therapeutic levels of allopurinol for this specific indication. It is however known that certain plasma levels of allopurinol are able to inhibit xanthine oxidase (XO), which may result in lower levels of circulating ROS and free radicals, and subsequently less tissue damage ^[1-4].

Previous reports have confirmed that maternal treatment with allopurinol crosses the ovine [1,2], porcine ^[5] and human ^[6,7] placenta to achieve fetal plasma concentrations of allopurinol and its active metabolite oxypurinol. Our rat model showed that treatment with 30 mg.kg-1 allopurinol significantly suppressed XO-activity two hours after oral administration for at least 24 hours in all tissues examined with fetal allopurinol and oxypurinol plasma levels of 2.1 μ g ml⁻¹ and 4.0 μ g mg⁻¹ respectively (**Chap**ter 4). Although higher doses of oral allopurinol did cause higher blood levels of both allopurinol and oxypurinol, they did not result in significantly more XO inhibition but only make the occurrence of adverse side effects of the drug more likely (Chapter 5). Peak plasma-levels of allopurinol between 4 and 7 μ g ml⁻¹ were found in the fetal sheep in our sheep model after the intravenous administration of 20 mg.kg⁻¹ allopurinol to the pregnant ewe and were maintained past 1 h after the start of maternal administration ^[1]. These allopurinol levels led to improved cardiovascular function (less fetal cardiac workload during and immediately after asphyxia) ^[1] and reduced neuronal damage in the hippocampal area in the treated sheep fetuses (Chapter 6).

Placental transfer in pregnant women was first examined by Boda *et al*, who found allopurinol levels of more than 7 μ g ml⁻¹ in cord blood already 23 minutes after oral allopurinol administration of 600 mg to women going through uncomplicated labor ^[6]. In a clinical pilot study, performed by our group, median allopurinol levels of 2.0 μ g ml⁻¹ (range 0.2-7.3 μ g ml⁻¹) were seen in the cord blood of infants, 56 minutes (range 18 to 190 minutes) after a single intravenous dose of 500 mg allopurinol to the mother when suspicion of fetal hypoxia prompted immediate delivery [7]. Furthermore, an inverse correlation between the levels of allopurinol and oxypurinol and S100ß (an important tissue marker for neuronal damage) in cord blood was found. Infants with an allopurinol level $\geq 2 \ \mu g \ ml^{-1}$ in their cord blood had significantly lower S100ß values than placebo treated infants. In our subsequently performed randomized clinical trial (RCT), in 95% of all measured cord blood samples an allopurinol plasma concentration of $\geq 2 \ \mu g \ ml^{-1}$ was measured at the time of delivery after maternal allopurinol administration of 500mg intravenously (~6mg.kg⁻¹) (Chapter 8). No allopurinol concentrations lower than 2 µg ml⁻¹ were found if the time interval between the end of allopurinol infusion and delivery was at least 7 minutes. In this multicenter RCT, however, these values did not significantly lower brain damage markers (S100ß, neuroketal and 8-isoprostane) in cord blood (Chapter 9), probably due to the fact that the extent of hypoxia was relatively mild in the recruited group of patients.

In summary all performed studies investigating maternally administered allopurinol show a high correlation between paired maternal and fetal concentrations of allopurinol and oxypurinol, independent of time of administration or dose. These data indicate that placental transfer of allopurinol is fast, reliable and consistent in all species examined. Fetal levels of allopurinol and oxypurinol were always lower than maternal levels, reflecting a potential passive transfer across the placenta.

Collectively, allopurinol remains of interest as a neuroprotective drug in perinatology. Our fol-

low-up study of two small prospective randomized controlled trials in neonates (described in **chapter 3**) investigating effects of postnatal allopurinol in asphyxiated neonates, showed beneficial effects of allopurinol on mortality and severe disabilities at 4-8 years of age in moderately asphyxiated neonates ^[8]. Beneficial effects of maternal treatment with allopurinol are supported by its protective effects on umbilical blood flow ^[1,9], and on the fetal heart ^[1] and fetal brain ^[7,10] (**Chapter 6**) during and following periods of ischemia and reperfusion. No adverse effects were seen in the treated women or human infants after allopurinol administration in all performed studies (**Chapter 3, 8** and **9**) ^[8,10].

There are, however, also some drawbacks to the use of allopurinol in clinical perinatology. In our randomized clinical trial, maternal administration of allopurinol during suspected fetal hypoxia did not significantly lower brain damage markers (S100ß, neuroketal and 8-isoprostane) in cord blood (Chapter 9). Fetal surveillance during labor remains difficult and is often false positive in detecting fetal hypoxia, which might result in substantial over-treatment. Furthermore, although no adverse effects were seen in the mothers and infants in all clinically performed research on allopurinol, there are some potential adverse effects. Maternal treatment with high doses of allopurinol resulted in impairing fetal peripheral vascular reactivity to constrictor agonists ^[9] and on the fetal cardiovascular defense to acute hypoxia in fetal sheep (Chapter 5). These effects, however, were only seen after the administration of very high doses of allopurinol (100 mg/kg), which is 5 times higher than the highest clinically used dose. Finally the xanthine oxidase pathway is not the only pathway responsible for the production of reactive oxygen species. Several other pathways may promote oxidative stress, such as described in Chapter 2. A combined therapeutic approach using drugs that influence different parts of the neurotoxic cascade might therefore further improve the outcome of antioxidant therapy ^[11].

Gender differences

Post-hoc analysis of the multicenter RCT did reveal a potential neuroprotective effect in girls as indicated by the lower S100ß and neuroketal values in the treatment group. There is increasing evidence that gender differences play a role in the effectiveness of pharmacologic neuroprotection after perinatal ischemia-reperfusion ^[12]. A wide variety of clinical and experimental studies suggest gender differences in sensitivity and response to cerebral injury. In perinatology male fetuses seem to be more prone in developing (a suspicion of) fetal hypoxia than female fetuses ^[13], which is also reflected in our clinical trial in which we included significantly more male than female fetuses (62% vs 38% respectively). Not only are males more susceptible to perinatal insults, they also suffer more long-term cognitive deficits as compared to females with comparable injury ^[12]. This gender difference might be explained by different pathways for programmed cell-death or by hormonal influences ^[14-17]. Animal studies using induced neonatal hypoxia-ischemia suggest that this sex discrepancy may be caused by the presence of certain sex-specific hormones, such as circulating estradiol ^[18] and testosterone ^[19,20]. Substantially elevated testosterone present in human male fetuses may enhance neuronal excitotoxicity following a hypoxic-ischemic insult. Further evidence suggests that following such injury, male and female cells diverge in the proportional activation of caspase-dependent and caspase-independent pathways leading to apoptotic death [14,16,21,22]. Interestingly, research has revealed the sexes to differentially favor one of these two pathways, with females largely utilizing the caspase-dependent pathway and males relying more heavily on the caspase-independent pathway of cell death following an hypoxic-ischemic insult ^[14,16,21,22]. Finally, data indicate that females may possess a gene-linked advantage through a family of inhibitors of apoptosis, the most potent being X-linked IAP (XIAP) [23-^{25]}. XIAP is known to act on the caspase-dependent apoptotic pathway, and it is possible that increased expression of XIAP in females may contribute to an advantage for females following perinatal hypoxia-ischemia. Taken together, this evidence suggests interplay of hormonal differences and genetically determined apoptotic mechanisms, through which perinatal females may be provided with a level of protection against hypoxic-ischemic injury that is greater than for perinatal males ^[12].

Suggestions for future research and implications for clinical practice

In conclusion, the studies presented in this thesis indicate that maternally administered allopurinol rapidly crosses the placenta, is effective in suppressing xanthine oxidase activity, has little adverse side effects (as long as clinically compatible dosages are used) and is potentially effective in preventing hypoxic ischemic damage to the heart and brain in fetal sheep and human female neonates. The latter, however, is based on post-hoc analyses and should be more intensively investigated before being implemented in perinatal medicine.

Maternally administered allopurinol seems to have too little beneficial effects when used as a solemn therapeutic approach for perinatal asphyxia in a clinical setting, probably because fetal surveillance during labor is still too insensitive to optimally detect imminent fetal asphyxia. This may result in a suboptimal selection of patients really needing the treatment.

Future research could therefore combine fetal allopurinol (pre)treatment (prevention) with another promising fetal neuroprotective agent like melatonin ^[26], and postnatal neuroprotective interventions (repair), such as hypothermia, postnatally administered allopurinol, 2-IB ^[14], erythropoietin ^[27,28] or stem cells ^[29] if the newborn infant appears to be really asphyxiated at birth (i.e. Apgar score < 7 at 5 minutes, pH umbilical artery < 7.05). Alternatively, in case of un-

expected birth asphyxia, allopurinol treatment via the umbilical cord at the resuscitation table might result in a more optimal selection of infants needing treatment. This approach bypasses any placental problems, which may result in suboptimal placental transfer of allopurinol. And finally, any future research focusing on perinatal neuroprotective strategies should always take gender differences into account.

REFERENCES OF CHAPTER 10

1. Derks JB, Oudijk MA, Torrance HL, *et al.* Allopurinol reduces oxidative stress in the ovine fetal cardiovascular system after repeated episodes of ischemia-reperfusion. Pediatr Res. 2010;68(5):374-380.

Masaoka N, Nakajima Y, Hayakawa Y, et al. Transplacental effects of allopurinol on suppression of oxygen free radical production in chronically instrumented fetal lamb brains during intermittent umbilical cord occlusion. J Matern Fetal Neonatal Med. 2005;18(1):1-7.

3. Graham S, Day RO, Wong H, *et al.* Pharmacodynamics of oxypurinol after administration of allopurinol to healthy subjects. Br J Clin Pharmacol. 1996;41(4):299-304.

4. Nemeth N, Lesznyak T, Szokoly M, Furka I, Miko I. Allopurinol prevents erythrocyte deformability impairing but not the hematological alterations after limb ischemia-reperfusion in rats. J Invest Surg. 2006;19(1):47-56.

5. van Dijk AJ, Parvizi N, Taverne MA, Fink-Gremmels J. Placental transfer and pharmacokinetics of allopurinol in late pregnant sows and their fetuses. J Vet Pharmacol Ther. 2008;31(6):489-495.

6. Boda D, Németh P, Kiss P, Orvos H. Treatment of mothers with allopurinol to produce therapeutic blood levels in newborns. Prenat Neonat Med. 1999;4:130-132, 133, 134.

7. Torrance HL, Benders MJ, Derks JB, et al. Maternal allopurinol during fetal hypoxia lowers cord blood levels of the brain injury marker S-100B. Pediatrics. 2009;124(1):350-357.

8. Kaandorp JJ, van Bel F, Veen S, *et al.* Long-term neuroprotective effects of allopurinol after moderate perinatal asphyxia: Follow-up of two randomised controlled trials. Arch Dis Child Fetal Neonatal Ed. 2012;97(3):F162-6.

9. Herrera EA, Kane AD, Hansell JA, et al. A role for xanthine oxidase in the control of fetal cardiovascular function in late gestation sheep. J Physiol. 2012;590(Pt 8):1825-1837.

10. Kaandorp JJ, Derks JB, Oudijk MA, *et al.* Antenatal allopurinol reduces hippocampal brain damage after acute birth asphyxia in late gestation fetal sheep. Reprod Sci. 2013.

11. Robertson NJ, Tan S, Groenendaal F, *et al.* Which neuroprotective agents are ready for bench to bedside translation in the newborn infant? J Pediatr. 2012;160(4):544-552.e4.

12. Hill CA, Fitch RH. Sex differences in mechanisms and outcome of neonatal hypoxia-ischemia in rodent models: Implications for sex-specific neuroprotection in clinical neonatal practice. Neurol Res Int. 2012;2012:867531.

13. Bekedam DJ, Engelsbel S, Mol BW, Buitendijk SE, van der Palde Bruin KM. Male predominance in fetal distress during labor. Am J Obstet Gynecol. 2002;187(6):1605-1607.

14. Nijboer CH, Kavelaars A, van Bel F, Heijnen CJ, Groenendaal F. Gender-dependent pathways of hypoxia-ischemia-induced cell death and neuroprotection in the immature P3 rat. Dev Neurosci. 2007;29(4-5):385-392.

15. Wen Y, Yang S, Liu R, *et al*. Estrogen attenuates nuclear factor-kappa B activation induced by transient cerebral ischemia. Brain Res. 2004;1008(2):147-154.

16. Renolleau S, Fau S, Charriaut-Marlangue C. Gender-related differences in apoptotic pathways after neonatal cerebral ischemia. Neuroscientist. 2008;14(1):46-52.

17. Spence RD, Hamby ME, Umeda E, *et al.* Neuroprotection mediated through estrogen receptor-alpha in astrocytes. Proc Natl Acad Sci U S A. 2011;108(21):8867-8872.

18. McCarthy MM. Estradiol and the developing brain. Physiol Rev. 2008;88(1):91-124.

19. Yang SH, Perez E, Cutright J, *et al.* Testosterone increases neurotoxicity of glutamate in vitro and ischemia-reperfusion injury in an animal model. J Appl Physiol (1985). 2002;92(1):195-201.

20. Christine Knickmeyer R, Baron-Cohen S. Fetal testosterone and sex differences. Early Hum Dev. 2006;82(12):755-760.

21. Zhu C, Kang W, Xu F, *et al*. Erythropoietin improved neurologic outcomes in newborns with hypoxic-ischemic encephalopathy. Pediatrics. 2009;124(2):e218-26.

22. Lang JT, McCullough LD. Pathways to ischemic neuronal cell death: Are sex differences relevant? J Transl Med. 2008;6:33-5876-6-33.

23. Deveraux QL, Takahashi R, Salvesen GS, Reed JC. X-linked IAP is a direct inhibitor of cell-death proteases. Nature. 1997;388(6639):300-304.

24. Deveraux QL, Roy N, Stennicke HR, *et al.* IAPs block apoptotic events induced by caspase-8 and cytochrome c by direct inhibition of distinct caspases. EMBO J. 1998;17(8):2215-2223.

25. Deveraux QL, Leo E, Stennicke HR, Welsh K, Salvesen GS, Reed JC. Cleavage of human inhibitor of apoptosis protein XIAP results in fragments with distinct specificities for caspases. EMBO J. 1999;18(19):5242-5251.

26. Miller SL, Yan EB, Castillo-Melendez M, Jenkin G, Walker DW. Melatonin provides neuroprotection in the late-gestation fetal sheep brain in response to umbilical cord occlusion. Dev Neurosci. 2005;27(2-4):200-210.

27. McPherson RJ, Juul SE. Erythropoietin for infants with hypoxic-ischemic encephalopathy. Curr Opin Pediatr. 2010;22(2):139-145.

28. Fan X, van Bel F, van der Kooij MA, Heijnen CJ, Groenendaal F. Hypothermia and erythropoietin for neuroprotection after neonatal brain damage. Pediatr Res. 2013;73(1):18-23.

29. van Velthoven CT, Kavelaars A, Heijnen CJ. Mesenchymal stem cells as a treatment for neonatal ischemic brain damage. Pediatr Res. 2012;71(4 Pt 2):474-481.

Chapter 11 Summary Nederlandse samenvatting



SUMMARY

espite advances in obstetric practice, acute intra-partum fetal hypoxia-ischemia (HI) or birth asphyxia remains one of the most common forms of fetal stress. Hypoxia-ischemia damages the susceptible fetal brain and especially reperfusion of previously ischemic brain tissue is recognized as an important mechanism for substantial additional brain injury due to the excessive formation of free radicals known as reactive oxygen species (ROS). The resulting hypoxic-ischemic encephalopathy (HIE) is associated with the development of cerebral palsy and cognitive disabilities later in life.

As explained in more detail in Chapter 1 to 3, current postnatal (pharmacologic) treatment of birth asphyxia-related damage to the newborn brain has a limited neuroprotective effect, even when therapy has been started within the "window of opportunity", being within 6 hours after birth. This might be due to the fact that maximal formation of ROS occurs within the first 30-to-60 minutes after reperfusion and reoxygenation. Moreover, already during fetal hypoxia, substantial amounts of ROS are being produced. We therefore postulated that antenatal (i.e. maternal) pharmacological neuroprotection of the fetus, combined with postnatal neuroprotective strategies (such as hypothermia), might be a more optimal approach to prevent this free radical induced brain damage.

Chapter 2 summarizes the molecular mechanisms underlying early reperfusion-reoxygenation damage and focuses on the most investigated pharmacological agents (phenobarbital, vitamin C and E, allopurinol, melatonin and xenon) to be given antenatally to the mother to neuroprotect the hypoxic fetus. One potential strategy to diminish the detrimental effects of fetal hypoxia would be to decrease the associated excessive generation of ROS from stimulated pro-oxidant mechanisms within the cell, such as the xanthine oxidase pathway. Administration of the xanthine oxidase inhibitor allopurinol reduces the production of superoxide free radical formation, thereby limiting the amount of hypoxia-reperfusion damage. Furthermore, allopurinol also has a non-protein bound iron (pro-radical) chelating and direct free radical (hydroxyl) scavenging effect, especially when high dosages are administered.

Chapter 3 describes the long-term effects of neonatal allopurinol-treatment in infants suffering from moderate to severe birth asphyxia. Fifty-four term infants were included when suffering from moderate to severe birth asphyxia in two previously performed trials. Infants received either two doses of 20 mg/kg allopurinol intravenously starting within 4 hours after birth (with an interval of 12 hours) or served as controls. Children, who survived and could be tested, were assessed with the Wechsler Preschool and Primary Scales of Intelligence test (WPPSI) or Wechsler Intelligence Scale for Children (WISC) and underwent a neurological examination. The effect of allopurinol on severe adverse outcome (defined as mortality or severe disability at the age of 4-8 years) was examined in the total group of asphyxiated infants and in a predefined subgroup of moderately asphyxiated infants (based on the amplitude integrated electroencephalogram). The mean age during follow-up (n = 23) was 5 years and 5 months (SD 1y 2mo). There were no differences in long-term outcome between the allopurinol treated infants and controls. However, subgroup analysis of the moderately asphyxiated group showed significantly less severe adverse outcome in the allopurinol treated infants compared to controls (25% vs 65%; RR 0.40 (95%CI 0.17 to 0.94)). The reported data may suggest a (neuro)protective effect of neonatal allopurinol-treatment, but only in moderately asphyxiated infants. These beneficial effects are in line with previous findings on neonatal head cooling after acute fetal hypoxia at term by Gluckman *et al.* Most likely no advantage of treatment occurs anymore when the interval to the initiation of treatment is too long or when brain damage is already too severe. It is conceivable that earlier allopurinol treatment, i.e. maternally administered allopurinol during labor in case of suspected fetal hypoxia, may earlier limit the amount of free radical production and subsequent hypoxia-reperfusion injury.

The aim of this thesis was to investigate the applicability of maternally administered allopurinol for fetal neuroprotection and to define strategies for future practice.

PART I: ANIMAL STUDIES

In Chapter 4 the capacity of allopurinol to be taken up by the mother, cross the placenta, rise to therapeutic levels and suppress xanthine oxidase activity in the fetus was established in a rodent model. On day 20 of pregnancy, Wistar dams were given 30 or 100 mg.kg⁻¹ allopurinol orally. Maternal and fetal plasma allopurinol and oxypurinol concentrations were measured, and xanthine oxidase activity in the placenta and maternal and fetal tissues determined. There were significant strong positive correlations between maternal and fetal plasma allopurinol (r=0.97, P<0.05) and oxypurinol (r=0.88, P<0.05) levels. Under baseline conditions, maternal heart (2.18±0.62mU.mg⁻¹), maternal liver (0.29±0.08mU.mg⁻¹), placenta (1.36±0.42mU.mg⁻¹), fetal heart (1.64±0.59mU.mg⁻¹) and fetal liver (0.14±0.08mU.mg⁻¹) samples all showed significant xanthine oxidase activity. This activity was suppressed in all tissues 2h after allopurinol administration and remained suppressed 24h later (P<0.05), despite allopurinol and oxypurinol levels returning towards baseline. The data provide a rodent model of xanthine oxidase inhibition in the mother, placenta and fetus, allowing investigation of the role of xanthine oxidase-derived reactive oxygen species in the maternal, placental and fetal physiology during healthy and complicated pregnancy.

Hypoxia is a common challenge to the fetus, promoting a physiological defense to redistribute blood flow towards the brain and away from peripheral circulation. During acute hypoxia, reactive oxygen species (ROS) interact with nitric oxide (NO) to provide an oxidant tone, which contributes to the mechanisms redistributing the fetal cardiac output, however the source of ROS responsible for this defence is unknown. In Chapter 5 we investigated whether ROS derived from xanthine oxidase (XO) contribute to the fetal peripheral vasoconstrictor response to hypoxia via interaction with NO-dependent mechanisms. Fifteen pregnant ewes and their fetuses were surgically prepared for long-term recording at 118 days of gestation (term is approximately 145 days). After 5 days recovery, mothers were infused i.v. for 30 min with either vehicle, low (30 mg.kg⁻¹) or high (150mg.kg⁻¹) allopurinol with or without fetal NO blockade. Allopurinol inhibited the increase in fetal plasma uric acid and suppressed the fetal femoral vasoconstrictor, glycemic and lactate acidemic responses during hypoxia (all P<0.05), effects that were restored to control levels with fetal NO blockade. The data provide evidence for the activation of fetal XO in vivo during hypoxia and for XO-derived ROS in contributing to the fetal peripheral vasoconstriction, part of the fetal defense to hypoxia. The data in this study are of significance to the understanding of the physiological control of the fetal cardiovascular system during hypoxic stress. The findings are also of clinical relevance in the context of obstetric trials, if high doses of allopurinol are being administered to pregnant women when the fetus shows signs of hypoxic distress.

In **Chapter 6** we tested the hypothesis that maternal treatment with allopurinol during fetal asphyxia limits ischemia-reperfusion (I/R) damage to the fetal brain in ovine pregnancy. The I/R challenge was induced by 5 repeated measured compressions of the umbilical cord each lasting 10 min, in chronically-instrumented fetal sheep at 0.8 of gestation. Relative to control fetal brains, the I/R challenge induced significant neuronal damage in the fetal hippocampal cornu ammonis zone 3 and 4. Maternal treatment with allopurinol during the I/R challenge restored the fetal neuronal damage towards control scores. The results of this study suggest that maternal treatment with allopurinol offers potential neuroprotection to the fetal brain in the clinical management of perinatal asphyxia.

PART II: CLINICAL TRIAL

Chapter 7 provides a detailed description of the study protocol of our clinical trial (the ALLO-trial). The ALLO-trial was conducted as a randomized double blind placebo controlled multicenter trial among laboring women with suspected fetal hypoxia in 11 participating hospitals. Fetal hypoxia was suspected in case of an abnormal fetal heart rate, significant ST-changes on fetal ECG or fetal scalp pH<7.20. When immediate delivery was foreseen based on these symptoms, women were allocated to receive allopurinol 500 mg IV or placebo IV. Primary outcome measures were the amount of \$100ß (a marker for brain tissue damage) and the severity of oxidative stress (measured by isoprostane, neuroprostane), both measured in umbilical cord blood. Secondary outcome measures were neonatal mortality, serious composite neonatal morbidity and long-term neurological outcome. Furthermore pharmacokinetics and pharmacodynamics were investigated.

Chapter 8 describes drug disposition (maternal plasma concentrations, cord blood concentrations) and drug safety (maternal and fetal adverse events). Data from the randomized double blind multicenter trial were used. In 55 of 111 women in the intervention arm of the study the total dose of allopurinol was administered to the mother before delivery of the child, and a cord blood sample was available. Within 5 minutes after the end of maternal allopurinol infusion target plasma concentrations of allopurinol of ≥ 2

mg/L were present in cord blood. Of all analyzed cord blood samples, 95% (52/55) had a target allopurinol plasma concentration at the moment of delivery. No adverse events were observed in the neonates. Two mothers had a red and/or painful arm during infusion.

In Chapter 9 the main outcomes of the clinical ALLO-trial are described. In the ALLO-trial 222 women were randomized to receive allopurinol (ALLO, n=111) or placebo (CONT, n=111). Cord S100ß was not significantly different between the two groups; 44.5 pg/mL (20.2 to 71.4) in the ALLO-group vs 54.9 pg/mL (26.8 to 94.7) in the CONT-group (DiM -7.69 (95%CI -24.9 to 9.52)). The distribution of infants with an S100ß value >p75 was not significantly different either (ALLO, n=23 (23%) vs CONT, n=25 (27%); RR_{S100B>p75} 0.85 (95%CI 0.53 to 1.4)). Post-hoc subgroup-analysis showed a marked gender difference in treatment effect of allopurinol in favor of girls. There were significantly less girls with an S100ß value >p75 in the ALLO-group (ALLO, n=5 (12%)) as compared to the placebo-treated group (CONT, n=10 (31%); RR_{S100B>p75} 0.37 (95%CI 0.14 to 0.99)). This difference was not found in boys (ALLO, n=18 (32%) vs CONT, n=15 (25%); $RR_{S100B>p75}$ 1.4 (95%CI 0.84 to 2.3)). In conclusion, maternal treatment with allopurinol during fetal hypoxia did not significantly lower neuronal damage markers in cord blood, probably due to the fact that fetal hypoxia was relatively mild in the investigated population. Posthoc analysis however revealed a potential beneficial treatment effect in girls.

In conclusion (as stated in the general discussion in **Chapter 10**), the studies presented in this thesis indicate that maternally administered allopurinol rapidly crosses the placenta, is effective in suppressing xanthine oxidase activity, has little adverse side effects (as long as clinically compatible dosages are used) and is potentially effective in preventing hypoxic ischemic damage to the heart and brain in fetal sheep and human female neonates. The latter, however, is based on post-hoc analyses and should be more intensively investigated before being implemented in perinatal medicine.

Maternally administered allopurinol seems to have too little beneficial effects when used as a solemn therapeutic approach for perinatal asphyxia in a clinical setting, probably because fetal surveillance during labor is still too insensitive to optimally detect imminent fetal asphyxia. This may result in a suboptimal selection of patients really needing the treatment.

Future research could therefore combine fetal allopurinol (pre)treatment (prevention) with another promising fetal neuroprotective agent like melatonin, and postnatal neuroprotective interventions (repair), such as hypothermia, postnatally administered allopurinol, 2-IB, erythropoietin or stem cells if the newborn infant appears to be really asphyxiated at birth (i.e. Apgar score < 7 at 5 minutes, pH umbilical artery < 7.05). Alternatively, in case of unexpected birth asphyxia, allopurinol treatment via the umbilical cord at the resuscitation table might result in a more optimal selection of infants needing treatment. This approach bypasses any placental problems, which may result in suboptimal placental transfer of allopurinol. And finally, any future research focusing on perinatal neuroprotective strategies should always take gender differences into account.

NEDERLANDSE SAMENVATTING

Hersenschade ten gevolge van asfyxie (zuurstoftekort) bij de pasgeborene is een ernstige complicatie binnen de verloskunde. Pasgeborenen die blootgesteld zijn aan asfyxie kunnen direct of op latere leeftijd ernstige problemen ervaren in hun neurologische ontwikkeling. In milde gevallen kunnen zich leerproblemen, gedragsproblemen of geheugenproblemen voordoen. In ernstigere gevallen kan er zelfs sprake zijn van sterfte, spasticiteit of mentale retardatie. Herkennen, voorkomen en behandelen van zuurstofproblemen tijdens de bevalling is daarom van groot belang.

Het ontstaan van hersenschade

Op het moment van asfyxie wordt een reeks chemische reacties in gang gezet in het lichaam welke in hoofdstuk 2 meer in detail worden besproken. Kort gezegd komt er ten gevolge van asfyxie een grote hoeveelheid aan schadelijke stoffen vrij, vrije radicalen genaamd. Vrije radicalen zorgen voor schade aan alles wat ze tegenkomen, zoals celmembranen, celkernen en DNA. Dit kan overal in het lichaam gebeuren, maar vooral het hart en de hersenen van pasgeborenen zijn hiervoor erg gevoelig. Op het moment van zuurstofgebrek zelf kan er al hersenschade ontstaan, maar als er daarna weer zuurstofrijk bloed het hart of hersenen instoomt (reperfusie) wordt de grootste hoeveelheid aan deze vrije radicalen gevormd. Logischerwijs geldt; hoe meer schade er ontstaat, hoe ernstiger de neurologische gevolgen op latere leeftijd waarschijnlijk zullen zijn.

Tot op heden is koeling van de pasgeborene op de Neonatale Intensive Care Unit (NICU) de enige therapie waarvan wetenschappelijk bewezen is dat het de mate van hersenschade na asfyxie kan beperken. Deze behandeling wordt echter pas ingezet na de bevalling, wanneer het grootste deel van de schadelijke stoffen al is gevormd.

Allopurinol

Allopurinol is een geneesmiddel dat al sinds lange tijd gegeven wordt voor jicht, een gewrichtsaandoening die wordt veroorzaakt door verhoogde urinezuurspiegels in het bloed. Naast het verlagen van urinezuurspiegels kan allopurinol ook de vorming van vrije radicalen voorkómen door de remming van een enzym; genaamd xantine-oxidase, en op die manier mogelijk hersenschade bij de pasgeborene beperken.

In de afgelopen 15-20 jaar komen uit verschillende onderzoeken steeds meer aanwijzingen dat allopurinol inderdaad hersenschade ten gevolge van asfyxie kan verminderen. Bij de mens werd het middel voor deze indicatie in de jaren negentig van de vorige eeuw als eerste onderzocht door het na de geboorte te geven aan ernstig asfyctische kinderen. Bij dit onderzoek werd een verminderde radicalenproductie en behoud van hersendoorbloeding gezien bij de kinderen die na de geboorte allopurinol kregen toegediend. Toen deze studie in 2003 met meer patienten werd herhaald bleken er echter geen significante verschillen zichtbaar op korte termijn uitkomsten. Een mogelijke verklaring hiervoor is dat het gemiddelde tijdstip van toediening (3 uur na de geboorte) te laat was om een significante daling in vrije radicalen productie te kunnen bewerkstelligen. In hoofdstuk 3 worden de resultaten beschreven van het vervolgonderzoek bij deze kinderen op 4-8 jarige leeftijd. De kinderen die nog leefden en konden worden getest werden getest met de Wechsler Preschool and Primary Scales of Intelligence test (WPP-SI) of Wechsler Intelligence Scale for Children (WISC), afhankelijk van hun leeftijd op dat moment. Het effect van allopurinol op ernstige morbiditeit (sterfte of ernstige neurologische beperkingen op 4-8 jarige leeftijd) werd onderzocht in de gehele studiegroep (n=54) en in een voorgespecificeerde groep van matig ernstig asfyctische kinderen. De gemiddelde leeftijd tijdens follow-up (n=23) was 5 jaar en 5 maanden (SD 1 jaar, 2 maanden). Er waren geen verschillen in lange termijn uitkomsten tussen de kinderen die allopurinol hadden gekregen en de kinderen die het medicijn niet hadden gehad. In de subgroep van matig ernstig asfyctische kinderen werd wel een significant effect van allopurinol gevonden op het verminderen van sterfte of ernstige neurologische beperkingen (25% vs 65%; RR 0.40, 95%CI 0.17-0.94). Deze uitkomsten suggereren een mogelijk neuroprotectief effect van neonatale behandeling met allopurinol, maar alleen in matig ernstig asfyctische kinderen. Deze uitkomsten zijn vergelijkbaar met de studies naar neonatale koeling van de onderzoeksgroep van Dr. Gluckman. Waarschijnlijk heeft de ingezette therapie onvoldoende effect als de behandeling te laat wordt ingezet of de hersenschade al te ver gevorderd is.

Om de allopurinol zo vroeg mogelijk bij het kind te kunnen krijgen ontstond de gedachte om het aan de moeder te geven op het moment van een verdenking op foetale nood. Op deze manier komt het middel, via de placenta, al bij het kind nog voordat het geboren is.

Met dit idee werd van 2005 tot 2007 een pilotonderzoek verricht. Hierbij werden 53 vrouwen onderzocht bij wie tijdens de bevalling een dusdanige verdenking op foetale nood bestond dat acuut ingrijpen noodzakelijk was door middel van een keizersnede danwel vacuumpompof tangverlossing. De helft van deze vrouwen kreeg op dat moment 500 mg allopurinol i.v. en de andere helft kreeg een niet werkzaam medicijn (placebo). Uit dit pilotonderzoek kwam ten eerste naar voren dat allopurinol zeer snel de placenta passeert. Ten tweede bleek er een significant verband te bestaan tussen de hoogte van de spiegels van allopurinol en de waarde van een marker voor hersenweefselschade (S100ß) in het navelstrengbloed. Hoe hoger de allopurinol-spiegels waren, hoe lager de waarde van S100ß.

Dierexperimentele studies

Hoofdstuk 4 tot en met 6 beschrijven dierexperimentele studies die zijn uitgevoerd aan de Universiteit van Cambridge, Engeland. In hoofdstuk 4 werden verschillende eigenschappen van allopurinol onderzocht bij zwangere ratten. Zowel de mate van opname van allopurinol door de moeder (wanneer oraal toegediend), de mate van placenta-passage, het bereiken van therapeutische spiegels in de foetus als de mate van remming van xanthine-oxidase bij de foetus werden onderzocht. Op de 20^e dag van de zwangerschap (normale zwangerschapsduur bij raten is 21 dagen), kregen zwangere ratten 30 of 100 mg.kg⁻¹ allopurinol oraal toegediend. Maternale en foetale plasma allopurinol en oxypurinol spiegels werden gemeten en xanthine oxidase activiteit werd bepaald in de placenta en in maternale en foetale weefsels. Er waren significant positieve correlaties tussen maternale en foetale allopurinol (r=0.97, P<0.05) waarden. Onder normale omstandigheden was er sprake van verhoogde xanthine oxidase activiteit in het maternale hart (2.18±0.62mU.mg⁻¹), de maternale lever (0.29±0.08mU.mg⁻¹), de placenta (1.36±0.42mU. mg¹), het foetale hart (1.64±0.59mU.mg¹) en de foetale lever (0.14±0.08mU.mg⁻¹). Deze activiteit was onderdrukt in alle weefsels 2 uur na allopurinol toediening (zowel bij 30 als bij 100 mg.kg⁻¹ allopurinol) en bleef onderdrukt tot minstens 24 uur later (P<0.05), ondanks dalende allopurinol spiegels.

Hypoxie is een veel voorkomend probleem voor het ongeboren kind, waarbij er een fysiologisch compensatie mechanisme is door bloeddoorstroming van verschillende organen te verplaatsen (redistribueren) van de perifere circulatie (bijvoorbeeld huid en spieren) naar meer belangrijke organen, zoals de hersenen. Tijdens acute hypoxie zorgen vrije zuurstofradicalen tezamen met stikstof oxide voor een veranderde vaatwandtonus die bijdraagt aan het onderliggende mechanisme van redistributie van het bloed. De bron van deze vrije zuurstof radicalen

is echter nog grotendeels onbekend. Hoofdstuk 5 laat zien dat vrije radicalen die vrijkomen bij de omzetting van hypoxanthine in xanthine door xanthine oxidase (XO), bijdragen aan de foetale perifere vasoconstrictie (samenknijpen van vaten) bij hypoxie via stikstofoxide-afhankelijke mechanismen. Vijftien zwangere schapen en hun ongeboren lammeren werden operatief voorzien van meetapparatuur op dag 118 van de zwangerschap (normale zwangerschapsduur bij schapen is ongeveer 145 dagen). Na een herstelperiode van 5 dagen, kregen de moeders ofwel een placebo medicijn, ofwel een "lage" dosis allopurinol (30 mg.kg-1) ofwel een "hoge" dosis allopurinol (150mg.kg-1) intraveneus toegediend, met of zonder blokkade van stikstofoxide. Allopurinol remde de stijging van foetale urinezuurspiegels en onderdrukte foetale femorale vasoconstrictie, glycemische en zuur-base reacties tijdens hypoxie (allen P<0.05). Deze effecten werden teniet gedaan wanneer foetaal stikstof oxide werd geblokkeerd. De data in deze studie leveren bewijs voor de activatie van foetale XO tijdens hypoxie bij schapen. Het toont tevens aan dat de door XO veroorzaakte vrije radicalen productie mede verantwoordelijk is voor de foetale perifere vasoconstrictie die onderdeel is van de foetale beschermingsmechanismen bij hypoxie. De data zijn van groot belang in het begrijpen van de fysiologische controle mechanismen van het foetale cardiovasculaire systeem tijdens hypoxie en zijn daarnaast belangrijk bij het verrichten van klinische trials wanneer hoge doses van allopurinol worden toegediend aan de zwangere vrouw als de foetus tekenen van hypoxie vertoont.

In **hoofdstuk 6** werd, opnieuw bij zwangere schapen, de hypothese getoetst dat maternale behandeling met allopurinol tijdens foetale asfyxie reperfusieschade aan het foetale brein kan verminderen. Episodes van ischemie en reperfusie werden bewerkstelligd door 10 foetale lammeren op dag 118 van de zwangerschap, vijf opeenvolgende navelstrengcompressies van 10 minuten te laten ondergaan. Ten opzichte van 6 controle-lammeren veroorzaakte ischemie en reperfusie significant meer histologische neuronale schade in cornu ammonis zone 3 en 4 van de foetale hippocampus (een deel van het brein dat o.a. verantwoordelijk is voor het geheugen). Maternale behandeling met allopurinol tijdens de episodes van ischemie en reperfusie brachten de gevonden histologische scores voor neuronale schade weer terug naar controle waarden. De resultaten van deze studie suggereren wederom dat maternale behandeling met allopurinol een potentieel neuroprotectief effect heeft op het foetale brein ten tijde van perinatale asfyxie.

De humane klinische studie – de ALLO-trial

Met de hoopgevende uitkomsten van het dierexperimentele onderzoek en het eerder uitgevoerde humane pilotonderzoek werd besloten het humane pilotonderzoek op landelijk niveau te herhalen, zodat er met grotere aantallen patienten en in verschillende patiëntengroepen uitspraken kunnen worden gedaan.

De ALLO-trial, een dubbelblind gerandomiseerde placebo gecontroleerde multicenter trial includeerde patiënten van oktober 2009 tot december 2011 in 11 verschillende ziekenhuizen in Nederland. De onderzoeksmethoden waren hetzelfde als bij het pilotonderzoek en worden in detail beschreven in **hoofdstuk** 7. Op het moment dat een patiënte de inclusiecriteria bereikte was er sprake van een dusdanige verdenking op foetale hypoxia dat werd gestreefd naar de direct geboorte van het kind (abnormaal cardiotocogram (CTG), significant STAN-event of een micro bloed onderzoek (MBO) met een pH < 7.20), en werd 500 mg allopurinol of een placebo medicijn i.v. toegediend via een spuitenpomp. Nadat het kind geboren was, werd navelstrengbloed afgenomen voor het meten van de S100ß -waarde (een bekende biochemische marker voor hersenweefselschade), de mate van vrije radicalenproductie (neuroprostaan en isoprostaan) en de allopurinol-spiegels.

In hoofdstuk 8 onderzochten we de placenta-passage van allopurinol bij de mens door middel van het bepalen van maternale plasma concentraties van allopurinol en allopurinol concentraties in navelstrengbloed. Daarnaast beoordeelden we de veiligheid van het middel door te kijken naar ongewenste effecten bij moeder en/of kind. Data uit de ALLO-trial werden hiervoor gebruikt: in de trial kregen 70 van de 111 vrouwen de totale hoeveelheid allopurinol toegediend vóór de geboorte van het kind. Van 55 van deze 111 moeder-kind paren die allopurinol kregen toegediend was navelstrengbloed beschikbaar voor het bepalen van de allopurinol spiegels. In deze groep werden binnen 5 minuten na het einde van de maternale allopurinol infusie streef-spiegels van allopurinol in het navelstrengbloed gevonden van ≥2 mg/L. Van allopurinolspiegels ≥2 mg/L is aangetoond dat het XO-activiteit kan remmen. Van al de geanalyseerde samples had 95% (52/55) een waarde boven deze streef-spiegels ten tijde van de geboorte van het kind. Omdat allopurinol al geruime tijd wordt gegeven voor jicht weten we veel over de bijwerkingen van het medicijn. De belangrijkste bijwerkingen die op kunnen treden zijn gewoonlijk zeldzaam en meestal van lichte aard. De meest voorkomende reacties zijn huidreacties, zoals jeuk en roodheid. Nooit eerder werden ernstige bijwerkingen beschreven bij een éénmalige dosis en ook in deze trial werden geen nadelige effecten waargenomen bij de neonaten. Twee moeders hadden een rode en/of pijnlijke arm tijdens de infusie van allopurinol.

In **hoofdstuk 9** worden de uitkomsten van de ALLO-trial beschreven. In totaal werden 222 vrouwen geincludeerd, van wie 111 allopurinol kregen toegediend en 111 vrouwen een placebo-medicijn. De waarde van de hersenschade marker S100ß was niet significant verschillend tussen de twee groepen (ALLO, 44.5 pg/ mL (20.2 tot 71.4) vs placebo, 54.9 pg/mL (26.8 tot 94.7); DiM -7.69 (95%CI -24.9 tot 9.52)). Ook het aantal kinderen met een S100ß waarden >p75 was niet significante verschillend tussen

de twee behandelgroepen (ALLO, n=23 (23%) vs placebo, n=25 (27%); RR_{S100R>p75} 0.85 (95%CI 0.53 tot 1.4)). Post-hoc subgroep analyse toonde een opvallend sexe verschil in behandeleffect in het voordeel van meisjes. Er waren significant minder meisjes met een S100ß waarde >p75 in de ALLO-groep (ALLO, n=5 (12%)) ten opzichte van de placebo-groep (placebo, n=10 (31%); RR_{S100B>p75} 0.37 (95%CI 0.14 to 0.99)). Dit verschil werd niet terug gezien bij jongens (RR_{S100B>p75} 1.4 (95%CI 0.84 to 2.3)). Samengevat verlaagde maternale behandeling met allopurinol ten tijde van een verdenking op foetale hypoxie niet significant neuronale hersenschade markers in het navelstrengbloed. Post-hoc analyse toonde echter wel een mogelijk voordelig effect van maternale behandeling van allopurinol bij meisjes.

Zoals wordt beschreven in de "general discussion" in **hoofdstuk 10** laten de studies in dit proefschrift zien dat maternale toediening van allopurinol snel de placenta passeert, effectief is in het remmen van xanthine oxidase en weinig nadelige effecten heeft (zolang klinisch compatibele doseringen worden aangehouden). Ook lijkt het middel potentieel effectief in het verminderen van schade aan het hart en de hersenen bij foetale schapen en vrouwelijke neonaten. Dit laatste is echter gebaseerd op post-hoc analyses en zal eerst meer uitgebreid moeten worden onderzocht alvorens het toe te passen in de perinatologische zorg.

Maternale allopurinol lijkt te weinig significant voordelige effecten te hebben wanneer het wordt toegepast als op zichzelf staande therapeutische benadering voor perinatale asfyxie in de klinische setting. Mogelijk is dit het gevolg van de nog altijd relatief insensitieve foetale bewakingsmethoden die resulteren in een suboptimale selectie van kinderen die daadwerkelijk de behandeling nodig hebben.

Toekomstig onderzoek zou zich daarom kunnen richten op het combineren van foetale (voor-)behandeling op het moment van een verdenking op foetale nood met bijvoorbeeld allopurinol en een ander veelbelovend foetaal neuroprotectief middel als melatonine (preventie), in combinatie met neonatale interventies (herstel), zoals hypothermie, postnataal toegediende allopurinol, 2-IB, erythropoietine of stamcellen op het moment dat de pasgeborene daadwerkelijk tekenen van ernstige asfyxie vertoont (bijvoorbeeld een Apgar score < 7 na 5 minuten en een arteriele navelstreng pH < 7.05). In het geval van onverwachte asfyxie na de geboorte van het kind kan worden overwogen directe behandeling met allopurinol toe te passen op de resuscitatie-tafel door het toedienen van allopurinol via de navelstreng. Dit optimaliseert de selectie van de juiste patiëntengroep en vermijdt mogelijke placentaire problemen (abruptio placentae, placenta-insufficiëntie) die kunnen resulteren in een suboptimale passage van allopurinol door de placenta. Tot slot dient toekomstig onderzoek op het gebied van perinatale hypoxie altijd rekening te houden met mogelijke gender-verschillen bij het analyseren van de gevonden uitkomsten. **Chapter 12** List of publications List of co-authors Dankwoord Over de auteur



LIST OF PUBLICATIONS

Kaandorp JJ, van den Broek MPH, van den Broek MJNL, *et al.* Rapid target allopurinol concentrations in the hypoxic fetus after maternal administration during labor. Accepted by Arch Dis Child-Fetal Neonatal Ed. 2013 Nov.

Kane AD, Hansell J, Herrera EA, Allison BJ, Niu Y, Brain KL, **Kaandorp JJ**, Derks JB, Giussani DA. Xanthine oxidase and the fetal cardiovascular defense to hypoxia in late gestation ovine pregnancy. Accepted by the J of Physiology. 2013 Nov.

Kane AD, Camm EJ, Richter HG, Lusby C, Tijsseling D, **Kaandorp JJ**, Derks JB, Ozanne SE and Giussani DA. Maternal to fetal allopurinol transfer and xanthine oxidase suppression in the late gestation pregnant rat. Accepted by Physiological Reports. 2013 Nov.

Kaandorp JJ, Derks JB, Oudijk MA, *et al*. Antenatal allopurinol reduces hippocampal brain damage after acute birth asphyxia in late gestational fetal sheep. Reproductive sciences. 2013 Jun (Epud Ahead of Print)

Kaandorp JJ, van Bel F, Veen S, *et al.* Long-term neuroprotective effects of allopurinol after moderate perinatal asphyxia. Follow-up of two randomised controlled trials. Arch Dis Child-Fetal Neonatal Ed, 2012 May; 97(3): F162-6 **Kaandorp JJ**, Derks JB, Benders MJNL en van Bel F. Farmacologische neuroprotectie tijdens en na asfyxie. Proceedings Slagercongres 2011.

Kaandorp JJ, Derks JB, Benders MJNL, Mol BWJ, Visser GHA en van Bel F. Allopurinol: Een nieuwe behandeling voor hersenschade ten gevolge van asfyxie? VOG Magazine (Vakblad voor Obstetrische en Gynaecologische Verpleegkundigen). Nov 2010.

Kaandorp JJ, Benders MJ, Rademaker CM, *et al*. Antenatal allopurinol for reduction of birth asphyxia induced brain damage (ALLO-Trial); a randomized double blind placebo controlled multicenter study. BMC Pregnancy and Childbirth 2010;10:8.

Kaandorp JJ, Benders MJ, Derks JB, van Bel F. Foetal hypoxia is an important determinant of birth asphyxia and subsequent adverse outcome: antenatal neuroprotection at term. Paediatrics and Child Health 2010;20:356-61.

Hack KE, **Kaandorp JJ**, Derks JB, Elias SG, Pistorius L, Visser GH. The value of umbilical artery Doppler velocimetry in the antenatal surveillance of uncomplicated monochorionic twin pregnancies. Ultrasound Obstet Gynecol. 2008;31(6):662-8.

LIST OF CO-AUTHORS

Beth J Allison

Department of Physiology, Development and Neuroscience, University of Cambridge, Cambridge, UK

Saskia CMJER Bakker

Department of Paediatrics, Gelre Hospital, Apeldoorn, the Netherlands

Sidarto Bambang Oetomo

Department of Neonatology, Maxima Medical Center, Veldhoven, the Netherlands

Frank van Bel

Department of Perinatology, University Medical Center, Utrecht, the Netherlands

Manon JNL Benders

Department of Perinatology, University Medical Center, Utrecht, the Netherlands

Kitty WM Bloemenkamp

Department of Obstetrics, Leiden University Medical Center, Leiden, Netherlands

Inge P de Boer

Diakonessen Hospital, Utrecht, the Netherlands

Janine Boon Diakonessen Hospital, Utrecht, the Netherlands

Arie F Bos

Department of Neonatology, University of Groningen, University Medical Center, Groningen, the Netherlands

Kirsty L Brain

Department of Physiology, Development and Neuroscience, University of Cambridge, Cambridge, UK

M PH van den Broek

Department of Clinical Pharmacy, University Medical Centre, Utrecht, the Netherlands

Emily J Camm

Department of Physiology, Development and Neuroscience, University of Cambridge, Cambridge, UK

Jan B Derks

Department of Perinatology, University Medical Center, Utrecht, the Netherlands

Ruurd M van Elburg

Department of Neonatology, VU University Medical Center, Amsterdam, the Netherlands; Danone Research, Center for Specialised Nutrition, Wageningen, The Netherlands

Maureen TM Franssen

Department of Obstetrics, University of Groningen, University Medical Center, Groningen, the Netherlands

AW Danilo Gavilanes

Department of Neonatology, Maastricht University Medical Center, Maastricht, the Netherlands

Dino A Giussani

Department of Physiology, Development and Neuroscience, University of Cambridge, Cambridge, UK

Floris Groenendaal

Department of Perinatology, University Medical Centre, Utrecht, the Netherlands

Timo R de Haan

Department of Neonatology, Amsterdam Medical Centre, Amsterdam, the Netherlands

Jeremy A Hansell

Department of Physiology, Development and Neuroscience, University of Cambridge, Cambridge, UK

LIST OF PUBLICATIONS - LIST OF CO-AUTHORS - DANKWOORD - OVER DE AUTEUR

Marline G Harmsen

Department of Perinatology, University Medical Centre, Utrecht, the Netherlands

Emilio A Herrera

Department of Physiology, Development and Neuroscience, University of Cambridge, Cambridge, UK

Anjoke JM Huisjes

Department of Obstetrics and Gynaecology, Gelre Hospital, Apeldoorn, the Netherlands

Corrie JWFM Jacobs

Department of Paediatrics, Jeroen Bosch Hospital, 's Hertogenbosch, the Netherlands

Andrew D Kane

Department of Physiology, Development and Neuroscience, University of Cambridge, Cambridge, UK

Jeannette S von Lindern

Department of Paediatrics, Groene Hart Hospital, Gouda, the Netherlands

Ciara Lusby

Department of Physiology, Development and Neuroscience, University of Cambridge, Cambridge, UK; University of Cambridge Metabolic Research Laboratories and MRC Metabolic Diseases Unit, Institute of Metabolic Science, Addenbrooke's Hospital, Cambridge, UK

Claudia A van Meir

Department of Obstetrics and Gynaecology, Groene Hart Hospital, Gouda, the Netherlands

Ben Willem J Mol

Department of Obstetrics, Amsterdam Medical Center, Amsterdam, the Netherlands

Peter GJ Nikkels

Department of pathology, University Medical Center, Utrecht, the Netherlands

Youguo Niu

Department of Physiology, Development and Neuroscience, University of Cambridge, Cambridge, UK

Martijn A Oudijk

Department of Perinatology, University Medical Center, Utrecht, the Netherlands

Susan E Ozanne

University of Cambridge Metabolic Research Laboratories and MRC Metabolic Diseases Unit, Institute of Metabolic Science, Addenbrooke's Hospital, Cambridge, UK

Maria G van Pampus

Department of Obstetrics, University of Groningen, University Medical Center, Groningen, the Netherlands

12 LIST OF PUBLICATIONS - LIST OF CO-AUTHORS - DANKWOORD - OVER DE AUTEUR

Martina M Porath

Department of Obstetrics, Maxima Medical Center, Veldhoven, the Netherlands

Karin MA Rademaker

Department of Clinical Pharmacy, University Medical Center, Utrecht, the Netherlands

Hans G Richter

Department of Physiology, Development and Neuroscience, University of Cambridge, Cambridge, UK

Monique Rijken

Department of Neonatology, Leiden University Medical Center, Leiden, Netherlands

Robbert JP Rijnders

Department of Obstetrics and Gynaecology, Jeroen Bosch Hospital, 's Hertogenbosch, the Netherlands

Elise Roze

Department of Neonatology, University of Groningen, University Medical Center, Groningen, the Netherlands

Liesbeth Scheepers

Department of Obstetrics, Maastricht University Medical Center, Maastricht, the Netherlands

Ewoud Schuit

Julius Center for Health Sciences and Primary Care, Utrecht, the Netherlands

Deodata Tijsseling

Department of Perinatology, University Medical Center, Utrecht, the Netherlands

Helen L Torrance

Department of Perinatology, University Medical Center, Utrecht, the Netherlands

Cuno SPM Uiterwaal

Julius Center for Health Sciences and Primary Care, Utrecht, the Netherlands

Monica MA Uniken Venema

Department of Clinical Psychology, Wilhelmina Children's Hospital, University Medical Center Utrecht, Utrecht, Netherlands

Sylvia Veen

Department of Neonatology, Leiden University Medical Center, Leiden, Netherlands

Gerard HA Visser

Department of Perinatology, University Medical Center, Utrecht, the Netherlands

Christine Willekes

Department of Obstetrics, Maastricht University Medical Center, Maastricht, the Netherlands

Maurice GAJ Wouters

Department of Obstetrics, VU University Medical Center, Amsterdam, the Netherlands